# **University of Alberta**

# Exercise, Epinephrine and IL-6 Mediated Regulation of Adipose Tissue Metabolism

by

Zhongxiao Wan

A thesis submitted to the Faculty of Graduate Studies and Research in partial fulfillment of the requirements for the degree of

> Doctor of Philosophy in Nutrition and Metabolism

Department of Agricultural, Food and Nutritional Science

©Zhongxiao Wan Fall 2012 Edmonton, Alberta

Permission is hereby granted to the University of Alberta Libraries to reproduce single copies of this thesis and to lend or sell such copies for private, scholarly or scientific research purposes only. Where the thesis is converted to, or otherwise made available in digital form, the University of Alberta will advise potential users of the thesis of these terms.

The author reserves all other publication and other rights in association with the copyright in the thesis and, except as herein before provided, neither the thesis nor any substantial portion thereof may be printed or otherwise reproduced in any material form whatsoever without the author's prior written permission.

# ABSTRACT

The biochemical and molecular mechanisms underlying the beneficial effects of exercise remain elusive. Growing evidence suggests that white adipose tissue (WAT) is an important organ that exercise exerts beneficial effects on. The role of interleukin 6 (IL-6) in mediating WAT metabolism remains under debate.

The first objective of this thesis is to understand exercise, epinephrine and IL-6 in modulating pyruvate dehydrogenase kinase 4 (PDK4) and phosphoenolpyruvate carboxykinase (PEPCK) gene expression in WAT. Exercise and epinephrine increased PDK4 mRNA levels in WAT from lean rats. The effects of epinephrine on PDK4 were mediated via p38 mitogen-activated protein kinase (MAPK) and peroxisome proliferator-activated receptor gamma (PPAR $\gamma$ ). The ability of exercise and epinephrine to induce PDK4 mRNA was maintained in obese rats.

In contrast to epinephrine, IL-6 decreased the gene expression and protein content of PEPCK and PDK4 in cultured WAT. Although an acute bout of treadmill running did not activate IL-6 signalling in adipose tissue, the exercise-induced increases in PEPCK and PDK4 mRNA were attenuated in WAT from IL-6<sup>-/-</sup> mice.

The second broad objective of this thesis was to understand the role of IL-6 in mediating WAT metabolism by utilizing IL-6<sup>-/-</sup> mice fed a high fat diet (HFD). IL-6<sup>-/-</sup> HFD mice were more glucose and insulin intolerant than wild type controls and this was mirrored by reductions in the insulin-stimulated activation of protein kinase B (PKB) and increases in the phosphorylation of extracellular signal regulated kinases (ERK) and c-Jun N-terminal kinases (JNK) in WAT. Epinephrine stimulated lipolysis and hormone

sensitive lipase (HSL) phosphorylation were blunted, adiponectin mRNA expression and secretion, and AMPK phosphorylation were reduced in WAT from IL-6<sup>-/-</sup> mice. These results suggest a unique role for IL-6 in the maintenance of adipose tissue metabolism during the development of insulin resistance.

In summary, exercise, similar to thiazolidinediones (TZDs), is a potent stimulus that can induce WAT glyceroneogenic enzymes. IL-6 is essential for the metabolic homeostasis in WAT under conditions of metabolic stress: exercise and chronic nutrient excess.

# ACKNOWLEDGEMENT

There have been so many people involved in aiding my development and growth as a scientist and person that I would be remiss to mention as many of them as I can. Firstly, I would like to express my sincere appreciation to my supervisor Dr. David Wright, for always having a positive attitude, enthusiasm and passion for science. Thank you for your guidance, encouragement and constant patience throughout these years. My thanks also go to my second supervisor Dr. Catherine Chan. Thank you for your insightful comments on my research projects, for generously providing guidance and enormous patience on the technical aspect of this thesis.

I am also grateful to my supervisory committee members. To Dr. Catherine Field, thank you not only for your project assistance, but also for your patience and immense support of my future goals. To René Jacobs, thank you for your helpful advices and guidance on my research projects. I would also like to thank Dr. Walter Dixon and Dr. Raylene Reimer for your willingness to participate in my examination.

To all of the members of the Wright lab (past and present), Dr. Bruce Frier, Deon Williams, Dr. Brianne Thrush, Melanie Legare, Marie Soleil Beaudoin, Laelie Snook, Dr. Christopher Perry, Tara MacDonald, Laura Castellani, Mandy Stefanson, and Jared Root McCaig for all the fun times we have together in and out of lab. To all of the present members of the Holloway lab at the University of Guelph.

To my parents back home, thank you for letting me be abroad and unwavering support over the course of my entire education. Knowing that you are behind me has made it possible to achieve all of my goals.

# TABLE OF CONTENTS

Title page
Abstract
Acknowledgements
Table of Contents
List of Tables
List of Figures
List of Abbreviations
Chapter 1 Study Rationale1
1.1 Rationale1
1.2 Objectives and Hypothesis
1.2.1 Objective 1
1.2.2 Objective 2
1.2.3 Objective 3
1.2.4 Objective 4
1.3 Chapter Format
1.3.1 Chapter 2
1.3.2 Chapter 3
1.3.3 Chapter 4
1.3.4 Chapter 5
1.3.5 Chapter 67
1.3.6 Chapter 77
1.4 Literature Cited
Chapter 2 Literature Review
2.1 Overview of White Adipose Tissue 12
2.1.1 Essential role in metabolic homeostasis
2.1.2 Adipose tissue as an endocrine organ
2.1.3 Depot-specific phenotype in adipose tissue

2.2 WAT Mitochondria	. 15
2.2.1 Mitochondria in WAT glyceroneogenesis	. 15
2.2.1.1 Glyceroneogenesis pathway	. 16
2.2.1.2 PDK4	. 18
2.2.1.3 PEPCK	. 19
2.2.2 Mitochondria in WAT <i>de novo</i> lipogenesis	. 21
2.2.3 Mitochondria in WAT adipogenesis	. 22
2.2.4 Mitochondria in adiponectin synthesis	. 22
2.2.5 Adipose tissue mitochondrial dysfunction and insulin resistance	. 23
2.2.5.1 Evidence for adipose mitochondrial dysfunction in insulin resistance	. 23
2.2.5.2 Potential mechanisms by which impaired adipose mitochondrial func could lead to insulin resistance	tion 24
2.3 Adipose Tissue Insulin Resistance	. 25
2.3.1 Insulin action and signaling pathways in adipose tissue	. 25
2.3.2 MAPKs in adipose tissue insulin resistance	. 27
2.3.3 Adiponectin in adipose tissue insulin resistance	. 29
2.3.3.1 Adiponectin and adipose tissue biology	. 29
2.3.3.2 Regulation of adiponectin in adipose tissue	. 30
2.4 Response of WAT to Exercise	. 31
2.4.1 Exercise-induced adipose tissue mitochondrial biogenesis	. 31
2.4.2 Exercise-induced adipose tissue lipolysis	. 32
2.4.3 Effects of exercise on adipokines	. 33
2.4.3.1 Adiponectin	. 33
2.4.3.2 Leptin	. 34
2.4.4 Exercise on glucose uptake in adipose tissue	. 34
2.4.5 Mediators of exercise-induced adipose tissue metabolism	. 34
2.4.5.1 Catecholamines	. 34
2.4.5.2 Irisin	. 35
2.4.5.3 IL-6	. 36
2.4.5.4 The interactions between epinephrine and IL-6 during exercise	. 37
2.4.5.5 The interactions between IL-6 and other myokines during exercise	. 38

2.5 IL-6 in WAT Metabolism	39
2.5.1 IL-6 production and IL-6 signaling in adipose tissue	39
2.5.2 IL-6 and insulin resistance	40
2.5.2.1 In vitro effects of IL-6 on adipocyte metabolism	40
2.5.3 Effects of IL-6 on metabolism in genetically modified mouse model	
2.6 Thesis Overview	44
2.7 Literature Cited	
Chapter 3 Epinephrine Mediated Regulation of PDK4 mRNA in Rat Adipo	se Tissue
3.1 Introduction	82
3.2 Methods and Materials	
3.2.1 Materials	
3.2.2 Treatment of Rodents	85
3.2.3 Adipose Tissue Organ Culture	86
3.2.4 Western Blot Analysis	87
3.2.5 Lipolysis Assays	
3.2.6 Real Time RT-PCR	
3.2.7 Statistical Analysis	88
3.3 Results	89
3.3.1 Exercise, Fasting and Epinephrine Increase PDK4 mRNA Expr Epididymal Adipose Tissue	ession in 
3.3.2 Epinephrine Increases AMPK and P38 MAPK Signaling in Epididymal Adipose Tissue	Cultured
3.3.3 AMPK Agonists Do Not Increase PDK4 mRNA Expression in Rat Ep Adipose Tissue	pididymal 93
3.3.4 SB202190 Does Not Attenuate AMPK or PKA Signaling in Epididym	ıal 95
Adipose Tissue Cultures	
3.3.5 SB202190 Attenuates Epinephrine Induced Increases in PDK4 mRNA	97
3.3.6 Epinephrine Mediated Increases in PDK4 Are Inhibited by PPARγ A	Antagonist 
3.4 Discussion	
3.5 Literature Cited	104

Chapter 4 Epinephrine Induces PDK4 mRNA Expression in Adipose Tissue from Obese, Insulin Resistant Rats
4.1 Introduction
4.2 Methods and Materials
4.2.1 Methods and Procedures
4.2.2 Statistical analysis
4.3 Results and Discussion
4.4 Literature Cited
Chapter 5 IL-6 Indirectly Modulates the Induction of Glyceroneogenic Enzymes in Adipose Tissue during Exercise
5.1 Introduction
5.2 Methods and Materials
5.2.1 Materials
5.2.2 Treatment of Animals
5.2.3 Adipose Tissue Organ Culture
5.2.4 Lipolysis in Adipose Tissue Explants 120
5.2.5 Western Blot Analysis 120
5.2.6 Real Time PCR
5.2.7 Statistical Analysis 128
5.3 Results
5.3.1 Effects of <i>ex vivo</i> IL-6 Treatment on Signalling and Gene expression in eWAT
5.3.2 The IL-6 Mediated Activation of AMPK Occurs Independent of Lipolysis 13
5.3.3 Markers of IL-6 Signalling Are Not Activated in Adipose Tissue during Exercise
5.3.4 Exercise-Mediated Increases in PEPCK and PDK4 Expression Are Attenuated in IL-6 <sup>-/-</sup> Mice
5.3.5 AMPK Phosphorylation Is Altered in eWAT from IL-6 <sup>-/-</sup> Mice
5.3.6 There Are No Apparent Reductions in Beta Adrenergic Signalling in eWAT from IL-6 <sup>-/-</sup> Mice
5.4 Discussion
5.5 Literature Cited

Chapter 6 IL-6 Deficient Mice Fed a High Fat Diet Display Alterations in Adipose Tissue Metabolism
6.1 Introduction
6.2 Methods and Materials
6.2.1 Materials
6.2.2 Treatment of Animals
6.2.3 Glucose and Insulin Tolerance Tests 153
6.2.4 In Vivo Insulin Signaling 153
6.2.5 Adipose Tissue Explants 153
6.2.6 Plasma Adiponectin and Adiponectin Secretion 154
6.2.7 Mitochondrial Respiration in Adipose Tissue
6.2.8 Western Blot Analysis 155
6.2.9 Mitochondrial DNA156
6.2.10 Real Time PCR
6.2.11 Statistical Analysis157
6.3 Results
6.3.1 No Phenotypic Differences in Adipose Tissue Metabolism in Chow Fed IL-6 <sup>-/-</sup> Mice
6.3.2 Food Intake, Body Weight and Fat Pad Mass in WT and IL-6 <sup>-/-</sup> Fed a HFD . 159
6.3.3 IL-6 <sup>-/-</sup> Mice Have Impaired Glucose Homeostasis Compared to WT Fed a HFD
6.3.4 Insulin Signaling Is Attenuated in Adipose Tissue from IL-6 <sup>-/-</sup> Mice Fed a HFD
6.3.5 Beta-Adrenergic Signaling Is Attenuated in Adipose Tissue from IL-6 <sup>-/-</sup> Mice
6.3.6 IL-6 Does Not Potentiate Epinephrine-Stimulated Lipolysis
6.3.7 Reductions in Adiponectin mRNA Expression and Secretion in Adipose Tissue from IL-6 <sup>-/-</sup> Mice
6.3.8 Markers of Inflammation and Macrophage Infiltration in Adipose Tissue from IL-6 <sup>-/-</sup> Mice
6.3.9 Mitochondrial Content Is Not Different in Adipose Tissue from WT and IL-6 <sup>-/-</sup> Mice fed a HFD
6.4 Discussion

	6.5 Literature Cited	175
(	Chapter 7 General Discussion	181
	7.1 Summary of Main Findings	181
	7.2 Methodological Considerations	183
	7.2.1 Adipose tissue organ culture	183
	7.2.2 Selection of adipose depots	184
	7.2.3 Exercise models	185
	7.2.3.1 Swimming	185
	7.2.3.2 Treadmill running	185
	7.2.3.3 Gender differences in response to exercise	186
	7.2.4 IL-6 <sup>-/-</sup> mouse model	186
	7.2.5 HFD	188
	7.3 Exercise on Adipose Tissue Metabolism	189
	7.4 IL-6 and Adipose Tissue Metabolism	191
	7.5 Relevance of the Current Findings to Human Health Research	193
	7.6 Future Directions	195
	7.6 Literature Cited	197

# LIST OF TABLES

# LIST OF FIGURES

Figure 2-1 General description of glyceroneogenesis and pyruvate routes towards fatty     acid re-esterification in adipose tissue.     17
Figure 2-2   A schematic diagram of several central points of insulin signaling pathway and its action in WAT.     27
Figure 3-1 Effects of exercise, fasting and epinephrine on PDK4 mRNA and the phosphorylation of PDH protein expression
Figure 3-2 Effects of <i>ex vivo</i> epinephrine on p38 and AMPK signalling pathways92
Figure 3-3 Effects of ex vivo AICAR and metformin on AMPK and p38 signalling pathways and PDK4 mRNA.   94
<b>Figure 3-4</b> Effects of <i>ex vivo</i> SB on the phosphorylation of MK2, AMPK, HSL, lipolysis and PDK4 mRNA
<b>Figure 3-5</b> Effects of <i>ex vivo</i> GW9662 on rosiglitazone and epinephrine mediated increases in PDK4 mRNA and signalling
Figure 4-1Effects of epinephrine injection on PDK4, p38 and AMPK signallingpathways from chow and HFD rats.116
Figure 5-1 IL-6 signalling in cultured adipose tissue.   129
Figure 5-2 Effects of <i>ex vivo</i> IL-6 on PEPCK and PDK4
Figure 5-3 Effects of <i>ex vivo</i> IL-6 on adipose tissue lipolysis
Figure 5-4 Markers of IL-6 signalling in adipose tissue following exercise
<b>Figure 5-5</b> Exercise-mediated induction of PEPCK and PDK4 from WT and IL-6 <sup>-/-</sup> mice.
<b>Figure 5-6</b> Exercise mediated AMPK signaling from WT and IL-6 <sup>-/-</sup> mice
<b>Figure 5-7</b> Beta adrenergic signalling in adipose tissue from WT and IL-6 <sup>-/-</sup> mice 139
<b>Figure 6-1</b> Adipose tissue metabolism in chow fed WT and IL-6 <sup>-/-</sup> mice
Figure 6-2 Glucose and insulin tolerance in WT and IL-6 <sup>-/-</sup> mice fed a HFD
<b>Figure 6-3</b> <i>In vivo</i> insulin signaling in adipose tissue from WT and IL-6 <sup>-/-</sup> mice fed a HFD
<b>Figure 6-4</b> <i>Ex vivo</i> epinephrine-stimulated lipolysis in adipose tissue from WT and IL-6 <sup>-/-</sup> mice fed a HFD

Figure 6-5 Adiponectin expression, secretion and plasma levels in WT and I	L-6 <sup>-/-</sup> mice
fed a HFD	166
Figure 6-6 mRNA markers of inflammation in eWAT from WT and IL-6 <sup>-/-</sup> f	ed a HFD. 167
<b>Figure 6-7</b> AMPK and mitochondrial content and respiration in adipose tissue and IL-6 <sup>-/-</sup> fed a HFD	e from WT 169

# LIST OF ABBREVIATIONS

ACC	acetyl coA carboxylase
AICAR	5-Aminoimidazole-4-carboxyamide ribonucleoside
АМРК	5'AMP activated protein kinase
ANOVA	analysis of variance
aPKCç/λ	atypical isoforms of PKC
ATF3	activating transcription factor 3
ATGL	adipose triglyceride lipase
ATOC	adipose tissue organ culture
AUC	area under the curve
β2MG	beta 2 microglobulin
BSA	bovine serum albumin
cAMP	cyclic adenosine monophosphate
CCAC	Canadian Council on Animal Care
cDNA	complimentary deoxyribonucleic acid
C/EBP	CCAAT/enhancer binding protein
Cidea	cell death-inducing DNA fragmentation factor- $\alpha$ -like effecter A
CORE1	ubiquinone:cytochrome c oxidoreductase core I subunit
CPT1	carnitine palmitoyltransferase I
CREB	cAMP response element binding protein
CXCL1	chemokine (C-X-C motif) ligand 1
DCA	dichloroacetate
DIO	diet induced obese

ER	endoplasmic reticulum
ERK1/2	extracellular signal regulated kinases 1/2
eWAT	epididymal white adipose tissue
ES	embryonic stem
ETS	electron transport system
FIRKO	fat-specific disruption of the insulin receptor knockout
FNDC5	fibronectin type III domain 5
G-3-P	glycerol 3-phosphate
GLP-1	glucagon-like peptide-1
GLUT4	glucose transporter type 4
gp130	glycoprotein 130
GTT	glucose tolerance test
HDL-C	high density lipoprotein-cholesterol
HFD	high fat diet
HMW	high-molecular weight
HSL	hormone sensitive lipase
HSP60	heat shock protein 60
IL-6	interleukin-6
IL-6 <sup>-/-</sup>	IL-6 deficient
IL-6Ra	IL-6 receptor alpha
iNOS	inducible nitric oxide synthase
IP	intraperitoneal
IR	insulin receptor

IRS	insulin receptor substrate
ITT	insulin tolerance test
JAK	Janus kinase
JNKs	c-Jun N-terminal kinases
KRB	Krebs-Ringer Buffer
LIF	leukemia inhibitory factor
LKB-1	liver kinase B1
МАРК	mitogen-activated protein kinase
MK2	MAPK activated protein kinase 2
mtDNA	mitochondrial DNA
mTOR	mammalian target of rapamycin
mtTFAM	mitochondrial transcription factor A
ND5	NADH dehydrogenase subunit 5
NEFAs	non-esterified fatty acids
NRF-1	nuclear respiratory factor 1
OLETF	Otsuka Long Evans Tokushima Fatty
PDC	pyruvate dehydrogenase complexes
PDK1/2	phosphoinositide dependent kinase-1 and 2
PDK4	pyruvate dehydrogenase kinase 4
PDP	pyruvate dehydrogenase phosphatase
РЕРСК	phosphoenolpyruvate carboxykinase
PGC-1a	peroxisome proliferator-activated receptor gamma coactivator 1-alpha
PI3K	phosphatidylinositol 3-kinase

PIP3	phosphatidylinositol 3, 4, 5-trisphosphate	
РКА	protein kinase A	
РКВ	protein kinase B	
PP2A	protein phosphatase type 2A	
ΡΡΑRγ	peroxisome proliferator-activated receptor gamma	
PPRE	peroxisome proliferator hormone response elements	
PRDM16	PRD1-BF1-RIZ1 homologous domain containing 16	
rh	recombinant human	
S6K1	p70S6 kinase	
SAPK/JNK	stress-activated protein kinase/Jun-amino-terminal kinase	
SAT	subcutaneous adipose tissue	
SEM	standard error of the mean	
SHP-2	Src homology2-containing tyrosine phosphatise	
siRNA	small interfering RNA	
SOCS3	suppressor of cytokine signaling 3	
SREBP-1c	sterol regulatory element-binding protein-1c	
STAT3	signal transducer and activator of transcription 3	
T2DM	type 2 diabetes	
TBST	Tris buffered saline/0.1% Tween 20	
TGs	triglycerides	
<b>TNF-</b> α	tumor necrosis factor-alpha	
Tyr	tyrosine	
TZDs	thiazolidinediones	

UCP1	uncoupling protein 1
VAT	visceral adipose tissue
WAT	white adipose tissue
WT	wild type
ZDF	Zucker Diabetic Fatty

# **Chapter 1 Study Rationale**

## **1.1 Rationale**

There is increasing body of evidence to support that white adipose tissue (WAT) might play an important role in regulating the beneficial effects of exercise on systemic metabolism. For instance, exercise has been reported to induce mitochondrial biogenesis (1-2) and improve insulin sensitivity in WAT (3-5). Catecholamines are central modulators for regulating exercise-induced WAT metabolism such as inducing lipolysis (6) and mitochondrial biogenesis (1). Skeletal muscle has been identified as an endocrine organ that secretes "myokines" with one of the best characterized being interleukin 6 (IL-6) (7). It has been proposed that increases in muscle derived IL-6 may be a potential mediator involved in exercise-induced WAT metabolism (8). For example, IL-6 has been reported to stimulate adipose tissue lipolysis (9) and to activate 5'AMP activated protein kinase (AMPK) (10) in adipocytes.

Thiazolidinediones (TZDs) are a commonly prescribed class of insulin-sensitizing drugs that are thought to mediate a large portion of their beneficial effects through their up-regulation of the glyceroneogenic pathway in WAT (11-12). Given the deleterious side effects of TZDs (13), this thesis initially sought to determine if exercise could mimic TZD's effects on glyceroneogenic enzymes with a special focus on phosphoenolpyruvate carboxykinase (PEPCK) and pyruvate dehydrogenase kinase 4 (PDK4), two key enzymes involved in WAT glyceroneogenesis (11-12). Additionally, this thesis aimed to elucidate the potential involvement of epinephrine and IL-6 in the regulation of exercise-induced

PEPCK and PDK4 mRNA expression and the potential signaling pathways that might be involved.

Currently, the role of IL-6 in the development of systemic insulin resistance is still under debate (14). Owing to the central role of skeletal muscle and liver in the regulation of glucose homeostasis, studies investigating the role of IL-6 in glucose homeostasis have focused primarily on these tissues (15-20). Contradictory results with negative (17-18), neutral (16, 19) and positive (15, 20) effects of IL-6 on liver and/or skeletal muscle insulin sensitivity have been reported.

Interstitial concentrations of IL-6 surrounding adipose tissue are orders of magnitude greater than levels in the circulation (21) suggesting that adipose tissue derived IL-6 could play a pivotal role in the maintenance of adipose tissue metabolism and function in an autocrine and/or paracrine manner. Although evidence from *in vitro* studies suggests that IL-6 causes insulin resistance in adipocytes (22-24), there is also evidence that IL-6 may favorably regulate adipose tissue function presumably through its modulation of AMPK (10). Given the purported role of IL-6 in the control of adipose tissue function and the growing appreciation for adipose tissue in the control of systemic energy homeostasis (25), this thesis further targeted the metabolic action of IL-6 on rodent adipose tissue by studying IL-6<sup>-/-</sup> mice under both chow and a high fat diet (HFD) conditions.

#### **1.2 Objectives and Hypothesis**

#### 1.2.1 Objective 1

To determine the regulation of PDK4 mRNA expression in epididymal adipose tissue (eWAT) from lean rats by exercise (swimming) and epinephrine and the potential signalling pathways involved.

*Hypothesis 1* Swimming exercise and epinephrine will increase PDK4 mRNA expression in rats eWAT.

*Hypothesis* 2 Epinephrine will increase p38 mitogen-activated protein kinase (MAPK) and AMPK signaling and these pathways will be involved in epinephrine induced PDK4 mRNA expression in rats eWAT.

*Hypothesis 3* Peroxisome proliferator-activated receptor gamma (PPARγ) will be involved in the pathway through which epinephrine increases PDK4 mRNA expression in rats eWAT.

# 1.2.2 Objective 2

To determine if 2 hours of swimming exercise and acute increases in epinephrine *in vivo* can induce PDK4 mRNA expression in eWAT from insulin resistant rats and if the reputed signaling pathways mediating this effect, i.e. p38MAPK are intact.

*Hypothesis 1* An acute bout of exercise and acute increases in epinephrine *in vivo* will induce PDK4 mRNA expression in eWAT from obese, insulin resistant rats.

*Hypothesis* 2 p38MAPK activation by epinephrine will be intact in eWAT from obese, insulin resistant rats.

# 1.2.3 Objective 3

To determine 1) the effects of *ex vivo* IL-6 treatment in mouse eWAT on PEPCK and PDK4 expression; 2) if IL-6 signaling is activated and if PEPCK and PDK4 mRNA expression is increased in mouse eWAT immediately after an acute bout of treadmill running; and 3) through the utilization of IL-6<sup>-/-</sup> mouse model, to elucidate the role of IL-6 in modulating the effects of treadmill running on PEPCK and PDK4 mRNA expression in mouse eWAT and the reputed signaling pathways therein.

*Hypothesis 1* IL-6 will directly attenuate the expression of PEPCK and PDK4 in cultured mouse eWAT.

*Hypothesis 2* A single bout of exercise will activate IL-6 signalling and induce PEPCK and PDK4 mRNA expression in mouse eWAT.

*Hypothesis 3* The induction of glyceroneogenic enzyme expression will be potentiated in eWAT from IL- $6^{-/-}$  mice.

# 1.2.4 Objective 4

To investigate the metabolic action of IL-6 on rodent adipose tissue by characterizing the adipose tissue phenotypes from age matched wild type and IL- $6^{-/-}$  mice under both chow and high fat diet (HFD).

*Hypothesis 1* eWAT from IL-6<sup>-/-</sup> mice fed a chow diet will display reductions in insulin action *ex vivo*, have blunted  $\beta$ -adrenergic responsiveness *ex vivo* and decreases in adipose tissue mitochondrial content and respiration compared to age matched WT controls under the same diet.

*Hypothesis 2* eWAT from IL-6<sup>-/-</sup> mice fed a HFD will demonstrate reductions in insulin action *in vivo* and *ex vivo* and have blunted  $\beta$ -adrenergic responsiveness *ex vivo* compared to age matched WT groups under the same diet.

*Hypothesis 3* eWAT from IL- $6^{-/-}$  mice fed a HFD will have decreases in AMPK phosphorylation, mitochondrial content and respiration, and reductions in adiponectin expression and secretion in comparison with age matched WT groups under the same diet.

# **1.3 Chapter Format**

## **1.3.1 Chapter 2**

Chapter 2 gives a review of the literature about the study of adipose tissue metabolism related to this thesis research. Firstly, a general introduction to adipose tissue biology and the significant role of adipose tissue mitochondria in mediating adipose tissue metabolism will be outlined. Secondly, insulin signaling pathways, the role of MAPK signaling pathways in adipose tissue insulin resistance and a potential role of adiponectin in regulating adipose tissue proper will be summarized. Thirdly, the effects of exercise on adipose tissue metabolism and the potential mediators (catecholamines, irisin and IL-6) involved in exercise induced adipose tissue metabolism will be provided. Lastly, the role of IL-6 in adipose tissue insulin resistance will be discussed extensively.

## **1.3.2 Chapter 3**

Chapter 3 contains the first objective of this thesis research that has been published in the American Journal of Physiology - Cell Physiology (Wan Z, Thrush AB, Legare M, Frier BC, Sutherland LN, Williams DB, Wright DC. 2010). Epinephrine Mediated Regulation of PDK4 mRNA in Rat Adipose Tissue. Am J Physiol Cell Physiol.

2010, 299(5):C1162-70. Firstly, the effect of exercise and *in vivo* epinephrine treatment on PDK4 mRNA expression in eWAT from lean and insulin sensitive rats will be discussed. Secondly, the effects of *ex vivo* epinephrine treatment on PDK4 mRNA expression in eWAT from lean and insulin sensitive rats and the potential signaling pathways (AMPK, p38MAPK and PPAR $\gamma$ ) that might be involved will be presented.

#### **1.3.3 Chapter 4**

Chapter 4 contains the second objective of this thesis research that has been published in *Obesity* (Wan Z, Frier BC, Williams DB, Wright DC. 2012). *Epinephrine Induces PDK4 mRNA Expression in Adipose Tissue from Obese, Insulin Resistant Rats. Obesity, 2012, 20(2):453-56.* The study of exercise and *in vivo* epinephrine treatment on PDK4 mRNA expression in eWAT from obese and insulin resistant rats will be presented.

#### **1.3.4 Chapter 5**

Chapter 5 contains the third objective of this thesis research that has been published in *PLoS One* (Wan Z, Ritchie I, Beaudoin MS, Castellani L, Chan CB, Wright DC. 2012). *IL-6 Indirectly Modulates the Induction of Glyceroneogenic Enzymes in Adipose Tissue during Exercise. PLoS One, 2012;7(7):e41719.* First, the effects of *ex vivo* IL-6 treatment in mouse adipose tissue on PEPCK and PDK4 mRNA and protein expression, free fatty acids and glycerol release, and AMPK and p38 MAPK signaling pathways will be presented. Second, the effects of an acute bout of treadmill running on PEPCK and PDK4 mRNA expression and IL-6 signaling in mouse adipose tissue will be shown. Lastly, by utilizing IL-6<sup>-/-</sup> mice, the role of IL-6 in modulating the effects of exercise on the expression of PEPCK and PDK4 mRNA expression and the potential signaling pathways that might be involved will be discussed.

#### **1.3.5 Chapter 6**

Chapter 6 contains the fourth objective of this thesis research that has been *under preparation for submission*. Firstly, adipose tissue insulin action, beta-adrenergic responsiveness and mitochondrial content/function between age matched WT and IL-6<sup>-/-</sup> mice under chow diet will be characterized. Secondly, the systemic glucose and insulin tolerance between age matched WT and IL-6<sup>-/-</sup> mice following 10 weeks of HFD will be characterized by intraperitoneal glucose and insulin tolerance test. Thirdly, adipose tissue insulin action and beta-adrenergic responsiveness between age matched WT and IL-6<sup>-/-</sup> mice under HFD diet will be presented. Lastly, the potential mechanisms that might be involved in adipose tissue insulin resistance from IL-6<sup>-/-</sup> HFD mice will also be explored.

# **1.3.6 Chapter 7**

Chapter 7 is a general discussion relating the findings of the thesis research from Chapter 3 to Chapter 6. Firstly, methodological considerations for the utilization of specific models will be discussed. Secondly, the results we have obtained from Chapter 3 to Chapter 6 will be discussed and put into a wider perspective. Thirdly, implications for future research that can be taken from the current thesis will also be outlined.

#### **1.4 Literature Cited**

1. Sutherland LN, Bomhof MR, Capozzi LC, Basaraba SA, Wright DC. Exercise and adrenaline increase PGC-1{alpha} mRNA expression in rat adipose tissue. J Physiol. 2009;587(Pt 7):1607-17.

2. Xu X, Ying Z, Cai M, Xu Z, Li Y, Jiang SY, et al. Exercise ameliorates high-fat diet-induced metabolic and vascular dysfunction, and increases adipocyte progenitor cell population in brown adipose tissue. Am J Physiol Regul Integr Comp Physiol. 2011;300(5):R1115-25.

3. Goodyear LJ, Hirshman MF, Horton ED, Knutson SM, Wardzala LJ, Horton ES. Exercise training normalizes glucose metabolism in a rat model of impaired glucose tolerance. Metabolism. 1991;40(5):455-64.

4. Hirshman MF, Goodyear LJ, Horton ED, Wardzala LJ, Horton ES. Exercise training increases GLUT-4 protein in rat adipose cells. Am J Physiol. 1993;264(6 Pt 1):E882-9.

Hussey SE, McGee SL, Garnham A, Wentworth JM, Jeukendrup AE, Hargreaves
M. Exercise training increases adipose tissue GLUT4 expression in patients with type 2
diabetes. Diabetes Obes Metab. 2011;13(10):959-62.

6. Horowitz JF. Fatty acid mobilization from adipose tissue during exercise. Trends Endocrinol Metab. 2003;14(8):386-92.

7. Pedersen BK, Akerstrom TC, Nielsen AR, Fischer CP. Role of myokines in exercise and metabolism. J Appl Physiol. 2007;103(3):1093-8.

8. Pedersen BK, Febbraio MA. Muscle as an endocrine organ: focus on musclederived interleukin-6. Physiol Rev. 2008;88(4):1379-406.

8

9. Petersen EW, Carey AL, Sacchetti M, Steinberg GR, Macaulay SL, Febbraio MA, et al. Acute IL-6 treatment increases fatty acid turnover in elderly humans in vivo and in tissue culture in vitro. Am J Physiol Endocrinol Metab. 2005;288(1):E155-62.

10. Kelly M, Keller C, Avilucea PR, Keller P, Luo Z, Xiang X, et al. AMPK activity is diminished in tissues of IL-6 knockout mice: the effect of exercise. Biochem Biophys Res Commun. 2004;320(2):449-54.

11. Cadoudal T, Distel E, Durant S, Fouque F, Blouin JM, Collinet M, et al. Pyruvate dehydrogenase kinase 4: regulation by thiazolidinediones and implication in glyceroneogenesis in adipose tissue. Diabetes. 2008;57(9):2272-9.

12. Cadoudal T, Blouin JM, Collinet M, Fouque F, Tan GD, Loizon E, et al. Acute and selective regulation of glyceroneogenesis and cytosolic phosphoenolpyruvate carboxykinase in adipose tissue by thiazolidinediones in type 2 diabetes. Diabetologia. 2007;50(3):666-75.

13. Stafylas PC, Sarafidis PA, Lasaridis AN. The controversial effects of thiazolidinediones on cardiovascular morbidity and mortality. Int J Cardiol. 2009;131(3):298-304.

14. Allen TL, Febbraio MA. IL6 as a mediator of insulin resistance: fat or fiction?Diabetologia. 2010;53(3):399-402.

15. Matthews VB, Allen TL, Risis S, Chan MH, Henstridge DC, Watson N, et al. Interleukin-6-deficient mice develop hepatic inflammation and systemic insulin resistance. Diabetologia. 2010;53(11):2431-41.

16. Fritsche L, Hoene M, Lehmann R, Ellingsgaard H, Hennige AM, Pohl AK, et al.IL-6 deficiency in mice neither impairs induction of metabolic genes in the liver nor

9

affects blood glucose levels during fasting and moderately intense exercise. Diabetologia. 2010;53(8):1732-42.

17. Klover PJ, Zimmers TA, Koniaris LG, Mooney RA. Chronic exposure to interleukin-6 causes hepatic insulin resistance in mice. Diabetes. 2003;52(11):2784-9.

18. Kim HJ, Higashimori T, Park SY, Choi H, Dong J, Kim YJ, et al. Differential effects of interleukin-6 and -10 on skeletal muscle and liver insulin action in vivo. Diabetes. 2004;53(4):1060-7.

19. Rotter Sopasakis V, Larsson BM, Johansson A, Holmang A, Smith U. Short-term infusion of interleukin-6 does not induce insulin resistance in vivo or impair insulin signalling in rats. Diabetologia. 2004;47(11):1879-87.

20. Weigert C, Hennige AM, Brodbeck K, Haring HU, Schleicher ED. Interleukin-6 acts as insulin sensitizer on glycogen synthesis in human skeletal muscle cells by phosphorylation of Ser473 of Akt. Am J Physiol Endocrinol Metab. 2005;289(2):E251-7.

21. Nielsen NB, Hojbjerre L, Sonne MP, Alibegovic AC, Vaag A, Dela F, et al. Interstitial concentrations of adipokines in subcutaneous abdominal and femoral adipose tissue. Regul Pept. 2009;155(1-3):39-45.

22. Fasshauer M, Kralisch S, Klier M, Lossner U, Bluher M, Klein J, et al. Adiponectin gene expression and secretion is inhibited by interleukin-6 in 3T3-L1 adipocytes. Biochem Biophys Res Commun. 2003;301(4):1045-50.

23. Rotter V, Nagaev I, Smith U. Interleukin-6 (IL-6) induces insulin resistance in 3T3-L1 adipocytes and is, like IL-8 and tumor necrosis factor-alpha, overexpressed in human fat cells from insulin-resistant subjects. J Biol Chem. 2003;278(46):45777-84.

24. Lagathu C, Bastard JP, Auclair M, Maachi M, Capeau J, Caron M. Chronic interleukin-6 (IL-6) treatment increased IL-6 secretion and induced insulin resistance in adipocyte: prevention by rosiglitazone. Biochem Biophys Res Commun. 2003;311(2):372-9.

25. Rosen ED, Spiegelman BM. Adipocytes as regulators of energy balance and glucose homeostasis. Nature. 2006;444(7121):847-53.

# **Chapter 2 Literature Review**

#### 2.1 Overview of White Adipose Tissue

#### 2.1.1 Essential role in metabolic homeostasis

It has now been firmly established that white adipose tissue (WAT) is a multifunctional organ that interacts extensively with other tissues in whole-body metabolic homeostasis through a network of endocrine, paracrine, and autocrine signals (1-2). WAT contains several types of cells: mature adipocytes and other cells usually termed as the stromal vascular fraction including endothelial cells, fibroblasts, macrophages and preadipocytes (2).

The primary physiological functions of WAT can be classified into two aspects. First, it is the largest depot in the body for the triglycerides (TGs) storage under the fed state. In contrast, during fasting and exercise, adipose tissue releases glycerol and fatty acids into the circulation to be used as fuel. Thus, adipose tissue serves as a buffer for regulating lipid flux (3). Second, WAT releases a broad array of endocrine and paracrine factors generally referred to as adipokines (4). These adipokines interact with other organs such as the liver, skeletal muscle and hypothalamus to modulate a wide range of metabolic processes including inflammation, angiogenesis and food intake (4-8).

# 2.1.2 Adipose tissue as an endocrine organ

WAT is an endocrine organ secreting vast quantities of adipokines with potent metabolic effects (9). The number of identified adipokines has expanded rapidly and includes leptin, adiponectin, apelin, visfatin, resistin, vascular endothelial growth factor,

omentin, insulin-like growth factor 1, retinol binding protein-4, tumor necrosis factoralpha (TNF- $\alpha$ ), interleukin-6 (IL-6), and monocyte chemotactic protein-1 (9). Two of the best characterized adipokines are leptin and adiponectin.

Leptin was first described in 1994 as the product of the *ob* gene in obese *ob/ob* mice (10). Human obesity is characterized by high circulating leptin levels (11). The central effects of leptin primarily include inhibition of appetite (5). This is exerted mainly through its control on the hypothalamus and other specific regions of the central nervous system (5). The peripheral effects of leptin mainly include stimulating fatty acid oxidation in skeletal muscle (12), modulating the immune system such as increasing cytokine production and macrophage adhesion (13) and inducing angiogenesis in the vascular system (14).

Adiponectin was first described in 1995 and is exclusively secreted by adipocytes (15). Circulating adiponectin is composed of trimer, hexamer, and high-molecular weight (HMW) forms (16). Adiponectin has been attributed to play a number of beneficial roles in the body including improvement of insulin sensitivity (17-18) and vascular function (19), anti-atherogenic, anti-inflammatory actions (20) and cardioprotective effects (21).

# 2.1.3 Depot-specific phenotype in adipose tissue

Rodent models are commonly used for the investigations of mechanisms involved in adipose tissue metabolism. Generally, WAT in rodent models is classified anatomically as subcutaneous adipose tissue (SAT) or visceral adipose tissue (VAT). SAT is primarily located below the skin in the abdominal, gluteal and femoral areas, whereas VAT includes depots of fat deposited close to or even within the viscera of the abdominal cavity, such as gonadal, mesenteric, omental and retroperitoneal fat (22). It is

well documented that the metabolic features of WAT depend on its anatomical location in the body (23). Compared with SAT, VAT has greater basal and catecholamine-induced lipolysis and is less susceptible to the anti-lipolytic action of insulin (24-25). This is mainly owing to regional variation in the action of catecholamines and insulin, the lipolytic action of catecholamines being more effective and the antilipolytic effect of insulin being weaker in VAT than in SAT (26). VAT expresses higher glucocorticoid receptor levels (27) and 11β-hydroxysteroid dehydrogenase type 1, thus glucocorticoids have a greater impact on VAT than on SAT (28). VAT also has a higher rate of insulin stimulated glucose uptake than SAT (29) and exhibits lower capillary density and angiogenic capacity (30). VAT is richer in mitochondrial content (31-32) and has higher rates of lipid synthesis (33), which is in accordance with higher expression of lipogenesis related genes (34). Finally, there is higher macrophage infiltration into VAT during the progression of obesity (35). These fundamental differences among WAT depots might explain their differential influence on adipose tissue metabolism such as lipolysis, glucose uptake (26, 29, 36) and endocrine function (37).

Omental-mesenteric, gonadal and retroperitoneal adipose tissues are considered to be representative of VAT and are most commonly used in rodent metabolic studies. According to "the portal theory", an increased delivery of non-esterified fatty acids (NEFAs) from VAT to the liver via the portal vein, can cause liver insulin resistance, therefore visceral fat overaccumulation has a more negative impact on health than subcutaneous fat (38). However, it should be noted that only omental-mesenteric fat drains into the portal vein directly (39). Some findings from rodent studies suggest that SAT might actually have beneficial effects on metabolism. For example, peroxisome proliferator-activated receptor gamma (PPAR $\gamma$ ) agonists improve insulin sensitivity by redistributing VAT to SAT (40-41). Likewise, when SAT in donor mice was transplanted into the VAT of the recipient mice, mice showed improved metabolic characteristics (36). Thus, when studying adipose tissue metabolism, it is important to consider the existence of depot-specific effects.

#### 2.2 WAT Mitochondria

Mitochondria are energy-producing organelles and are the main sites of lipid and glucose oxidation. Mitochondrial biogenesis involves both translation of nuclear and mitochondrial genome-encoded proteins, followed by translocation of the nascent proteins across the mitochondrial membranes and proper assembly of these proteins within mitochondria (42). Although traditionally the role of mitochondria in WAT has been received little attention, accumulating evidence strongly support the importance of mitochondrial function for WAT biology and whole body energy homeostasis. The requirement for a large mitochondrial mass in white fat can be linked to key functions of the adipose tissue.

## 2.2.1 Mitochondria in WAT glyceroneogenesis

In conditions of increased metabolic demand such as during exercise and fasting, there is a marked rise in lipolytic hormones which stimulate the breakdown of TG molecules within fat cells. The liberated fatty acids are either oxidized within the mitochondria in adipocytes (<0.5% of the total), released into the circulation to be used as fuel source (~60-70%) or re-esterified back to TG (~30-40%) (43). The re-esterification of fatty acids is an energy consuming process and is assumed to be the largest single

consumer of ATP in adipocytes (44). Meanwhile, fatty acid re-esterification requires sufficient provision of glycerol-3-phosphate (G-3-P) to serve as the backbone of the TG molecule and at least in rodent adipose tissue, this occurs primarily through the glyceroneogenesis pathway. Thus, intact mitochondria are required for the provision of not only ATP, but glyceroneogenic enzymes as well. For instance, mitochondrial content has been hypothesized to control the availability of glyceroneogenic substrates (45). The majority of this thesis project focuses on WAT glyceroneogenesis, so it will be reviewed in more detail than other functions of the adipocytes.

# 2.2.1.1 Glyceroneogenesis pathway

Glyceroneogenesis was first defined by Reshef and colleagues in 1970 as the *de novo* synthesis of G-3-P from sources such as pyruvate, lactate or amino acids (46). It is essentially an abbreviated version of gluconeogenesis. Glyceroneogenesis is the major pathway whereby G-3-P is formed in WAT (47-49) for fatty acid re-esterification when blood glucose levels are low such as under fasting states. Two key regulatory enzymes involved in this process are pyruvate dehydrogenase kinase 4 (PDK4) and cytosolic isoform of phosophoenolpyruvate carboxykinase (PEPCK-C). PDK4 inactivates pyruvate dehydrogenase complexes leading to decreases in the formation of acetyl-CoA and a greater flux of pyruvate towards oxaloacetate (50). PEPCK-C catalyzes decarboxylation of oxaloacetate to form phosphoenolpyruvate. Phosphoenopyruvate is then converted into dihydroxyacetone phosphate, with the reduction of dihydroxyacetone phosphate into G-3-P via G-3-P dehydrogenase (47). The activation of PDK4 and PEPCK-C will favor the glyceroneogenic pathway in adipose tissue (50-52). A general description of glyceroneogenesis pathway in adipose tissue is shown in *Figure 2-1*.



#### Figure 2-1 General description of glyceroneogenesis and pyruvate routes towards fatty acid reesterification in adipose tissue.

When blood glucose levels become limiting, lipolysis releases glycerol and fatty acids from triglycerides (TG) stored in fat cells. Almost 100% of the glycerol is released because it is not phosphorylated significantly in adipose per se. In contrast, approximately 40% of released fatty acids are re-esterified back to TGs mainly through the glyceroneogenic pathway. Gluconeogenic substrates other than glucose and glycerol (i.e. pyruvate, lactate, alanine, and citric acid cycle anions) are converted into pyruvate first. Pyruvate can either be used in the tricarboxylic acid cycle after carboxylation by pyruvate dehydrogenase complexes (PDC) into acetyl-CoA or be favored towards glyceroneogenesis if carboxylated to oxaloacetate (OA) by pyruvate carboxylase (PC) in the mitochondria. Pyruvate dehydrogenase kinase 4 (PDK4) inactivates PDC, thereby regulating pyruvate availability for glyceroneogenesis in adipocytes. The cytosolic isoform of phosophoenolpyruvare carboxykinase (PEPCK-C) is involved in the conversion of OA to P-enolpyruvate, the precursor of glycerol-3-phosphate (G-3-P). Thus, the activation of PDK4 and PEPCK-C will favor the glyceroneogenic pathway in adipose tissue. Modified from Cadoudal et al., Diabetes, 2008.

#### 2.2.1.2 PDK4

# PDK4 and insulin resistance

On the one hand, PDK4 is postulated to be involved in the etiology of insulin resistance (53). Skeletal muscle PDK4 protein expression is selectively increased in insulin deficient (54-55) and insulin resistant states (56-57). PDK4 knockout (KO) mice are more glucose tolerant compared to wild type (WT) mice (58). In adipocytes, induction of PDK4 by prolactin resulted in decreased insulin-stimulated glucose uptake (59).

On the other hand, PDK4 has been thought to mediate a portion of the beneficial effects of thiazolidinediones (TZDs) on adipose tissue lipid metabolism. TZD treatment of adipocytes leads to increased expression of PDK4 concomitant with an increased rate of glyceroneogenesis (50). PDK4 appears to be essential for the upregulation of glyceroneogenesis by TZDs because the attenuation of PDK4 activity pharmacologically with leelamine or genetically by small interfering RNA (siRNA) both result in decreased fatty acid re-esterification (50).

#### **Regulation of PDK4 expression by exercise and fasting**

Pilegaard et al. (60) were the first to report induction of transcription of PDK4 by exercise in human skeletal muscle. They demonstrated that both five consecutive days of one-legged knee extensor exercise (60–90 min) and a single 4h low-intensity cycling bout induced marked increases in pdk4 mRNA in human skeletal muscle. Likewise, short-term high-intensity exercise can also activate pdk4 mRNA (61) in skeletal muscle. Similarly, starvation has been shown to induce pdk4 mRNA expression in both rodent and human skeletal muscle (62-63). Meanwhile, exercise and fasting have also been reported to
induce PDK4 in liver (64-66). Mechanistically, 5'AMP activated protein kinase (AMPK) and p38 mitogen-activated protein kinase (MAPK) signalling cascades have been implicated in the regulation of PDK4. The activation of AMPK has been postulated to either increase (67) or have no effect (68) on *pdk4* mRNA in lean, insulin-sensitive muscle or decrease (68) *pdk4* mRNA in obese and diabetic skeletal muscle. Dexamethasone has been shown to increase PDK4 activity in isolated cardiomyocytes through the phosphorylation and nuclear translocation of p38AMPK (69). The overexpression of the peroxisome proliferator-activated receptor gamma coactivator 1alpha (PGC-1 $\alpha$ ) in rodent skeletal muscle (70) and C2C12 muscle cells (71) results in increases in *pdk4* gene expression, suggesting that PGC-1 $\alpha$  might be a key regulator of PDK4. However, considering AMPK is a modulator of PGC-1 $\alpha$  (72-73) and the discrepant effects of AMPK activation on PDK4 in skeletal muscle (67-68), it is also possible that the regulation of PDK4 might be through a PGC-1 $\alpha$ -independent pathway.

#### 2.2.1.3 PEPCK

#### **PEPCK and insulin resistance**

Several lines of evidence suggest that dysregulation of adipocyte PEPCK expression could be an etiological factor in type 2 diabetes (T2DM) (74). First, pharmacologically, TZDs have been reported to induce *pepck* mRNA expression in rodent and human adipocytes and adipose tissue explants concomitant with an increased rate of fatty acid re-esterification and reduced release of fatty acid into the circulation (51-52, 75-76). Second, Olswang et al. (77) genetically engineered mice that have an ablation of PEPCK in WAT. These mice have reduced adipose tissue size and lipid content, and adipose tissue pieces from these mice released higher amounts of NEFAs.

Furthermore, approximately 25% of the mice have mild insulin resistance with notable lipodystrophy. Franckhauser et al. (78) created transgenic mice that overproduced PEPCK specifically in WAT. Although these mice were obese, they never became diabetic and appeared to have modestly increased sensitivity to insulin. However, the following findings from the same research group also demonstrated that PEPCK overexpression in WAT cause insulin resistance when challenged with a high fat diet (HFD) (79). It was suggested that under HFD conditions, the storage capacity of adipose tissue was overwhelmed, consequently resulting in ectopic fat deposition in liver and skeletal muscle, and leading to insulin resistance (79). Millward et al. (80) demonstrated that mice with reduced *pepck* mRNA in WAT develop systemic insulin resistance. Thus, most of the evidence suggests that dysregulation of PEPCK in adipose tissue might cause imbalance of whole body energy homeostasis.

# **Regulation of PEPCK by exercise and fasting**

The mRNA of *pepck* was upregulated by exercise and fasting in mouse liver (81-82). Glyceroneogenesis and *pepck* mRNA have been shown to be increased under fasting condition in rodent adipose tissue (46, 83-84). Further, it was demonstrated that the PEPCK metabolic response to fasting decreased as rats aged (84). Catecholamines might be one mediator for regulating PEPCK in adipose tissue as the infusion of epinephrine in rats resulted in a marked increase in glyceroneogenesis in epididymal adipose tissue (47). Mechanistically,  $\beta$ -2 adrenergic agonists have been reported to stimulate *pepck* mRNA expression through cAMP in adipocytes (85-86).

#### **Regulation of PEPCK by inflammation in WAT**

Long-term increases in inflammation (87-88) and in particular IL-6 (89), attenuates PEPCK in fat cells, which might contribute to the increased fatty acid mobilization under inflammatory conditions (89). Furthermore, PEPCK inhibition by inflammation leads to inhibition of glyceroneogenesis and adipogenic deficiency in adipocytes, which might be a mechanism contributing to the adipose dysfunction in obesity (88).

#### 2.2.2 Mitochondria in WAT *de novo* lipogenesis

De novo lipogenesis in WAT requires acetyl-CoA formed in the mitochondria by the oxidation of pyruvate. However, acetyl-CoA is not able to cross the mitochondrial membrane by itself. Acetyl-CoA and oxaloacetate combine via citrate synthase to form citrate, move to the cytosol and then are simultaneously converted back to acetyl-CoA and oxaloacetate by ATP citrate lyase (90). This shuttle provides the cytosolic acetyl-CoA necessary for *de novo* lipogenesis and requires ATP at several steps (90). To support the significant role played by mitochondria in *de novo* lipogenesis, Kaaman et al. (91) demonstrated that the mitochondrial DNA (mtDNA) number in isolated human subcutaneous adipocytes has a strong positive correlation with basal and insulin stimulated *de novo* lipogenesis. Meanwhile, inhibition of citrate export from the mitochondria at an early differentiation stage of 3T3-L1 adipocytes resulted in decreased accumulation of fat droplets (92). Taken together, the above evidence imply that mitochondrial dysfunction in adipocytes, is associated with a decreased capacity to store fatty acids as TG within adipocytes.

#### 2.2.3 Mitochondria in WAT adipogenesis

Adipogenesis is a biological process involving proliferation of preadipocytes from precursor cells, followed by differentiation into mature adipocytes (93). The acquisition of a mature adjocyte phenotype is characterized by changes in the activity of a number of transcription factors including cAMP response element-binding protein (CREB), CCAAT/enhancer binding protein (C/EBP) family members and PPARy, that directly Mitochondrial biogenesis has been considered as a influence adipogenesis (94). fundamental aspect of white adipose cell differentiation because mitochondrial proteins are generated at very high levels during the differentiation process (95-96). Adjogenesis is accompanied by changes in mitochondrial structure and biochemical composition, upregulation of components of the mtDNA replication and transcription machinery (93, 97), as well as increased oxygen consumption rate (98). The essential role of citrate export from the mitochondrial matrix to the cytosol for effective differentiation into adipocytes at the early adipogenic stage of 3T3-L1 cells has also been reported (92). Accordingly, chemical inhibition of mitochondrial complex machinery inhibits adipogenic differentiation (99).

## 2.2.4 Mitochondria in adiponectin synthesis

It has been reported that mitochondrial function is essential for adiponectin synthesis in adipocytes. Enhancement of mitochondrial biogenesis through rosiglitazone and over-expression of nuclear respiratory factor 1 (NRF-1) increased adiponectin synthesis whereas impairment of mitochondrial function through chemical uncouplers and mitochondrial transcription factor A (mtTFAM)-specific siRNA reduced adiponectin synthesis (100). Mechanistically, mitochondrial dysfunction decreases adiponectin synthesis through pathways involving endoplasmic reticulum (ER) stress, c-Jun N-terminal kinases (JNKs), and activating transcription factor 3 (ATF3) (100).

# 2.2.5 Adipose tissue mitochondrial dysfunction and insulin resistance

# 2.2.5.1 Evidence for adipose mitochondrial dysfunction in insulin resistance

Accumulating evidence from clinical observations, and animal and cell culture studies have substantiated a strong relationship between adipose tissue mitochondrial dysfunction and insulin resistance. Activities of mitochondrial respiratory chain complexes and mtDNA are decreased in the adipose tissue or adipocytes from different experimental models of obesity (101-103), subjects with morbid obesity and T2DM (104-108). Impairment of adipose tissue mitochondrial function in insulin resistant states might result from reductions in PGC-1. PGC-1 family members are key components regulating the respiratory chain and the biogenesis of mitochondria (109). For instance, PGC-1 $\alpha$ expression is reduced in adipose tissue from several insulin resistant rodent models including Zucker Diabetic Fatty (ZDF) rats, HFD rats, db/db, ob/ob and diet induced obese (DIO) mice (101, 103, 110-111) and in adipose tissues from individuals with insulin resistance or T2DM (91, 104, 112-113). PGC-1ß is suppressed in the inguinal fat in both DIO and db/db mice (103). Conversely, reversal of impaired mitochondrial function in adipose tissue could be accomplished through use of PPAR $\gamma$  agonist TZDs treatment (45, 103-104) and exercise (114). These alterations are concomitant with increases in whole body glucose homeostasis.

Whether impairment in adipose tissue mitochondrial function is a cause or consequence of impaired glucose homeostasis is still unclear. On the one hand, there is evidence suggesting that mitochondrial dysfunction precedes the development of impaired glucose homeostasis. A recent twin study suggests that a decline in mtDNA copy number and mitochondrial activity to catabolise branched chain amino acids may be induced upon development of obesity which precedes development of glucose intolerance (115). The mice with disruption of the insulin receptor knockout (FIRKO) specifically in adipose tissue are protected against obesity, as well as obesity-related glucose intolerance and other metabolic abnormalities. This phenotype could be related to the fact that FIRKO mice from 6 to 36 months of age have sustained high mRNA expression of the nuclear encoded mitochondrial genes involved in glycolysis and tricarboxylic acid cycle, as well as PGC-1 $\alpha$  and PGC-1 $\beta$  mRNAs (116). On the other hand, there is evidence to demonstrate that impairment of mitochondrial content may not be an initiating factor for insulin resistance, but appears to play a key role in the worsening of the disease condition. For instance, reduction of WAT mitochondrial content in Otsuka Long Evans Tokushima Fatty (OLETF) rats developed after overt diabetes (114). Similarly, reductions in mitochondrial content and PGC-1 $\alpha$  in HFD rats occurred after the onset of glucose intolerance (101).

# 2.2.5.2 Potential mechanisms by which impaired adipose mitochondrial function could lead to insulin resistance

First, as detailed above, dynamic mitochondrial activity is crucial for adipocyte function. Thus, impairments in adipose tissue mitochondrial function may alter whole body glucose homeostasis through its effects on glyceroneogenic, lipogenic, and adipogenic pathways. Of primary interest to this thesis research, the disorders of glyceroneogenic pathways would lead to increased NEFAs output into the blood. Increased circulating NEFAs can cause insulin resistance by interfering with the insulin signaling cascade mainly in the liver and skeletal muscle (117). Second, impaired mitochondrial function in adipose tissue can directly lead to insulin resistance in fat cells. For instance, siRNA mediated reduction of TFAM during adipocyte differentiation lead to impaired insulin-stimulated glucose uptake (118). Third, adiponectin provides a physiological link by which the energy status of adipose tissue may influence systemic insulin sensitivity.

# 2.3 Adipose Tissue Insulin Resistance

#### 2.3.1 Insulin action and signaling pathways in adipose tissue

The insulin signaling cascades in the whole body including adipose tissue comprise a highly integrated and complex network (119). Briefly, under normal physiological conditions, insulin secreted from the pancreas reaches its target tissues and interacts with insulin receptor (IR). The activated IR subsequently phosphorylates insulin receptor substrate (IRS) proteins and then activates two major signaling pathways: (1) phosphatidylinositol 3-kinase (PI3K)-Akt/protein kinase B (PKB) pathway, which is involved in most of the metabolic actions of insulin and (2) the Ras- mitogen-activated protein kinase (MAPK) pathway, which is implicated in most of the transcriptional and mitogenic effects of insulin (120). A schematic diagram of several central points of insulin signaling pathway and its action in WAT is shown below in *figure 2-2*.

There are at least 6 known substrates of IRS proteins expressed in insulinsensitive tissues with IRS-1 and -2 being the main IRS involved in insulin-mediated metabolic functions (121). IRS proteins are activated mainly through tyrosine (Tyr) phosphorylation (119). The PI3K cascade is activated by Tyr-phosphorylated IRS proteins and catalyzes the formation of phosphatidylinositol 3, 4, 5-trisphosphate (PIP<sub>3</sub>). PIP<sub>3</sub> stimulates phosphoinositide dependent kinase-1 and 2 (PDK1/2) activity, which in turn activates its downstream effectors: Akt/PKB, mammalian target of rapamycin (mTOR), p70S6 kinase (S6K1), as well as the atypical isoforms of PKC (aPKC $\varsigma/\lambda$ ) (122).

Akt/PKB is a key molecular player of insulin-stimulated metabolic pathways. The phosphorylation of Akt/PKB at both serine-473 and threonine-308 is responsible for regulating most of the PI3K-mediated metabolic function of insulin (119). Thus, Akt/PKB phosphorylation states are usually used to demonstrate the insulin stimulation status within a tissue. In adipose tissue, activation of Akt/PKB has the following two major physiological functions: 1) regulates insulin-stimulated glucose uptake via stimulating glucose transporter type 4 (GLUT4) vesicle translocation to the plasma membrane (123); and 2) mediates the phosphorylation and activation of phosphodiesterase 3B (PDE3B), PDE3B then catalyzes the hydrolysis of cyclic adenosine monophosphate (cAMP) to 5'AMP, thus leading to attenuated protein kinase A (PKA) activity and lipolysis (124-125). It should be noted, however, that insulin can also regulate lipolysis through an Akt independent signaling pathway (126).



Figure 2-2 A schematic diagram of several central points of insulin signaling pathway and its action in WAT.

Insulin binding to IR leads to the activation of IRS, activating PI3K and its downstream effector PIP<sub>3</sub>. PIP<sub>3</sub> stimulates PDK1/2 activity, which result in the activation of its downstream signaling proteins including Akt/PKB, mTOR, S6K1 and aPKC $\varsigma/\lambda$ . The phosphorylation of Akt/PKB is a key regulator of insulin signal transduction and has two main physiological functions in WAT. First, it can induce the translocation of GLUT4 from the intracellular compartment to the plasma membrane, largely accounting for increased insulin stimulated glucose uptake. Second, it can phosphorylate and activate PDE3B. The activation of PDE3B inactivates cAMP, resulting in attenuated PKA activity and lipolysis. +, stimulation or upregulation.

## 2.3.2 MAPKs in adipose tissue insulin resistance

A major cause of insulin resistance is the perturbation of insulin signaling system via phosphorylation of IR and IRS on certain serine/threonine residues, as well as the downstream signaling pathways (119). MAPKs are Ser/Thr specific protein kinases that mediate extracellular stimuli into a wide range of cellular responses. There are at least 4 subfamilies of MAPKs: the extracellular signal regulated kinases 1/2 (ERK1/2), ERK5, JNKs and p38MAPK (127). Of primary interest to this thesis, all except ERK5 have been

reported to be involved in the negative regulation of insulin signaling in adipose tissue in response to external stimuli such as NEFAs and inflammation (128-129).

An increased basal activity of ERK1/2, p38MAPK and JNKs has been reported in adipocytes from patients with T2DM (129). Activation of p38 MAPK/JNKs pathways has been proposed to functionally link visceral adiposity with whole body insulin resistance (130). Constitutive activation of ERK1/2, p38 or JNKs in 3T3-L1 adipocytes induces insulin resistance by affecting GLUT4 expression and insulin signaling proteins such as IRS, PI3K and Akt (131).

ERK1/2 is required for early adipocyte differentiation (132). Mice deficient in ERK1 have reduced adiposity and are resistant to diet-induced insulin resistance owing to decreased adipogenesis and elevated energy expenditure (133). Conversely, mice deficient in signaling adaptor p62, an ERK inhibitor, have high basal levels of ERK activity and develop characteristics of metabolic syndrome (134).

JNK1 is a negative modulator of insulin signaling, inhibition of which improves insulin sensitivity in insulin resistant rodent models (135). Mice lacking JNK1 specifically in adipose tissue protect against HFD-induced insulin resistance in the liver through a reduction in IL-6 secretion from adipose tissue (136). The phosphorylation of IRS-1 at serine 307 is an important mechanism by which the activation of JNK can inhibit insulin signaling in adipocytes (137).

In contrast to ERK1/2 and JNK, the role of p38 MAPK in insulin sensitivity *in vivo* has not been established yet. Meanwhile, the role of p38 MAPK in regulating glucose disposal in adipocytes *in vitro* is still in debate. On the one hand, there are studies suggesting that p38 MAPK plays an important role in insulin-stimulated glucose uptake

(138-139). On the other hand, there are also evidence supporting that p38 MAPK is not involved in insulin-stimulated glucose disposal in adipocytes (140-141).

# 2.3.3 Adiponectin in adipose tissue insulin resistance

#### 2.3.3.1 Adiponectin and adipose tissue biology

Although most studies have been focusing on exploring the endocrine properties of adiponectin in its target tissues such as the skeletal muscle and liver, an increasing body of evidence begin to suggest that adiponectin plays an important role for regulating adipose tissue proper.

Globular adiponectin enhances both basal and insulin-stimulated glucose uptake in isolated rat adipocytes through an AMPK dependent signaling pathway (142). Overexpression of adiponectin in 3T3-L1 fibroblasts has been shown to result in increased adipocyte differentiation and lipid accumulation (143), whereas others reported adiponectin disrupted adipogenesis in 3T3-F442A cells (144). Recently, adiponectin has been reported to inhibit spontaneous and catecholamine-induced lipolysis in human adipocytes from SAT of non-obese individuals, while this effect seems absent in obese subjects (145). Similarly, adiponectin suppresses basal and catecholamine-stimulated lipolysis in mouse adipoytes (146).

Overexpression of adiponectin specifically in WAT of FVB mice resulted in decreased adipose mass and impaired adipocyte differentiation (144). Adiponectin transgenic mice through a dominant mutation in the collagenous domain of adiponectin develop a phenotype that resembles the effects of chronic exposure to  $\beta$ 3 adrenergic treatment (147). Specifically, these mice have increased sensitivity to adrenergic stimuli,

increased mitochondrial density in adipocytes, smaller adipocyte size and upregulation of mRNAs involved in lipid storage (147). Thus, it has been proposed by Asterholm et al. (147) that adiponectin plays a crucial role in maintaining the metabolic flexibility of adipose tissue, making adipose tissue more sensitive to metabolic challenges. Meanwhile, the adipose tissues of these transgenic mice have increased expression of genes involved in oxidative phosphorylation, as well as mitochondrial markers including PRD1-BF1-RIZ1 homologous domain containing 16 (PRDM16) and cell death-inducing DNA fragmentation factor- $\alpha$ -like effecter A (Cidea) (148).

Both adipoR1 and adipoR2 are expressed in human and murine adipocytes (149-150), suggesting that adiponectin has autocrine/paracrine effects on adipose tissue. Further studies are required to elucidate the physiological roles of adiponectin in adipose tissue and its potential involvement in adipose tissue insulin resistance under metabolic challenges such as HFD.

# 2.3.3.2 Regulation of adiponectin in adipose tissue

Adiponectin is initially synthesized as a single polypeptide of 30 kDa and is then assembled in the ER, primarily into 3 forms: trimer, hexamer and HMW multimer (151). The different isoforms of adiponectin have been shown to play distinct biological roles (152-154). The expression, assembly and secretion of adiponectin are regulated at both transcriptional and post-translational level (155). PPAR $\gamma$  is a major player in these processes.

At the transcriptional level, PPAR $\gamma$  may increase adiponectin mRNA directly through its action on adiponectin gene promoter. A functional PPAR $\gamma$ -recognizing peroxisome proliferator hormone response elements (PPRE) site has been identified both in the human (156) and mouse (157) adiponectin promoters. Point mutations at this site resulted in reduced basal and TZD-induced transactivation of adiponectin promoter (156). At the post-translational level, PPAR $\gamma$  induces the transcription of genes such as Erol-L $\alpha$ , DsbA-L and ERp44, which encode proteins involved in adiponectin assembly and secretion in the ER (151, 158-159).

#### 2.4 Response of WAT to Exercise

#### 2.4.1 Exercise-induced adipose tissue mitochondrial biogenesis

More than 2 decades ago, Stallknecht et al. (160) demonstrated that exercise training (12 weeks of daily swimming, 6 h per day) increased mitochondrial enzyme activity in the epididymal adipose tissue of rats. The recent results from our research group confirmed this. Sutherland et al. (161) reported that swim training for 4 weeks (2h per day) increased the activity of citrate synthase and mitochondrial respiratory chain enzyme content in epididymal and retroperitoneal fat from rats. Meanwhile, both exercise training and an acute bout of exercise increased PGC-1 $\alpha$  mRNA expression. Similarly, Xu et al. (162) reported that treadmill running increased mitochondrial density in mouse epididymal fat by transmission electron microscopy of *in situ* mitochondria. Collectively, the effects of exercise on adipose tissue mitochondrial biogenesis are not exercise type specific because they are observed not only in swim training but also in treadmill running. Exercise induced increases in adipose tissue mitochondrial enzymes and activity further demonstrates the plasticity of adipose tissue in response to exercise.

However, the evidence from human studies is conflicting. Ruschke et al. (163) reported that 4 weeks of training led to increases in PGC-1 $\alpha$  mRNA expression only in

human subcutaneous but not omental fat. Camera et al. (164) reported that short-term endurance training did not alter citrate synthase activity, mitochondrial volume or PGC- $1\alpha$  mRNA expression in human subcutaneous adipose tissue although training-induced adaptations in skeletal muscle were observed. As has been discussed in Section1.1.3, subcutaneous fat is less sensitive to adrenergic stimuli than visceral fat. Catecholamines are one of the critical regulators of exercise-induced adipose tissue metabolism (165), thus depot-specific responsiveness to catecholamines might be one contributing factor for the discrepant results from human studies. Meanwhile, the duration and intensity of exercise, as well as the type of exercise might also be important factors for determining exercise-induced mitochondrial biogenesis in WAT.

## 2.4.2 Exercise-induced adipose tissue lipolysis

During endurance exercise, when carbohydrate reserves are depleted, the mobilization of NEFAs in adipose tissue becomes the main source for energy for the working muscle. The elevated circulating levels of catecholamines are one of the main regulators of adipose tissue lipolysis during exercise. When catecholamines, the hormone adrenaline and the neurotransmitter noradrenaline, bind  $\beta$ -adrenergic receptors, which are seven-transmembrane domain receptors coupled to stimulatory G proteins, there is an activation of adenylyl cyclase leading to a rise in intracellular cAMP levels and the activation of PKA. Activated PKA phosphorylates the lipid droplet protein perilipin and hormone sensitive lipase (HSL), thereby promoting hydrolysis of TG (166-169). HSL is the rate-limiting step for the liberation of fatty acids from adipose tissue into the circulation in this process (170). Reversible phosphorylation at several sites is a hallmark of HSL regulation. It has been demonstrated that Ser-563, Ser-659, and Ser-660 are major

PKA phosphorylation sites responsible for stimulating HSL with Ser-659 and Ser-660 as the major activity controlling sites (171-172).

For a long time, HSL has been viewed as the sole rate-limiting lipase for TG hydrolysis. In 2004, three research groups identified a new enzyme and independently named it adipose triglyceride lipase (ATGL) (173), desnutrin (174) and calciumindependent phospholipase A<sub>2</sub>zeta respectively. In 3T3-L1 adipocytes, ATGL can markedly enhance both basal and catecholamine-stimulated lipolysis (173). ATGLdeficient mice have reduced isoproterenol stimulated lipolysis (175). Thus, it seems likely that ATGL is also involved in catecholamine-stimulated lipolysis, however this is still controversial. Two recent findings indicate that ATGL might be a critical regulator of exercise-induced lipolysis. First, Huijsman et al. (176) demonstrated that ATGL deficient mice have impaired running ability and reduced plasma NEFAs availability (176). Second, Wohlers et al. (177) further showed that a single bout of treadmill running significantly increased ATGL content in omental/mesenteric fat of female mice (177). Owing to the insufficient time since its discovery, further studies are required to elucidate the signaling pathways for exercise-induced activation of ATGL in adipose tissue.

# 2.4.3 Effects of exercise on adipokines

## 2.4.3.1 Adiponectin

Circulating adiponectin is not affected (178-179) or even decreased (180) in response to exercise in healthy subjects with normal weight; while in obese and overweight subjects, exercise, independently (181-183), or combined with diet-induced weight loss (184), significantly increased plasma adiponectin levels. Generally, it was considered that the exercise-induced modification of adiponectin is to a large extent determined by changes in body composition (185). Meanwhile, exercise training intensity might also be a factor for influencing plasma adiponectin level, at least in rodent models (186).

# 2.4.3.2 Leptin

Training has disparate effects concerning leptin concentration with both decreased (187-188) or no alterations (189) having been reported. Generally, plasma leptin concentrations would be decreased if exercise training results in reduced fat mass (190).

# 2.4.4 Exercise on glucose uptake in adipose tissue

As early as two decades ago, it was reported that 10 weeks of exercise training normalized impaired adipose cell glucose metabolism from streptozocin-treated rats (191). Mechanistically, exercise training increased total GLUT4 protein in adipose cells of rats and adipose tissue of patients with T2DM (192-193). Similarly, exercise training has been reported to increase the response of WAT to insulin through improved activation of IRS/PI3K/Akt/PKB pathway in isolated adipocytes of rats (194). Taken together, the increase in exercise-mediated glucose uptake through increased IRS/PI3K/Akt/PKB/GLUT4 pathway in adipose tissue are at least partially responsible for the improvement in insulin sensitivity of T2DM following exercise training.

# 2.4.5 Mediators of exercise-induced adipose tissue metabolism

## 2.4.5.1 Catecholamines

As mentioned above, catecholamines increase during exercise and serve to stimulate adipose tissue lipolysis (168). Epinephrine increases the mRNA expression of

PGC-1 $\alpha$  and mitochondrial respiratory chain enzymes, and is considered to mediate at least partially exercise-induced adipose tissue mitochondrial biogenesis (161).

Increases in circulating catecholamine levels have been shown to activate multiple signalling pathways in adipose tissue. Of particular interest to this thesis research is the AMPK and p38 MAPK signalling pathways.  $\beta$ -adrenergic agonists have been reported to increase the activity of AMPK (165, 195-196) and p38 MAPK (197-199) in adipocytes. Studies from both rodent and humans have shown that acute exercise can cause AMPK activation in adipose tissue (165, 196, 200), however this is not a universal finding (201-202). Meanwhile, it has been confirmed that the effects of exercise on AMPK activity are mediated by catecholamines in the adipocytes of female rats (165). The duration and intensity of exercise, as well as the type of exercise might be important factors for influencing the phosphorylation status of AMPK in adipose tissue (201). Furthermore, whether there is gender specific difference in regards to exercise induced AMPK phosphorylation in adipose tissue also needs to be further elucidated.

It seems that the activation of lipolysis plays a major role in the activation of AMPK and p38 MAPK signalling pathways in adipose tissue. The activation of AMPK by catecholamine occurred as a consequence of lipolysis and subsequent increases in FA re-esterification (195). Similarly,  $\beta$ -adrenergic agonists-induced activation of p38 MAPK was attenuated when lipolysis was inhibited (199). Whether exercise and epinephrine can affect *pdk4* and *pepck* mRNA in WAT through AMPK and/or p38 MAPK signaling pathways in adipose tissue remains unknown.

## 2.4.5.2 Irisin

A recent study from Spiegelman's group has identified a novel peptide as a PGC-

1α dependent myokine that induces the differentiation of mouse subcutaneous white fat cells into a brown adipocytes phenotype during exercise (203). In this study, exercise and overexpression of PGC-1α specifically in muscle cells induced the expression of type I membrane protein fibronectin type III domain 5 (FNDC5) (203). FNDC5 was then cleaved and secreted into the circulation (203). The secreted protein product of FNDC5 was named irisin (after Iris, the Greek messenger goddess) (203). Irisin induced the expression of thermogenic genes in WAT such as uncoupling protein 1 (UCP1) and Cidea, partly through increased PPARα expression (203). It seems irisin is necessary for the exercise-induced thermogenic program in the WAT as injection of anti-FNDC5 reduced the effect of exercise on *ucp1* and *cidea* mRNA expression (203).

# 2.4.5.3 IL-6

Exercise is a robust perturbation that leads to marked changes in a wide range of hormonal factors. Skeletal muscle has been recently identified as an endocrine organ that secretes "myokines" (reviewed in (204)). One of the most extensively studied is IL-6. IL-6 secretion is increased from contracting skeletal muscle and is thought to be the primary contributor to the increased plasma IL-6 during exercise (205). Pedersen and colleagues have proposed that increases in muscle derived IL-6 plays a role in mediating exercise-associated metabolic changes and may be a potential mediator involved in exercise-induced adipose tissue metabolism (206).

On the one hand, it has been posited that IL-6 stimulates lipolysis in adipose tissue (207-208) and plays a role in regulating exercise-induced increases in adipose tissue AMPK activity (209-210). Conversely, the long-held view that muscle-derived IL-6 signals to adipose tissue during exercise is increasingly being challenged. For instance IL-6 infusion in healthy humans was shown to selectively stimulate lipolysis in skeletal muscle and have no effect on adipose tissue lipolysis (211). Similarly, IL-6 was found to not play a role in regulating subcutaneous adipose tissue AMPK activity during exercise (201). Recently, exercise in rats has been shown to induce the activation of IL-6 signalling in the liver and that IL-6 could contribute to increase hepatic glucose production by directly upregulating *pepck* mRNA expression (81). Of interest, previous work has demonstrated that long-term increases in inflammation (212-213), and in particular IL-6 (89), attenuate PEPCK expression in fat cells. These findings suggest that increases in IL-6 may serve to dampen the exercise-mediated induction of glyceroneogenic enzymes in adipose tissue. However, it should be noted that interstitial concentrations of IL-6 ranged from 2000 to 3000 pg/ml (214), the circulation concentration of IL-6 is around 35 pg/ml even after extremely large volumes of exercise such as a marathon (215). Therefore, interstitial concentrations of IL-6 are several orders of magnitude higher than levels in the circulation after exercise. Thus, it is of interest to see if IL-6 signalling is activated in adipose tissue during exercise considering the already high concentrations of IL-6 surrounding this tissue at rest. Meanwhile, the role of IL-6 in modulating the effects of exercise on glyceroneogenic enzyme expression in WAT and the reputed signalling pathways therein remains unknown.

# 2.4.5.4 The interactions between epinephrine and IL-6 during exercise

There are several lines of evidence suggesting that epinephrine increases IL-6 production when administered at supraphysiological concentrations (216-219).  $\beta$ -adrenergic activation via isoproterenol increases IL-6 at both mRNA and protein levels *in vitro* in human breast adipocytes (220). In contrast, there is evidence to suggest that

epinephrine is unlikely to mediate the increased IL-6 mRNA and protein release during exercise. An epinephrine infusion in resting humans, with the circulating epinephrine level comparable to that of exercise-induced increases in plasma epinephrine, only increased plasma IL-6 at a modest levels (about 6-fold), which is much lower than the plasma IL-6 levels observed during exercise (about 30-fold) (221).

## 2.4.5.5 The interactions between IL-6 and other myokines during exercise

During exercise, skeletal muscle secretes a wide range of myokines other than IL-6, such as myostatin, IL-7, murine chemokine (C-X-C motif) ligand 1 (CXCL1) and leukemia inhibitory factor (LIF) (222). For instance, an acute bout of exercise increased *cxcl1* mRNA expression specifically in skeletal muscle and liver, as well as serum CXCL1 level (223). However the above effects are completely blunted from IL-6deficient mice (223). Overexpression of IL-6 in muscle cells increased serum CXCL1 and liver *cxcl1* mRNA expression (223). Thus, it is suggested that exercise-induced CXCL1 in the liver is regulated by muscle-derived IL-6 (223). Although exercise-induced CXCL1 in adipose tissue has not been reported at present, we can't exclude this possibility considering the duration and intensity of exercise, as well as the type of exercise might be important factors for determining exercise induced metabolism in WAT. Furthermore, it is tempting to speculate that increases in skeletal muscle-derived IL-6 signal to the liver during exercise and consequently result in increases in CXCL1. CXCL1 signals to regulate adipose tissue metabolism. Clearly, further studies are required to elucidate the crosstalk between muscle/liver/adipose tissue during exercise with a special focus on newly discovered myokines, hepatokines and adipokines.

#### 2.5 IL-6 in WAT Metabolism

## 2.5.1 IL-6 production and IL-6 signaling in adipose tissue

It has been long known that circulating and adipose tissue IL-6 levels are increased in patients with obesity, insulin resistance and T2DM (224). Chronic low grade elevations of IL-6 in adipose tissue have been posited to correlate with the degree of insulin resistance (225-226). However, it has also been proposed that increased circulating levels of IL-6 are related to increased fat mass instead of insulin resistance (227). Adipocytes and other cells in adipose tissue including stromal cells and macrophages all contribute to IL-6 production (228-230). Meanwhile, fat depot differences in the IL-6 concentration and its metabolic effects on adipocyte metabolism have also been reported, with omental adipose tissue releasing 3 times more IL-6 than subcutaneous adipose tissue (229).

IL-6 transduces cellular signals through receptors that consist of two subunits: glycoprotein 130 (gp130), a non-ligand binding protein responsible for intracellular signal transduction, and the ligand-specific alpha chain, IL-6 receptor alpha (IL-6R $\alpha$ ) (231-232). IL-6R $\alpha$  can act in both its soluble and membrane bound forms (232). Heterodimers of IL-6, IL-6R $\alpha$ , and gp130 activate two signaling pathways: 1) Janus kinase (JAK)-signal transducer and activator of transcription 3 (STAT3) signaling pathways (231); and 2) the Src homology2-containing tyrosine phosphatise (SHP-2)/ ERK/ MAPK pathway (232). Phosphorylated STAT3 is translocated to the nucleus to promote the transcription of its target genes, such as suppressor of cytokine signaling 3 (SOCS3), which is highly induced by IL-6 at the transcriptional level (233). SOCS3

serves a negative feedback role on IL-6 signaling (234). SOCS3 may also negatively affect insulin signaling via IRS proteins (235).

#### 2.5.2 IL-6 and insulin resistance

Currently, the role of IL-6 in the determination of insulin sensitivity is still in debate (236-237). This thesis review has attempted to demonstrate that the role of IL-6 in insulin resistance is neither exclusively beneficial nor harmful and appears to be tissueand context-specific. We primarily focused on three main tissues involved in peripheral insulin resistance and glucose homeostasis: *in vitro* evidence from adipose tissue and *in vivo* evidence in humans and rodents mainly in liver, skeletal muscle and adipose tissue.

# 2.5.2.1 In vitro effects of IL-6 on adipocyte metabolism

On the one hand, there is evidence pointing to a role of IL-6 negatively regulating insulin signaling in adipocytes. In 3T3-L1 adipocytes, IL-6 decreases adiponectin gene expression and secretion in a dose- and time-dependent manner (238), thus leading to reduced insulin sensitivity. Similarly, IL-6 reduces the mRNA expression of *irs-1*, *glut4*, and PPAR $\gamma$ , IRS-1 protein and tyrosine phosphorylation, decreases insulin-stimulated glucose uptake and lipogenesis, as well as insulin-induced activation of IR- $\beta$ , Akt/PKB, and ERK1/2 (228, 239). SOCS3 expression, a potential inhibitor of insulin signaling, is increased in adipocyte cell lines exposed to IL-6 *in vitro* (228, 240).

On the other hand, the effects of IL-6 on basal and insulin-stimulated glucose uptake are inconclusive. Stouthard et al. (241) demonstrated that treating 3T3-L1 adipocytes with recombinant murine IL-6 increased basal glucose transport, whereas others found no effect (220, 239, 242). Meanwhile, IL-6 has been reported to either

enhance (241), suppress (228, 239) or have no effect (220) on insulin-stimulated glucose uptake in 3T3-L1 adipocytes. A summary of the controversial reports regarding IL-6 effects on adipocyte metabolism is shown in *table 2-1*.

Reference	In vitro effects of IL-6 on adipocyte metabolism
Fasshauer et al., 2003	Decreases adiponectin gene expression and secretion in 3T3-
(238)	L1 adipocytes
Lagathu et al., 2003	Negatively affects insulin signaling pathways in 3T3-L1
(228); Rotter et al., 2003	adipocytes and human fat cells
(239)	
Stouthard et al., 1996	Increases basal and insulin stimulated glucose uptake in 3T3-
(241)	L1 adipocytes
Path et al., 2001 (220);	No effects on basal and insulin stimulated glucose uptake in
Huppertz, 1996 (242)	3T3-L1 adipocytes
Rotter et al., 2003 (239)	No effects on basal glucose uptake while suppresses insulin
	stimulated glucose uptake in 3T3-L1 adipocytes

Table 2-1 In Vitro Effects of IL-6 on Adipocyte Metabolism

In summary, studies investigating the role of IL-6 in insulin action in adipocytes *in vitro* have mainly looked at short-term effects of high IL-6 levels and the results are conflicting. In order to mimic the situation seen in individuals with obesity or T2DM, studies of the effect of long-term low-grade IL-6 challenge of primary cultured adipocytes and/or cultured adipose tissue on insulin sensitivity are warranted.

2.5.2.2 In vivo effects of IL-6 on whole body glucose homeostasis and lipid metabolism

The effects of IL-6 on whole body glucose homeostasis and lipid metabolism have been studied both in humans and rodents. The results are contradictory, with improvement, no effect and deterioration having been reported. Most of the data available *in vivo* in humans are related to the acute effects of IL-6. An acute increase in plasma IL-6 levels through recombinant human (rh) IL-6 infusion has been reported to increase (243-245), decrease (246-247) or not affect (207, 211, 248) plasma glucose levels. Of

note, the increased plasma glucose levels may be caused by concomitant increases in other hormones such as epinephrine and glucagon owing to the supra-physiological concentration of plasma IL-6 (243-244). Infusion of IL-6 at a physiological concentration inhibits endotoxin-induced TNF- $\alpha$  production (249), increases SAT glucose uptake (246) and enhances the glucose infusion rate (248), indicating that IL-6 might improve insulin sensitivity in vivo in humans. Meanwhile, functional lack of IL-6 in patients with rheumatoid arthritis through IL-6 receptor-inhibiting monoclonal antibody leads to increased plasma lipid and glucose levels, arguing against IL-6 as an insulin resistance agent in humans in vivo (250-251). Infusion of IL-6 has also been reported to increase fatty acid rate of appearance and oxidation (243, 252), systemic lipolysis (211, 252) and hsl mRNA in adipose tissue of T2DM (247). It was recently confirmed that rhIL-6 infusion stimulated lipolysis in skeletal muscle instead of adipose tissue could account for the systemic changes in fatty acid metabolism (211). However, there is also a report suggesting that infusion of rhIL-6 has no effect on lipolysis during low intensity exercise (253).

*In vivo* studies in rodents suggest that IL-6 has tissue-specific effects, promoting (254-258) or having no effect (259) on insulin resistance in liver. Similarly, contradictory results with negative (256-257), neutral (255, 259) and positive (260-261) effects of IL-6 on skeletal muscle insulin sensitivity have been reported. Intracerebroventricular administration of IL-6 decreases adipose tissue mass in rats (262), indicating that IL-6 might be beneficial for adipose tissue metabolism *in vivo*.

#### 2.5.3 Effects of IL-6 on metabolism in genetically modified mouse model

Mice lacking the gene encoding IL-6 (IL-6<sup>-/-</sup>) develop characteristic features of the metabolic syndrome such as mature-onset obesity and insulin resistance (263-265), but this is not a universal finding (266). The increased adiposity and insulin resistance were partly reversed following chronic central administration of IL-6, highlighting the importance of centrally mediated effects of IL-6 (262). IL- $6^{-/-}$  mice also have reduced fat utilization at room temperature (264), and impaired increases in energy expenditure when challenged with cold exposure or new cage stress (267). Meanwhile, IL-6<sup>-/-</sup> mice have decreased endurance and energy expenditure during exercise (264). Hepatocyte-specific IL-6R $\alpha$  deficient mice have hepatic inflammation and systemic insulin resistance (268). Transgenic  $ob/ob^{\text{IL-6}}$  mice generated by mating  $ob^{+/-}/\text{IL6}^{\text{tg}}$  mice with  $ob^{+/-}$  mice were more insulin sensitive than *ob/ob* mice under chow and HFD, and responsive to central leptin action, suggesting that rhIL-6 can promote systemic insulin sensitivity in mice (269). Whereas, overexpression of IL-6 in skeletal muscle by an electro-transfer procedure resulted in hypoadiponectaemia, liver inflammation, hyperinsulinaemia, hypoglycemia and reduced body fat (257).

To sum up, the effects of IL-6 on whole body energy homeostasis have been mainly focusing on the liver and skeletal muscle, with a striking paucity of data available in adipose tissue. Genetically-modified mouse models have shed new light on the metabolic action of IL-6 in the whole body with contradictory results being reported. Studies of the effect of IL-6 action on adipose tissue metabolism by utilizing IL-6 infusion *in vivo* and genetic mouse models are further required.

# 2.6 Thesis Overview

Thiazolidinediones (TZDs) are a commonly prescribed class of insulin sensitizing drugs that are thought to mediate a large portion of their beneficial effects through their up-regulation of the glyceroneogenic pathway in WAT (50, 270). Although there is an increasing body of evidence that WAT might play an important role in regulating the beneficial effects of exercise on systemic metabolism, it is not known if exercise can mimic TZD effects on PEPCK and PDK4 mRNA expression in adipose tissue. Furthermore, catecholamines are central modulators for regulating exercise-induced effects on WAT metabolism such as inducing lipolysis (168) and mitochondrial biogenesis (161). Skeletal muscle has been identified as an endocrine organ that secretes "myokines" with one of the best characterized being interleukin 6 (IL-6) (204). It has been proposed that increases in muscle derived IL-6 may be a potential mediator involved in exercise-induced WAT metabolism (271). It is not known if epinephrine and/or IL-6 play roles in the regulation of exercise-induced PEPCK and PDK4 mRNA expression and the signaling pathways that might be involved.

Currently, the role of IL-6 in the development of systemic insulin resistance is still under debate (272). Interstitial concentrations of IL-6 surrounding adipose tissue are orders of magnitude greater than levels in the circulation (214) suggesting that adipose tissue derived IL-6 could play a pivotal role in the maintenance of adipose tissue metabolism and function in an autocrine and/or paracrine manner. Although evidence from *in vitro* studies suggests that IL-6 causes insulin resistance in adipose tissue (228, 238-239), there is also evidence that IL-6 may favorably regulate adipose tissue function presumably through its modulation of AMPK (209). Although not a universal finding

(266), IL-6<sup>-/-</sup> mice have been reported to have impaired glucose homeostasis when fed a chow diet (263). Likewise, IL-6<sup>-/-</sup> mice developed hepatic inflammation and insulin resistance under HFD feeding conditions (273). It is not known if there is phenotypic differences in adipose tissue between age-matched WT and IL-6<sup>-/-</sup> mice under chow and HFD fed conditions.

Consequently, this thesis has been designed to:

1) determine the regulation of PDK4 mRNA expression in epididymal adipose tissue (eWAT) from lean rats by swimming and epinephrine and the potential signalling pathways involved.

2) determine if 2 hours of swimming exercise and acute increases in epinephrine *in vivo* can induce PDK4 mRNA expression in eWAT from insulin resistant rats and if the reputed signaling pathways mediating this effect, i.e. p38MAPK are intact.

3) determine a) the effects of *ex vivo* IL-6 treatment in mouse eWAT on PEPCK and PDK4 expression; b) if IL-6 signaling is activated and if PEPCK and PDK4 mRNA expression is increased in mouse eWAT immediately after an acute bout of treadmill running; and c) through the utilization of IL- $6^{-/-}$  mouse model, to elucidate the role of IL-6 in modulating the effects of treadmill running on PEPCK and PDK4 mRNA expression in mouse eWAT and the putative signaling pathways therein.

4) investigate the metabolic action of IL-6 on rodent adipose tissue by characterizing the adipose tissue phenotypes from age matched wild type and IL- $6^{-/-}$  mice under both chow and HFD conditions.

# 2.7 Literature Cited

1. Rosen ED, Spiegelman BM. Adipocytes as regulators of energy balance and glucose homeostasis. Nature. 2006;444(7121):847-53.

Ahima RS. Adipose tissue as an endocrine organ. Obesity (Silver Spring).
2006;14 Suppl 5:242S-9S.

3. Frayn KN. Adipose tissue as a buffer for daily lipid flux. Diabetologia. 2002;45(9):1201-10.

4. Trayhurn P, Wood IS. Adipokines: inflammation and the pleiotropic role of white adipose tissue. Br J Nutr. 2004;92(3):347-55.

Flier JS. Obesity wars: molecular progress confronts an expanding epidemic. Cell.
2004;116(2):337-50.

6. Sarkanen JR, Kaila V, Mannerstrom B, Raty S, Kuokkanen H, Miettinen S, et al. Human adipose tissue extract induces angiogenesis and adipogenesis in vitro. Tissue Eng Part A. 2012;18(1-2):17-25.

7. Bays HE, Gonzalez-Campoy JM, Bray GA, Kitabchi AE, Bergman DA, Schorr AB, et al. Pathogenic potential of adipose tissue and metabolic consequences of adipocyte hypertrophy and increased visceral adiposity. Expert Rev Cardiovasc Ther. 2008;6(3):343-68.

8. Baldock P. Reciprocal regulation of bone and energy metabolism. Horm Res Paediatr. 2011;76 Suppl 1:7-11.

9. Harwood HJ, Jr. The adipocyte as an endocrine organ in the regulation of metabolic homeostasis. Neuropharmacology. 2011.

Zhang Y, Proenca R, Maffei M, Barone M, Leopold L, Friedman JM. Positional cloning of the mouse obese gene and its human homologue. Nature. 1994;372(6505):425-32.

11. Fried SK, Ricci MR, Russell CD, Laferrere B. Regulation of leptin production in humans. J Nutr. 2000;130(12):3127S-31S.

12. Minokoshi Y, Kim YB, Peroni OD, Fryer LG, Muller C, Carling D, et al. Leptin stimulates fatty-acid oxidation by activating AMP-activated protein kinase. Nature. 2002;415(6869):339-43.

13. Fruhbeck G, Gomez-Ambrosi J, Muruzabal FJ, Burrell MA. The adipocyte: a model for integration of endocrine and metabolic signaling in energy metabolism regulation. Am J Physiol Endocrinol Metab. 2001;280(6):E827-47.

14. Sierra-Honigmann MR, Nath AK, Murakami C, Garcia-Cardena G, Papapetropoulos A, Sessa WC, et al. Biological action of leptin as an angiogenic factor. Science. 1998;281(5383):1683-6.

15. Scherer PE, Williams S, Fogliano M, Baldini G, Lodish HF. A novel serum protein similar to C1q, produced exclusively in adipocytes. J Biol Chem. 1995;270(45):26746-9.

 Schraw T, Wang ZV, Halberg N, Hawkins M, Scherer PE. Plasma adiponectin complexes have distinct biochemical characteristics. Endocrinology. 2008;149(5):2270-82.

17. Lindsay RS, Funahashi T, Hanson RL, Matsuzawa Y, Tanaka S, Tataranni PA, et al. Adiponectin and development of type 2 diabetes in the Pima Indian population. Lancet. 2002;360(9326):57-8.

18. Hajer GR, van der Graaf Y, Olijhoek JK, Edlinger M, Visseren FL. Low plasma levels of adiponectin are associated with low risk for future cardiovascular events in patients with clinical evident vascular disease. Am Heart J. 2007;154(4):750 e1-7.

19. Bussey CT, Kolka CM, Rattigan S, Richards SM. Adiponectin opposes endothelin-1-mediated vasoconstriction in the perfused rat hindlimb. Am J Physiol Heart Circ Physiol. 2011;301(1):H79-86.

20. Ouchi N, Kihara S, Arita Y, Okamoto Y, Maeda K, Kuriyama H, et al. Adiponectin, an adipocyte-derived plasma protein, inhibits endothelial NF-kappaB signaling through a cAMP-dependent pathway. Circulation. 2000;102(11):1296-301.

21. Gonon AT, Widegren U, Bulhak A, Salehzadeh F, Persson J, Sjoquist PO, et al. Adiponectin protects against myocardial ischaemia-reperfusion injury via AMP-activated protein kinase, Akt, and nitric oxide. Cardiovasc Res. 2008;78(1):116-22.

22. Casteilla L, Penicaud L, Cousin B, Calise D. Choosing an adipose tissue depot for sampling: factors in selection and depot specificity. Methods Mol Biol. 2008;456:23-38.

23. Ibrahim MM. Subcutaneous and visceral adipose tissue: structural and functional differences. Obes Rev. 2010;11(1):11-8.

24. Mauriege P, Galitzky J, Berlan M, Lafontan M. Heterogeneous distribution of beta and alpha-2 adrenoceptor binding sites in human fat cells from various fat deposits: functional consequences. Eur J Clin Invest. 1987;17(2):156-65.

25. Arner P. Differences in lipolysis between human subcutaneous and omental adipose tissues. Ann Med. 1995;27(4):435-8.

26. Arner P. Not all fat is alike. Lancet. 1998;351(9112):1301-2.

27. Rebuffe-Scrive M, Bronnegard M, Nilsson A, Eldh J, Gustafsson JA, Bjorntorp P. Steroid hormone receptors in human adipose tissues. J Clin Endocrinol Metab. 1990;71(5):1215-9.

28. Bujalska IJ, Kumar S, Stewart PM. Does central obesity reflect "Cushing's disease of the omentum"? Lancet. 1997;349(9060):1210-3.

29. Virtanen KA, Lonnroth P, Parkkola R, Peltoniemi P, Asola M, Viljanen T, et al. Glucose uptake and perfusion in subcutaneous and visceral adipose tissue during insulin stimulation in nonobese and obese humans. J Clin Endocrinol Metab. 2002;87(8):3902-10.

30. Gealekman O, Guseva N, Hartigan C, Apotheker S, Gorgoglione M, Gurav K, et al. Depot-specific differences and insufficient subcutaneous adipose tissue angiogenesis in human obesity. Circulation. 2011;123(2):186-94.

31. Deveaud C, Beauvoit B, Salin B, Schaeffer J, Rigoulet M. Regional differences in oxidative capacity of rat white adipose tissue are linked to the mitochondrial content of mature adipocytes. Mol Cell Biochem. 2004;267(1-2):157-66.

32. Okamoto Y, Higashiyama H, Rong JX, McVey MJ, Kinoshita M, Asano S, et al. Comparison of mitochondrial and macrophage content between subcutaneous and visceral fat in db/db mice. Exp Mol Pathol. 2007;83(1):73-83.

33. Jamdar SC. Glycerolipid biosynthesis in rat adipose tissue. Influence of adiposecell size and site of adipose tissue on triacylglycerol formation in lean and obese rats. Biochem J. 1978;170(1):153-60.

34. Palou M, Priego T, Sanchez J, Rodriguez AM, Palou A, Pico C. Gene expression patterns in visceral and subcutaneous adipose depots in rats are linked to their morphologic features. Cell Physiol Biochem. 2009;24(5-6):547-56.

35. Cancello R, Tordjman J, Poitou C, Guilhem G, Bouillot JL, Hugol D, et al. Increased infiltration of macrophages in omental adipose tissue is associated with marked hepatic lesions in morbid human obesity. Diabetes. 2006;55(6):1554-61.

36. Tran TT, Yamamoto Y, Gesta S, Kahn CR. Beneficial effects of subcutaneous fat transplantation on metabolism. Cell Metab. 2008;7(5):410-20.

37. Wajchenberg BL, Giannella-Neto D, da Silva ME, Santos RF. Depot-specific hormonal characteristics of subcutaneous and visceral adipose tissue and their relation to the metabolic syndrome. Horm Metab Res. 2002;34(11-12):616-21.

38. Kabir M, Catalano KJ, Ananthnarayan S, Kim SP, Van Citters GW, Dea MK, et al. Molecular evidence supporting the portal theory: a causative link between visceral adiposity and hepatic insulin resistance. Am J Physiol Endocrinol Metab. 2005;288(2):E454-61.

39. Rytka JM, Wueest S, Schoenle EJ, Konrad D. The portal theory supported by venous drainage-selective fat transplantation. Diabetes. 2011;60(1):56-63.

40. Miyazaki Y, Mahankali A, Matsuda M, Mahankali S, Hardies J, Cusi K, et al. Effect of pioglitazone on abdominal fat distribution and insulin sensitivity in type 2 diabetic patients. J Clin Endocrinol Metab. 2002;87(6):2784-91.

41. Laplante M, Festuccia WT, Soucy G, Gelinas Y, Lalonde J, Berger JP, et al. Mechanisms of the depot specificity of peroxisome proliferator-activated receptor gamma action on adipose tissue metabolism. Diabetes. 2006;55(10):2771-8. 42. Hood DA, Adhihetty PJ, Colavecchia M, Gordon JW, Irrcher I, Joseph AM, et al. Mitochondrial biogenesis and the role of the protein import pathway. Med Sci Sports Exerc. 2003;35(1):86-94.

43. Wang T, Zang Y, Ling W, Corkey BE, Guo W. Metabolic partitioning of endogenous fatty acid in adipocytes. Obes Res. 2003;11(7):880-7.

44. Rognstad R, Katz J. The balance of pyridine nucleotides and ATP in adipose tissue. Proc Natl Acad Sci U S A. 1966;55(5):1148-56.

45. Wilson-Fritch L, Nicoloro S, Chouinard M, Lazar MA, Chui PC, Leszyk J, et al. Mitochondrial remodeling in adipose tissue associated with obesity and treatment with rosiglitazone. J Clin Invest. 2004;114(9):1281-9.

46. Reshef L, Hanson RW, Ballard FJ. A possible physiological role for glyceroneogenesis in rat adipose tissue. J Biol Chem. 1970;245(22):5979-84.

47. Nye CK, Hanson RW, Kalhan SC. Glyceroneogenesis is the dominant pathway for triglyceride glycerol synthesis in vivo in the rat. J Biol Chem. 2008;283(41):27565-74.

48. Forest C, Tordjman J, Glorian M, Duplus E, Chauvet G, Quette J, et al. Fatty acid recycling in adipocytes: a role for glyceroneogenesis and phosphoenolpyruvate carboxykinase. Biochem Soc Trans. 2003;31(Pt 6):1125-9.

49. Reshef L, Olswang Y, Cassuto H, Blum B, Croniger CM, Kalhan SC, et al. Glyceroneogenesis and the triglyceride/fatty acid cycle. J Biol Chem. 2003;278(33):30413-6.

50. Cadoudal T, Distel E, Durant S, Fouque F, Blouin JM, Collinet M, et al. Pyruvate dehydrogenase kinase 4: regulation by thiazolidinediones and implication in glyceroneogenesis in adipose tissue. Diabetes. 2008;57(9):2272-9.

51. Duplus E, Benelli C, Reis AF, Fouque F, Velho G, Forest C. Expression of phosphoenolpyruvate carboxykinase gene in human adipose tissue: induction by rosiglitazone and genetic analyses of the adipocyte-specific region of the promoter in type 2 diabetes. Biochimie. 2003;85(12):1257-64.

52. Tordjman J, Chauvet G, Quette J, Beale EG, Forest C, Antoine B. Thiazolidinediones block fatty acid release by inducing glyceroneogenesis in fat cells. J Biol Chem. 2003;278(21):18785-90.

53. Sugden MC. PDK4: A factor in fatness? Obes Res. 2003;11(2):167-9.

54. Wu P, Inskeep K, Bowker-Kinley MM, Popov KM, Harris RA. Mechanism responsible for inactivation of skeletal muscle pyruvate dehydrogenase complex in starvation and diabetes. Diabetes. 1999;48(8):1593-9.

55. Sugden MC, Kraus A, Harris RA, Holness MJ. Fibre-type specific modification of the activity and regulation of skeletal muscle pyruvate dehydrogenase kinase (PDK) by prolonged starvation and refeeding is associated with targeted regulation of PDK isoenzyme 4 expression. The Biochemical journal. 2000;346 Pt 3:651-7.

56. Holness MJ, Kraus A, Harris RA, Sugden MC. Targeted upregulation of pyruvate dehydrogenase kinase (PDK)-4 in slow-twitch skeletal muscle underlies the stable modification of the regulatory characteristics of PDK induced by high-fat feeding. Diabetes. 2000;49(5):775-81.

57. Majer M, Popov KM, Harris RA, Bogardus C, Prochazka M. Insulin downregulates pyruvate dehydrogenase kinase (PDK) mRNA: potential mechanism contributing to increased lipid oxidation in insulin-resistant subjects. Mol Genet Metab. 1998;65(2):181-6.

58. Jeoung NH, Wu P, Joshi MA, Jaskiewicz J, Bock CB, Depaoli-Roach AA, et al. Role of pyruvate dehydrogenase kinase isoenzyme 4 (PDHK4) in glucose homoeostasis during starvation. The Biochemical journal. 2006;397(3):417-25.

59. White UA, Coulter AA, Miles TK, Stephens JM. The STAT5A-mediated induction of pyruvate dehydrogenase kinase 4 expression by prolactin or growth hormone in adipocytes. Diabetes. 2007;56(6):1623-9.

60. Pilegaard H, Keller C, Steensberg A, Helge JW, Pedersen BK, Saltin B, et al. Influence of pre-exercise muscle glycogen content on exercise-induced transcriptional regulation of metabolic genes. The Journal of physiology. 2002;541(Pt 1):261-71.

61. Nordsborg N, Bangsbo J, Pilegaard H. Effect of high-intensity training on exercise-induced gene expression specific to ion homeostasis and metabolism. J Appl Physiol. 2003;95(3):1201-6.

62. Wu P, Sato J, Zhao Y, Jaskiewicz J, Popov KM, Harris RA. Starvation and diabetes increase the amount of pyruvate dehydrogenase kinase isoenzyme 4 in rat heart. Biochem J. 1998;329 (Pt 1):197-201.

63. Peters SJ, Harris RA, Heigenhauser GJ, Spriet LL. Muscle fiber type comparison of PDH kinase activity and isoform expression in fed and fasted rats. Am J Physiol Regul Integr Comp Physiol. 2001;280(3):R661-8.

64. Hoene M, Lehmann R, Hennige AM, Pohl AK, Haring HU, Schleicher ED, et al. Acute regulation of metabolic genes and insulin receptor substrates in the liver of mice by one single bout of treadmill exercise. J Physiol. 2009;587(Pt 1):241-52.

65. Palou M, Priego T, Sanchez J, Villegas E, Rodriguez AM, Palou A, et al. Sequential changes in the expression of genes involved in lipid metabolism in adipose tissue and liver in response to fasting. Pflugers Arch. 2008;456(5):825-36.

66. Holness MJ, Bulmer K, Smith ND, Sugden MC. Investigation of potential mechanisms regulating protein expression of hepatic pyruvate dehydrogenase kinase isoforms 2 and 4 by fatty acids and thyroid hormone. Biochem J. 2003;369(Pt 3):687-95.

67. Jorgensen SB, Wojtaszewski JF, Viollet B, Andreelli F, Birk JB, Hellsten Y, et al. Effects of alpha-AMPK knockout on exercise-induced gene activation in mouse skeletal muscle. FASEB J. 2005;19(9):1146-8.

68. McAinch AJ, Cameron-Smith D. Adiponectin decreases pyruvate dehydrogenase kinase 4 gene expression in obese- and diabetic-derived myotubes. Diabetes Obes Metab. 2009;11(7):721-8.

69. Puthanveetil P, Wang Y, Wang F, Kim MS, Abrahani A, Rodrigues B. The increase in cardiac pyruvate dehydrogenase kinase-4 after short-term dexamethasone is controlled by an Akt-p38-forkhead box other factor-1 signaling axis. Endocrinology. 2010;151(5):2306-18.

70. Wende AR, Schaeffer PJ, Parker GJ, Zechner C, Han DH, Chen MM, et al. A role for the transcriptional coactivator PGC-1alpha in muscle refueling. J Biol Chem. 2007;282(50):36642-51.

71. Wende AR, Huss JM, Schaeffer PJ, Giguere V, Kelly DP. PGC-1alpha coactivates PDK4 gene expression via the orphan nuclear receptor ERRalpha: a mechanism for transcriptional control of muscle glucose metabolism. Mol Cell Biol. 2005;25(24):10684-94.
72. Terada S, Goto M, Kato M, Kawanaka K, Shimokawa T, Tabata I. Effects of lowintensity prolonged exercise on PGC-1 mRNA expression in rat epitrochlearis muscle. Biochem Biophys Res Commun. 2002;296(2):350-4.

73. Jager S, Handschin C, St-Pierre J, Spiegelman BM. AMP-activated protein kinase (AMPK) action in skeletal muscle via direct phosphorylation of PGC-1alpha. Proc Natl Acad Sci U S A. 2007;104(29):12017-22.

74. Cadoudal T, Leroyer S, Reis AF, Tordjman J, Durant S, Fouque F, et al. Proposed involvement of adipocyte glyceroneogenesis and phosphoenolpyruvate carboxykinase in the metabolic syndrome. Biochimie. 2005;87(1):27-32.

75. Bogacka I, Xie H, Bray GA, Smith SR. The effect of pioglitazone on peroxisome proliferator-activated receptor-gamma target genes related to lipid storage in vivo. Diabetes Care. 2004;27(7):1660-7.

76. Leroyer SN, Tordjman J, Chauvet G, Quette J, Chapron C, Forest C, et al. Rosiglitazone controls fatty acid cycling in human adipose tissue by means of glyceroneogenesis and glycerol phosphorylation. J Biol Chem. 2006;281(19):13141-9.

77. Olswang Y, Cohen H, Papo O, Cassuto H, Croniger CM, Hakimi P, et al. A mutation in the peroxisome proliferator-activated receptor gamma-binding site in the gene for the cytosolic form of phosphoenolpyruvate carboxykinase reduces adipose tissue size and fat content in mice. Proc Natl Acad Sci U S A. 2002;99(2):625-30.

78. Franckhauser S, Munoz S, Pujol A, Casellas A, Riu E, Otaegui P, et al. Increased fatty acid re-esterification by PEPCK overexpression in adipose tissue leads to obesity without insulin resistance. Diabetes. 2002;51(3):624-30.

79. Franckhauser S, Munoz S, Elias I, Ferre T, Bosch F. Adipose overexpression of phosphoenolpyruvate carboxykinase leads to high susceptibility to diet-induced insulin resistance and obesity. Diabetes. 2006;55(2):273-80.

80. Millward CA, Desantis D, Hsieh CW, Heaney JD, Pisano S, Olswang Y, et al. Phosphoenolpyruvate carboxykinase (Pck1) helps regulate the triglyceride/fatty acid cycle and development of insulin resistance in mice. J Lipid Res. 2010;51(6):1452-63.

81. Banzet S, Koulmann N, Simler N, Sanchez H, Chapot R, Serrurier B, et al. Control of gluconeogenic genes during intense/prolonged exercise: hormone-independent effect of muscle-derived IL-6 on hepatic tissue and PEPCK mRNA. J Appl Physiol. 2009;107(6):1830-9.

82. Haase TN, Ringholm S, Leick L, Bienso RS, Kiilerich K, Johansen S, et al. Role of PGC-1alpha in exercise and fasting-induced adaptations in mouse liver. Am J Physiol Regul Integr Comp Physiol. 2011;301(5):R1501-9.

83. Sponarova J, Mustard KJ, Horakova O, Flachs P, Rossmeisl M, Brauner P, et al. Involvement of AMP-activated protein kinase in fat depot-specific metabolic changes during starvation. FEBS Lett. 2005;579(27):6105-10.

84. Reshef L, Hanson RW, Ballard FJ. Glyceride-glycerol synthesis from pyruvate. Adaptive changes in phosphoenolpyruvate carboxykinase and pyruvate carboxylase in adipose tissue and liver. J Biol Chem. 1969;244(8):1994-2001.

85. Beale EG, Tishler EJ. Expression and regulation of cytosolic phosphoenolpyruvate carboxykinase in 3T3-L1 adipocytes. Biochem Biophys Res Commun. 1992;189(2):925-30.

86. Franckhauser S, Antras-Ferry J, Robin P, Robin D, Granner DK, Forest C. Expression of the phosphoenolpyruvate carboxykinase gene in 3T3-F442A adipose cells: opposite effects of dexamethasone and isoprenaline on transcription. Biochem J. 1995;305 (Pt 1):65-71.

87. Leroyer S, Vatier C, Kadiri S, Quette J, Chapron C, Capeau J, et al. Glyceroneogenesis is inhibited through HIV protease inhibitor-induced inflammation in human subcutaneous but not visceral adipose tissue. J Lipid Res. 2011;52(2):207-20.

88. Zhang J, Henagan TM, Gao Z, Ye J. Inhibition of glyceroneogenesis by histone deacetylase 3 contributes to lipodystrophy in mice with adipose tissue inflammation. Endocrinology. 2011;152(5):1829-38.

89. Feingold KR, Moser A, Shigenaga JK, Grunfeld C. Inflammation inhibits the expression of phosphoenolpyruvate carboxykinase in liver and adipose tissue. Innate Immun. 2011.

90. Ballard FJ, Hanson RW. The citrate cleavage pathway and lipogenesis in rat adipose tissue: replenishment of oxaloacetate. J Lipid Res. 1967;8(2):73-9.

91. Kaaman M, Sparks LM, van Harmelen V, Smith SR, Sjolin E, Dahlman I, et al. Strong association between mitochondrial DNA copy number and lipogenesis in human white adipose tissue. Diabetologia. 2007;50(12):2526-33.

92. Kajimoto K, Terada H, Baba Y, Shinohara Y. Essential role of citrate export from mitochondria at early differentiation stage of 3T3-L1 cells for their effective differentiation into fat cells, as revealed by studies using specific inhibitors of mitochondrial di- and tricarboxylate carriers. Mol Genet Metab. 2005;85(1):46-53.

93. Spiegelman BM, Puigserver P, Wu Z. Regulation of adipogenesis and energy balance by PPARgamma and PGC-1. Int J Obes Relat Metab Disord. 2000;24 Suppl 4:S8-10.

94. Rosen ED, Spiegelman BM. Molecular regulation of adipogenesis. Annu Rev Cell Dev Biol. 2000;16:145-71.

95. Wilson-Fritch L, Burkart A, Bell G, Mendelson K, Leszyk J, Nicoloro S, et al. Mitochondrial biogenesis and remodeling during adipogenesis and in response to the insulin sensitizer rosiglitazone. Mol Cell Biol. 2003;23(3):1085-94.

96. Kim BW, Choo HJ, Lee JW, Kim JH, Ko YG. Extracellular ATP is generated by ATP synthase complex in adipocyte lipid rafts. Exp Mol Med. 2004;36(5):476-85.

97. Powelka AM, Seth A, Virbasius JV, Kiskinis E, Nicoloro SM, Guilherme A, et al. Suppression of oxidative metabolism and mitochondrial biogenesis by the transcriptional corepressor RIP140 in mouse adipocytes. J Clin Invest. 2006;116(1):125-36.

98. Ducluzeau PH, Priou M, Weitheimer M, Flamment M, Duluc L, Iacobazi F, et al. Dynamic regulation of mitochondrial network and oxidative functions during 3T3-L1 fat cell differentiation. J Physiol Biochem. 2011;67(3):285-96.

99. Carriere A, Fernandez Y, Rigoulet M, Penicaud L, Casteilla L. Inhibition of preadipocyte proliferation by mitochondrial reactive oxygen species. FEBS Lett. 2003;550(1-3):163-7.

100. Koh EH, Park JY, Park HS, Jeon MJ, Ryu JW, Kim M, et al. Essential role of mitochondrial function in adiponectin synthesis in adipocytes. Diabetes. 2007;56(12):2973-81.

101. Sutherland LN, Capozzi LC, Turchinsky NJ, Bell RC, Wright DC. Time course of high-fat diet-induced reductions in adipose tissue mitochondrial proteins: potential mechanisms and the relationship to glucose intolerance. Am J Physiol Endocrinol Metab. 2008;295(5):E1076-83.

102. Choo HJ, Kim JH, Kwon OB, Lee CS, Mun JY, Han SS, et al. Mitochondria are impaired in the adipocytes of type 2 diabetic mice. Diabetologia. 2006;49(4):784-91.

103. Rong JX, Qiu Y, Hansen MK, Zhu L, Zhang V, Xie M, et al. Adipose mitochondrial biogenesis is suppressed in db/db and high-fat diet-fed mice and improved by rosiglitazone. Diabetes. 2007;56(7):1751-60.

104. Bogacka I, Xie H, Bray GA, Smith SR. Pioglitazone induces mitochondrial biogenesis in human subcutaneous adipose tissue in vivo. Diabetes. 2005;54(5):1392-9.

105. Dahlman I, Forsgren M, Sjogren A, Nordstrom EA, Kaaman M, Naslund E, et al. Downregulation of electron transport chain genes in visceral adipose tissue in type 2 diabetes independent of obesity and possibly involving tumor necrosis factor-alpha. Diabetes. 2006;55(6):1792-9.

106. Ost A, Svensson K, Ruishalme I, Brannmark C, Franck N, Krook H, et al. Attenuated mTOR signaling and enhanced autophagy in adipocytes from obese patients with type 2 diabetes. Mol Med. 2010;16(7-8):235-46.

107. Chattopadhyay M, Guhathakurta I, Behera P, Ranjan KR, Khanna M, Mukhopadhyay S, et al. Mitochondrial bioenergetics is not impaired in nonobese subjects with type 2 diabetes mellitus. Metabolism. 2011;60(12):1702-10.

108. Mustelin L, Pietilainen KH, Rissanen A, Sovijarvi AR, Piirila P, Naukkarinen J, et al. Acquired obesity and poor physical fitness impair expression of genes of mitochondrial oxidative phosphorylation in monozygotic twins discordant for obesity. Am J Physiol Endocrinol Metab. 2008;295(1):E148-54.

109. Liu C, Lin JD. PGC-1 coactivators in the control of energy metabolism. Acta Biochim Biophys Sin (Shanghai). 2011;43(4):248-57.

110. Valerio A, Cardile A, Cozzi V, Bracale R, Tedesco L, Pisconti A, et al. TNFalpha downregulates eNOS expression and mitochondrial biogenesis in fat and muscle of obese rodents. J Clin Invest. 2006;116(10):2791-8.

111. Kakuma T, Wang ZW, Pan W, Unger RH, Zhou YT. Role of leptin in peroxisome proliferator-activated receptor gamma coactivator-1 expression. Endocrinology. 2000;141(12):4576-82.

112. Hammarstedt A, Jansson PA, Wesslau C, Yang X, Smith U. Reduced expression of PGC-1 and insulin-signaling molecules in adipose tissue is associated with insulin resistance. Biochem Biophys Res Commun. 2003;301(2):578-82.

113. Semple RK, Crowley VC, Sewter CP, Laudes M, Christodoulides C, Considine RV, et al. Expression of the thermogenic nuclear hormone receptor coactivator PGC-1alpha is reduced in the adipose tissue of morbidly obese subjects. Int J Obes Relat Metab Disord. 2004;28(1):176-9.

114. Laye MJ, Rector RS, Warner SO, Naples SP, Perretta AL, Uptergrove GM, et al. Changes in visceral adipose tissue mitochondrial content with type 2 diabetes and daily voluntary wheel running in OLETF rats. J Physiol. 2009;587(Pt 14):3729-39.

115. Pietilainen KH, Naukkarinen J, Rissanen A, Saharinen J, Ellonen P, Keranen H, et al. Global transcript profiles of fat in monozygotic twins discordant for BMI: pathways behind acquired obesity. PLoS Med. 2008;5(3):e51.

116. Katic M, Kennedy AR, Leykin I, Norris A, McGettrick A, Gesta S, et al. Mitochondrial gene expression and increased oxidative metabolism: role in increased lifespan of fat-specific insulin receptor knock-out mice. Aging Cell. 2007;6(6):827-39.

117. Roden M, Price TB, Perseghin G, Petersen KF, Rothman DL, Cline GW, et al. Mechanism of free fatty acid-induced insulin resistance in humans. J Clin Invest. 1996;97(12):2859-65.

118. Shi X, Burkart A, Nicoloro SM, Czech MP, Straubhaar J, Corvera S. Paradoxical effect of mitochondrial respiratory chain impairment on insulin signaling and glucose transport in adipose cells. J Biol Chem. 2008;283(45):30658-67.

119. Taniguchi CM, Emanuelli B, Kahn CR. Critical nodes in signalling pathways: insights into insulin action. Nat Rev Mol Cell Biol. 2006;7(2):85-96.

120. Avruch J. Insulin signal transduction through protein kinase cascades. Mol Cell Biochem. 1998;182(1-2):31-48.

121. Fritsche L, Weigert C, Haring HU, Lehmann R. How insulin receptor substrate proteins regulate the metabolic capacity of the liver--implications for health and disease. Curr Med Chem. 2008;15(13):1316-29.

122. Boura-Halfon S, Zick Y. Phosphorylation of IRS proteins, insulin action, and insulin resistance. Am J Physiol Endocrinol Metab. 2009;296(4):E581-91.

123. Kohn AD, Summers SA, Birnbaum MJ, Roth RA. Expression of a constitutively active Akt Ser/Thr kinase in 3T3-L1 adipocytes stimulates glucose uptake and glucose transporter 4 translocation. J Biol Chem. 1996;271(49):31372-8.

124. Kitamura T, Kitamura Y, Kuroda S, Hino Y, Ando M, Kotani K, et al. Insulininduced phosphorylation and activation of cyclic nucleotide phosphodiesterase 3B by the serine-threonine kinase Akt. Mol Cell Biol. 1999;19(9):6286-96.

125. Berggreen C, Gormand A, Omar B, Degerman E, Goransson O. Protein kinase B activity is required for the effects of insulin on lipid metabolism in adipocytes. Am J Physiol Endocrinol Metab. 2009;296(4):E635-46.

126. Choi SM, Tucker DF, Gross DN, Easton RM, DiPilato LM, Dean AS, et al. Insulin regulates adipocyte lipolysis via an Akt-independent signaling pathway. Mol Cell Biol. 2010;30(21):5009-20.

127. Gehart H, Kumpf S, Ittner A, Ricci R. MAPK signalling in cellular metabolism: stress or wellness? EMBO Rep. 2010;11(11):834-40.

128. Gual P, Le Marchand-Brustel Y, Tanti JF. Positive and negative regulation of insulin signaling through IRS-1 phosphorylation. Biochimie. 2005;87(1):99-109.

129. Carlson CJ, Koterski S, Sciotti RJ, Poccard GB, Rondinone CM. Enhanced basal activation of mitogen-activated protein kinases in adipocytes from type 2 diabetes: potential role of p38 in the downregulation of GLUT4 expression. Diabetes. 2003;52(3):634-41.

130. Bluher M, Bashan N, Shai I, Harman-Boehm I, Tarnovscki T, Avinaoch E, et al. Activated Ask1-MKK4-p38MAPK/JNK stress signaling pathway in human omental fat tissue may link macrophage infiltration to whole-body Insulin sensitivity. J Clin Endocrinol Metab. 2009;94(7):2507-15.

131. Fujishiro M, Gotoh Y, Katagiri H, Sakoda H, Ogihara T, Anai M, et al. Three mitogen-activated protein kinases inhibit insulin signaling by different mechanisms in 3T3-L1 adipocytes. Mol Endocrinol. 2003;17(3):487-97.

132. Sale EM, Atkinson PG, Sale GJ. Requirement of MAP kinase for differentiation of fibroblasts to adipocytes, for insulin activation of p90 S6 kinase and for insulin or serum stimulation of DNA synthesis. EMBO J. 1995;14(4):674-84.

133. Bost F, Aouadi M, Caron L, Even P, Belmonte N, Prot M, et al. The extracellular signal-regulated kinase isoform ERK1 is specifically required for in vitro and in vivo adipogenesis. Diabetes. 2005;54(2):402-11.

134. Rodriguez A, Duran A, Selloum M, Champy MF, Diez-Guerra FJ, Flores JM, et al. Mature-onset obesity and insulin resistance in mice deficient in the signaling adapter p62. Cell Metab. 2006;3(3):211-22.

135. Hirosumi J, Tuncman G, Chang L, Gorgun CZ, Uysal KT, Maeda K, et al. A central role for JNK in obesity and insulin resistance. Nature. 2002;420(6913):333-6.

136. Sabio G, Das M, Mora A, Zhang Z, Jun JY, Ko HJ, et al. A stress signaling pathway in adipose tissue regulates hepatic insulin resistance. Science. 2008;322(5907):1539-43.

137. Gao Z, Zhang X, Zuberi A, Hwang D, Quon MJ, Lefevre M, et al. Inhibition of insulin sensitivity by free fatty acids requires activation of multiple serine kinases in 3T3-L1 adipocytes. Mol Endocrinol. 2004;18(8):2024-34.

138. Sweeney G, Somwar R, Ramlal T, Volchuk A, Ueyama A, Klip A. An inhibitor of p38 mitogen-activated protein kinase prevents insulin-stimulated glucose transport but

not glucose transporter translocation in 3T3-L1 adipocytes and L6 myotubes. J Biol Chem. 1999;274(15):10071-8.

139. Somwar R, Koterski S, Sweeney G, Sciotti R, Djuric S, Berg C, et al. A dominant-negative p38 MAPK mutant and novel selective inhibitors of p38 MAPK reduce insulin-stimulated glucose uptake in 3T3-L1 adipocytes without affecting GLUT4 translocation. J Biol Chem. 2002;277(52):50386-95.

140. Kayali AG, Austin DA, Webster NJ. Stimulation of MAPK cascades by insulin and osmotic shock: lack of an involvement of p38 mitogen-activated protein kinase in glucose transport in 3T3-L1 adipocytes. Diabetes. 2000;49(11):1783-93.

141. Turban S, Beardmore VA, Carr JM, Sakamoto K, Hajduch E, Arthur JS, et al. Insulin-stimulated glucose uptake does not require p38 mitogen-activated protein kinase in adipose tissue or skeletal muscle. Diabetes. 2005;54(11):3161-8.

142. Wu X, Motoshima H, Mahadev K, Stalker TJ, Scalia R, Goldstein BJ. Involvement of AMP-activated protein kinase in glucose uptake stimulated by the globular domain of adiponectin in primary rat adipocytes. Diabetes. 2003;52(6):1355-63.

143. Fu Y, Luo N, Klein RL, Garvey WT. Adiponectin promotes adipocyte differentiation, insulin sensitivity, and lipid accumulation. J Lipid Res. 2005;46(7):1369-79.

144. Bauche IB, El Mkadem SA, Pottier AM, Senou M, Many MC, Rezsohazy R, et al. Overexpression of adiponectin targeted to adipose tissue in transgenic mice: impaired adipocyte differentiation. Endocrinology. 2007;148(4):1539-49.

145. Wedellova Z, Dietrich J, Siklova-Vitkova M, Kolostova K, Kovacikova M, Duskova M, et al. Adiponectin inhibits spontaneous and catecholamine-induced lipolysis

in human adipocytes of non-obese subjects through AMPK-dependent mechanisms. Physiol Res. 2011;60(1):139-48.

146. Qiao L, Kinney B, Schaack J, Shao J. Adiponectin inhibits lipolysis in mouse adipocytes. Diabetes. 2011;60(5):1519-27.

147. Asterholm IW, Scherer PE. Enhanced metabolic flexibility associated with elevated adiponectin levels. Am J Pathol. 2010;176(3):1364-76.

148. Shetty S, Ramos-Roman MA, Cho YR, Brown J, Plutzky J, Muise ES, et al. Enhanced fatty acid flux triggered by adiponectin overexpression. Endocrinology. 2012;153(1):113-22.

149. Fasshauer M, Klein J, Kralisch S, Klier M, Lossner U, Bluher M, et al. Growth hormone is a positive regulator of adiponectin receptor 2 in 3T3-L1 adipocytes. FEBS Lett. 2004;558(1-3):27-32.

150. Tan BK, Chen J, Digby JE, Keay SD, Kennedy CR, Randeva HS. Upregulation of adiponectin receptor 1 and 2 mRNA and protein in adipose tissue and adipocytes in insulin-resistant women with polycystic ovary syndrome. Diabetologia. 2006;49(11):2723-8.

151. Qiang L, Wang H, Farmer SR. Adiponectin secretion is regulated by SIRT1 and the endoplasmic reticulum oxidoreductase Ero1-L alpha. Mol Cell Biol. 2007;27(13):4698-707.

152. Tsao TS, Murrey HE, Hug C, Lee DH, Lodish HF. Oligomerization statedependent activation of NF-kappa B signaling pathway by adipocyte complement-related protein of 30 kDa (Acrp30). J Biol Chem. 2002;277(33):29359-62.

153. Pajvani UB, Du X, Combs TP, Berg AH, Rajala MW, Schulthess T, et al. Structure-function studies of the adipocyte-secreted hormone Acrp30/adiponectin. Implications fpr metabolic regulation and bioactivity. J Biol Chem. 2003;278(11):9073-85.

154. Waki H, Yamauchi T, Kamon J, Ito Y, Uchida S, Kita S, et al. Impaired multimerization of human adiponectin mutants associated with diabetes. Molecular structure and multimer formation of adiponectin. J Biol Chem. 2003;278(41):40352-63.

155. Liu M, Liu F. Transcriptional and post-translational regulation of adiponectin. Biochem J. 2010;425(1):41-52.

156. Iwaki M, Matsuda M, Maeda N, Funahashi T, Matsuzawa Y, Makishima M, et al. Induction of adiponectin, a fat-derived antidiabetic and antiatherogenic factor, by nuclear receptors. Diabetes. 2003;52(7):1655-63.

157. Seo JB, Moon HM, Noh MJ, Lee YS, Jeong HW, Yoo EJ, et al. Adipocyte determination- and differentiation-dependent factor 1/sterol regulatory element-binding protein 1c regulates mouse adiponectin expression. J Biol Chem. 2004;279(21):22108-17.
158. Liu M, Zhou L, Xu A, Lam KS, Wetzel MD, Xiang R, et al. A disulfide-bond A oxidoreductase-like protein (DsbA-L) regulates adiponectin multimerization. Proc Natl Acad Sci U S A. 2008;105(47):18302-7.

159. Wang ZV, Schraw TD, Kim JY, Khan T, Rajala MW, Follenzi A, et al. Secretion of the adipocyte-specific secretory protein adiponectin critically depends on thiolmediated protein retention. Mol Cell Biol. 2007;27(10):3716-31.

160. Stallknecht B, Vinten J, Ploug T, Galbo H. Increased activities of mitochondrial enzymes in white adipose tissue in trained rats. Am J Physiol. 1991;261(3 Pt 1):E410-4.

161. Sutherland LN, Bomhof MR, Capozzi LC, Basaraba SA, Wright DC. Exercise and adrenaline increase PGC-1{alpha} mRNA expression in rat adipose tissue. J Physiol. 2009;587(Pt 7):1607-17.

162. Xu X, Ying Z, Cai M, Xu Z, Li Y, Jiang SY, et al. Exercise ameliorates high-fat diet-induced metabolic and vascular dysfunction, and increases adipocyte progenitor cell population in brown adipose tissue. Am J Physiol Regul Integr Comp Physiol. 2011;300(5):R1115-25.

163. Ruschke K, Fishbein L, Dietrich A, Kloting N, Tonjes A, Oberbach A, et al. Gene expression of PPARgamma and PGC-1alpha in human omental and subcutaneous adipose tissues is related to insulin resistance markers and mediates beneficial effects of physical training. Eur J Endocrinol. 2010;162(3):515-23.

164. Camera DM, Anderson MJ, Hawley JA, Carey AL. Short-term endurance training does not alter the oxidative capacity of human subcutaneous adipose tissue. Eur J Appl Physiol. 2010;109(2):307-16.

165. Koh HJ, Hirshman MF, He H, Li Y, Manabe Y, Balschi JA, et al. Adrenaline is a critical mediator of acute exercise-induced AMP-activated protein kinase activation in adipocytes. Biochem J. 2007;403(3):473-81.

166. Tansey JT, Huml AM, Vogt R, Davis KE, Jones JM, Fraser KA, et al. Functional studies on native and mutated forms of perilipins. A role in protein kinase A-mediated lipolysis of triacylglycerols. J Biol Chem. 2003;278(10):8401-6.

167. Londos C, Brasaemle DL, Gruia-Gray J, Servetnick DA, Schultz CJ, Levin DM, et al. Perilipin: unique proteins associated with intracellular neutral lipid droplets in adipocytes and steroidogenic cells. Biochem Soc Trans. 1995;23(3):611-5.

168. Horowitz JF. Fatty acid mobilization from adipose tissue during exercise. Trends Endocrinol Metab. 2003;14(8):386-92.

169. Egan JJ, Greenberg AS, Chang MK, Wek SA, Moos MC, Jr., Londos C. Mechanism of hormone-stimulated lipolysis in adipocytes: translocation of hormone-sensitive lipase to the lipid storage droplet. Proc Natl Acad Sci U S A. 1992;89(18):8537-41.

170. Ogasawara J, Nomura S, Rahman N, Sakurai T, Kizaki T, Izawa T, et al. Hormone-sensitive lipase is critical mediators of acute exercise-induced regulation of lipolysis in rat adipocytes. Biochem Biophys Res Commun. 2010;400(1):134-9.

171. Anthonsen MW, Ronnstrand L, Wernstedt C, Degerman E, Holm C. Identification of novel phosphorylation sites in hormone-sensitive lipase that are phosphorylated in response to isoproterenol and govern activation properties in vitro. J Biol Chem. 1998;273(1):215-21.

172. Shen WJ, Patel S, Natu V, Kraemer FB. Mutational analysis of structural features of rat hormone-sensitive lipase. Biochemistry. 1998;37(25):8973-9.

173. Zimmermann R, Strauss JG, Haemmerle G, Schoiswohl G, Birner-Gruenberger R, Riederer M, et al. Fat mobilization in adipose tissue is promoted by adipose triglyceride lipase. Science. 2004;306(5700):1383-6.

174. Villena JA, Roy S, Sarkadi-Nagy E, Kim KH, Sul HS. Desnutrin, an adipocyte gene encoding a novel patatin domain-containing protein, is induced by fasting and glucocorticoids: ectopic expression of desnutrin increases triglyceride hydrolysis. J Biol Chem. 2004;279(45):47066-75.

175. Haemmerle G, Lass A, Zimmermann R, Gorkiewicz G, Meyer C, Rozman J, et al. Defective lipolysis and altered energy metabolism in mice lacking adipose triglyceride lipase. Science. 2006;312(5774):734-7.

176. Huijsman E, van de Par C, Economou C, van der Poel C, Lynch GS, Schoiswohl G, et al. Adipose triacylglycerol lipase deletion alters whole body energy metabolism and impairs exercise performance in mice. Am J Physiol Endocrinol Metab. 2009;297(2):E505-13.

177. Wohlers LM, Jackson KC, Spangenburg EE. Lipolytic signaling in response to acute exercise is altered in female mice following ovariectomy. J Cell Biochem. 2011;112(12):3675-84.

178. Ferguson MA, White LJ, McCoy S, Kim HW, Petty T, Wilsey J. Plasma adiponectin response to acute exercise in healthy subjects. Eur J Appl Physiol. 2004;91(2-3):324-9.

179. Kobayashi J, Murase Y, Asano A, Nohara A, Kawashiri MA, Inazu A, et al. Effect of walking with a pedometer on serum lipid and adiponectin levels in Japanese middle-aged men. J Atheroscler Thromb. 2006;13(4):197-201.

180. Yatagai T, Nagasaka S, Taniguchi A, Fukushima M, Nakamura T, Kuroe A, et al. Hypoadiponectinemia is associated with visceral fat accumulation and insulin resistance in Japanese men with type 2 diabetes mellitus. Metabolism. 2003;52(10):1274-8.

181. Bluher M, Bullen JW, Jr., Lee JH, Kralisch S, Fasshauer M, Kloting N, et al. Circulating adiponectin and expression of adiponectin receptors in human skeletal muscle: associations with metabolic parameters and insulin resistance and regulation by physical training. J Clin Endocrinol Metab. 2006;91(6):2310-6. 182. Oberbach A, Tonjes A, Kloting N, Fasshauer M, Kratzsch J, Busse MW, et al. Effect of a 4 week physical training program on plasma concentrations of inflammatory markers in patients with abnormal glucose tolerance. Eur J Endocrinol. 2006;154(4):577-85.

183. Lee JA, Kim JW, Kim DY. Effects of yoga exercise on serum adiponectin and metabolic syndrome factors in obese postmenopausal women. Menopause. 2011.

184. Esposito K, Pontillo A, Di Palo C, Giugliano G, Masella M, Marfella R, et al. Effect of weight loss and lifestyle changes on vascular inflammatory markers in obese women: a randomized trial. JAMA. 2003;289(14):1799-804.

185. Giannessi D, Maltinti M, Del Ry S. Adiponectin circulating levels: a new emerging biomarker of cardiovascular risk. Pharmacol Res. 2007;56(6):459-67.

186. Garekani ET, Mohebbi H, Kraemer RR, Fathi R. Exercise training intensity/volume affects plasma and tissue adiponectin concentrations in the male rat. Peptides. 2011;32(5):1008-12.

187. Eriksson M, Johnson O, Boman K, Hallmans G, Hellsten G, Nilsson TK, et al. Improved fibrinolytic activity during exercise may be an effect of the adipocyte-derived hormones leptin and adiponectin. Thromb Res. 2008;122(5):701-8.

188. Jurimae J, Maestu J, Jurimae T. Leptin as a marker of training stress in highly trained male rowers? Eur J Appl Physiol. 2003;90(5-6):533-8.

189. Ishigaki T, Koyama K, Tsujita J, Tanaka N, Hori S, Oku Y. Plasma leptin levels of elite endurance runners after heavy endurance training. J Physiol Anthropol Appl Human Sci. 2005;24(6):573-8.

190. Bouassida A, Chamari K, Zaouali M, Feki Y, Zbidi A, Tabka Z. Review on leptin and adiponectin responses and adaptations to acute and chronic exercise. Br J Sports Med. 2010;44(9):620-30.

191. Goodyear LJ, Hirshman MF, Horton ED, Knutson SM, Wardzala LJ, Horton ES. Exercise training normalizes glucose metabolism in a rat model of impaired glucose tolerance. Metabolism. 1991;40(5):455-64.

192. Hirshman MF, Goodyear LJ, Horton ED, Wardzala LJ, Horton ES. Exercise training increases GLUT-4 protein in rat adipose cells. Am J Physiol. 1993;264(6 Pt 1):E882-9.

193. Hussey SE, McGee SL, Garnham A, Wentworth JM, Jeukendrup AE, Hargreaves M. Exercise training increases adipose tissue GLUT4 expression in patients with type 2 diabetes. Diabetes Obes Metab. 2011;13(10):959-62.

194. Peres SB, de Moraes SM, Costa CE, Brito LC, Takada J, Andreotti S, et al. Endurance exercise training increases insulin responsiveness in isolated adipocytes through IRS/PI3-kinase/Akt pathway. J Appl Physiol. 2005;98(3):1037-43.

195. Gauthier MS, Miyoshi H, Souza SC, Cacicedo JM, Saha AK, Greenberg AS, et al. AMP-activated protein kinase is activated as a consequence of lipolysis in the adipocyte: potential mechanism and physiological relevance. J Biol Chem. 2008;283(24):16514-24.

196. Park H, Kaushik VK, Constant S, Prentki M, Przybytkowski E, Ruderman NB, et al. Coordinate regulation of malonyl-CoA decarboxylase, sn-glycerol-3-phosphate acyltransferase, and acetyl-CoA carboxylase by AMP-activated protein kinase in rat tissues in response to exercise. J Biol Chem. 2002;277(36):32571-7.

197. Moule SK, Denton RM. The activation of p38 MAPK by the beta-adrenergic agonist isoproterenol in rat epididymal fat cells. FEBS Lett. 1998;439(3):287-90.

198. Cao W, Medvedev AV, Daniel KW, Collins S. beta-Adrenergic activation of p38 MAP kinase in adipocytes: cAMP induction of the uncoupling protein 1 (UCP1) gene requires p38 MAP kinase. J Biol Chem. 2001;276(29):27077-82.

199. Mottillo EP, Shen XJ, Granneman JG. beta3-adrenergic receptor induction of adipocyte inflammation requires lipolytic activation of stress kinases p38 and JNK. Biochim Biophys Acta. 2010;1801(9):1048-55.

200. Watt MJ, Holmes AG, Pinnamaneni SK, Garnham AP, Steinberg GR, Kemp BE, et al. Regulation of HSL serine phosphorylation in skeletal muscle and adipose tissue. Am J Physiol Endocrinol Metab. 2006;290(3):E500-8.

201. Brandt C, Jakobsen AH, Adser H, Olesen J, Iversen N, Kristensen JM, et al. IL-6 regulates exercise and training-induced adaptations in subcutaneous adipose tissue in mice. Acta Physiol (Oxf). 2011.

202. Kristensen JM, Johnsen AB, Birk JB, Nielsen JN, Jensen BR, Hellsten Y, et al. Absence of humoral mediated 5'AMP-activated protein kinase activation in human skeletal muscle and adipose tissue during exercise. J Physiol. 2007;585(Pt 3):897-909.

203. Bostrom P, Wu J, Jedrychowski MP, Korde A, Ye L, Lo JC, et al. A PGC1-alphadependent myokine that drives brown-fat-like development of white fat and thermogenesis. Nature. 2012;481(7382):463-8.

204. Pedersen BK, Akerstrom TC, Nielsen AR, Fischer CP. Role of myokines in exercise and metabolism. J Appl Physiol. 2007;103(3):1093-8.

205. Steensberg A, van Hall G, Osada T, Sacchetti M, Saltin B, Klarlund Pedersen B. Production of interleukin-6 in contracting human skeletal muscles can account for the exercise-induced increase in plasma interleukin-6. J Physiol. 2000;529 Pt 1:237-42.

206. Pedersen BK. Muscular IL-6 and Its Role as an Energy Sensor. Med Sci Sports Exerc. 2011.

207. Petersen EW, Carey AL, Sacchetti M, Steinberg GR, Macaulay SL, Febbraio MA, et al. Acute IL-6 treatment increases fatty acid turnover in elderly humans in vivo and in tissue culture in vitro. Am J Physiol Endocrinol Metab. 2005;288(1):E155-62.

208. Yang Y, Ju D, Zhang M, Yang G. Interleukin-6 stimulates lipolysis in porcine adipocytes. Endocrine. 2008;33(3):261-9.

209. Kelly M, Keller C, Avilucea PR, Keller P, Luo Z, Xiang X, et al. AMPK activity is diminished in tissues of IL-6 knockout mice: the effect of exercise. Biochem Biophys Res Commun. 2004;320(2):449-54.

210. Ruderman NB, Keller C, Richard AM, Saha AK, Luo Z, Xiang X, et al. Interleukin-6 regulation of AMP-activated protein kinase. Potential role in the systemic response to exercise and prevention of the metabolic syndrome. Diabetes. 2006;55 Suppl 2:S48-54.

211. Wolsk E, Mygind H, Grondahl TS, Pedersen BK, van Hall G. IL-6 selectively stimulates fat metabolism in human skeletal muscle. Am J Physiol Endocrinol Metab. 2010;299(5):E832-40.

212. Zhang J, Henagan TM, Gao Z, Ye J. Inhibition of glyceroneogenesis by histone deacetylase 3 contributes to lipodystrophy in mice with adipose tissue inflammation. Endocrinology.152(5):1829-38.

213. Leroyer S, Vatier C, Kadiri S, Quette J, Chapron C, Capeau J, et al. Glyceroneogenesis is inhibited through HIV protease inhibitor-induced inflammation in human subcutaneous but not visceral adipose tissue. J Lipid Res.52(2):207-20.

214. Nielsen NB, Hojbjerre L, Sonne MP, Alibegovic AC, Vaag A, Dela F, et al. Interstitial concentrations of adipokines in subcutaneous abdominal and femoral adipose tissue. Regul Pept. 2009;155(1-3):39-45.

215. Ostrowski K, Hermann C, Bangash A, Schjerling P, Nielsen JN, Pedersen BK. A trauma-like elevation of plasma cytokines in humans in response to treadmill running. J Physiol. 1998;513 (Pt 3):889-94.

216. DeRijk RH, Boelen A, Tilders FJ, Berkenbosch F. Induction of plasma interleukin-6 by circulating adrenaline in the rat. Psychoneuroendocrinology. 1994;19(2):155-63.

217. Frost RA, Nystrom GJ, Lang CH. Epinephrine stimulates IL-6 expression in skeletal muscle and C2C12 myoblasts: role of c-Jun NH2-terminal kinase and histone deacetylase activity. Am J Physiol Endocrinol Metab. 2004;286(5):E809-17.

218. Vicennati V, Vottero A, Friedman C, Papanicolaou DA. Hormonal regulation of interleukin-6 production in human adipocytes. Int J Obes Relat Metab Disord. 2002;26(7):905-11.

219. Helge JW, Stallknecht B, Pedersen BK, Galbo H, Kiens B, Richter EA. The effect of graded exercise on IL-6 release and glucose uptake in human skeletal muscle. J Physiol. 2003;546(Pt 1):299-305.

220. Path G, Bornstein SR, Gurniak M, Chrousos GP, Scherbaum WA, Hauner H. Human breast adipocytes express interleukin-6 (IL-6) and its receptor system: increased

IL-6 production by beta-adrenergic activation and effects of IL-6 on adipocyte function. J Clin Endocrinol Metab. 2001;86(5):2281-8.

221. Steensberg A, Toft AD, Schjerling P, Halkjaer-Kristensen J, Pedersen BK. Plasma interleukin-6 during strenuous exercise: role of epinephrine. Am J Physiol Cell Physiol. 2001;281(3):C1001-4.

222. Pedersen BK, Febbraio MA. Muscles, exercise and obesity: skeletal muscle as a secretory organ. Nat Rev Endocrinol. 2012.

223. Pedersen L, Pilegaard H, Hansen J, Brandt C, Adser H, Hidalgo J, et al. Exerciseinduced liver chemokine CXCL-1 expression is linked to muscle-derived interleukin-6 expression. J Physiol. 2011;589(Pt 6):1409-20.

224. Pickup JC. Inflammation and activated innate immunity in the pathogenesis of type 2 diabetes. Diabetes Care. 2004;27(3):813-23.

225. Bastard JP, Maachi M, Van Nhieu JT, Jardel C, Bruckert E, Grimaldi A, et al. Adipose tissue IL-6 content correlates with resistance to insulin activation of glucose uptake both in vivo and in vitro. J Clin Endocrinol Metab. 2002;87(5):2084-9.

226. Kern PA, Ranganathan S, Li C, Wood L, Ranganathan G. Adipose tissue tumor necrosis factor and interleukin-6 expression in human obesity and insulin resistance. Am J Physiol Endocrinol Metab. 2001;280(5):E745-51.

227. Carey AL, Bruce CR, Sacchetti M, Anderson MJ, Olsen DB, Saltin B, et al. Interleukin-6 and tumor necrosis factor-alpha are not increased in patients with Type 2 diabetes: evidence that plasma interleukin-6 is related to fat mass and not insulin responsiveness. Diabetologia. 2004;47(6):1029-37.

228. Lagathu C, Bastard JP, Auclair M, Maachi M, Capeau J, Caron M. Chronic interleukin-6 (IL-6) treatment increased IL-6 secretion and induced insulin resistance in adipocyte: prevention by rosiglitazone. Biochem Biophys Res Commun. 2003;311(2):372-9.

229. Fried SK, Bunkin DA, Greenberg AS. Omental and subcutaneous adipose tissues of obese subjects release interleukin-6: depot difference and regulation by glucocorticoid.J Clin Endocrinol Metab. 1998;83(3):847-50.

230. Cancello R, Henegar C, Viguerie N, Taleb S, Poitou C, Rouault C, et al. Reduction of macrophage infiltration and chemoattractant gene expression changes in white adipose tissue of morbidly obese subjects after surgery-induced weight loss. Diabetes. 2005;54(8):2277-86.

231. Heinrich PC, Behrmann I, Haan S, Hermanns HM, Muller-Newen G, Schaper F. Principles of interleukin (IL)-6-type cytokine signalling and its regulation. Biochem J. 2003;374(Pt 1):1-20.

232. Kamimura D, Ishihara K, Hirano T. IL-6 signal transduction and its physiological roles: the signal orchestration model. Rev Physiol Biochem Pharmacol. 2003;149:1-38.

233. Fujimoto M, Naka T. Regulation of cytokine signaling by SOCS family molecules.Trends Immunol. 2003;24(12):659-66.

234. Sommer U, Schmid C, Sobota RM, Lehmann U, Stevenson NJ, Johnston JA, et al. Mechanisms of SOCS3 phosphorylation upon interleukin-6 stimulation. Contributions of Src- and receptor-tyrosine kinases. J Biol Chem. 2005;280(36):31478-88. 235. Rui L, Yuan M, Frantz D, Shoelson S, White MF. SOCS-1 and SOCS-3 block insulin signaling by ubiquitin-mediated degradation of IRS1 and IRS2. J Biol Chem. 2002;277(44):42394-8.

236. Kristiansen OP, Mandrup-Poulsen T. Interleukin-6 and diabetes: the good, the bad, or the indifferent? Diabetes. 2005;54 Suppl 2:S114-24.

237. Carey AL, Febbraio MA. Interleukin-6 and insulin sensitivity: friend or foe? Diabetologia. 2004;47(7):1135-42.

238. Fasshauer M, Kralisch S, Klier M, Lossner U, Bluher M, Klein J, et al. Adiponectin gene expression and secretion is inhibited by interleukin-6 in 3T3-L1 adipocytes. Biochem Biophys Res Commun. 2003;301(4):1045-50.

239. Rotter V, Nagaev I, Smith U. Interleukin-6 (IL-6) induces insulin resistance in 3T3-L1 adipocytes and is, like IL-8 and tumor necrosis factor-alpha, overexpressed in human fat cells from insulin-resistant subjects. J Biol Chem. 2003;278(46):45777-84.

240. Shi H, Tzameli I, Bjorbaek C, Flier JS. Suppressor of cytokine signaling 3 is a physiological regulator of adipocyte insulin signaling. J Biol Chem. 2004;279(33):34733-40.

241. Stouthard JM, Oude Elferink RP, Sauerwein HP. Interleukin-6 enhances glucose transport in 3T3-L1 adipocytes. Biochem Biophys Res Commun. 1996;220(2):241-5.

242. Huppertz C, Schwartz C, Becker W, Horn F, Heinrich PC, Joost HG. Comparison of the effects of insulin, PDGF, interleukin-6, and interferon-gamma on glucose transport in 3T3-L1 cells: lack of cross-talk between tyrosine kinase receptors and JAK/STAT pathways. Diabetologia. 1996;39(12):1432-9.

243. Stouthard JM, Romijn JA, Van der Poll T, Endert E, Klein S, Bakker PJ, et al. Endocrinologic and metabolic effects of interleukin-6 in humans. Am J Physiol. 1995;268(5 Pt 1):E813-9.

244. Tsigos C, Papanicolaou DA, Kyrou I, Defensor R, Mitsiadis CS, Chrousos GP. Dose-dependent effects of recombinant human interleukin-6 on glucose regulation. J Clin Endocrinol Metab. 1997;82(12):4167-70.

245. Febbraio MA, Hiscock N, Sacchetti M, Fischer CP, Pedersen BK. Interleukin-6 is a novel factor mediating glucose homeostasis during skeletal muscle contraction. Diabetes. 2004;53(7):1643-8.

246. Lyngso D, Simonsen L, Bulow J. Metabolic effects of interleukin-6 in human splanchnic and adipose tissue. J Physiol. 2002;543(Pt 1):379-86.

247. Watt MJ, Carey AL, Wolsk-Petersen E, Kraemer FB, Pedersen BK, Febbraio MA. Hormone-sensitive lipase is reduced in the adipose tissue of patients with type 2 diabetes mellitus: influence of IL-6 infusion. Diabetologia. 2005;48(1):105-12.

248. Carey AL, Steinberg GR, Macaulay SL, Thomas WG, Holmes AG, Ramm G, et al. Interleukin-6 increases insulin-stimulated glucose disposal in humans and glucose uptake and fatty acid oxidation in vitro via AMP-activated protein kinase. Diabetes. 2006;55(10):2688-97.

249. Starkie R, Ostrowski SR, Jauffred S, Febbraio M, Pedersen BK. Exercise and IL-6 infusion inhibit endotoxin-induced TNF-alpha production in humans. FASEB J. 2003;17(8):884-6.

250. Choy EH, Isenberg DA, Garrood T, Farrow S, Ioannou Y, Bird H, et al. Therapeutic benefit of blocking interleukin-6 activity with an anti-interleukin-6 receptor monoclonal antibody in rheumatoid arthritis: a randomized, double-blind, placebocontrolled, dose-escalation trial. Arthritis Rheum. 2002;46(12):3143-50.

251. Febbraio MA, Rose-John S, Pedersen BK. Is interleukin-6 receptor blockade the Holy Grail for inflammatory diseases? Clin Pharmacol Ther. 2010;87(4):396-8.

252. van Hall G, Steensberg A, Sacchetti M, Fischer C, Keller C, Schjerling P, et al. Interleukin-6 stimulates lipolysis and fat oxidation in humans. J Clin Endocrinol Metab. 2003;88(7):3005-10.

253. Hiscock N, Fischer CP, Sacchetti M, van Hall G, Febbraio MA, Pedersen BK. Recombinant human interleukin-6 infusion during low-intensity exercise does not enhance whole body lipolysis or fat oxidation in humans. Am J Physiol Endocrinol Metab. 2005;289(1):E2-7.

254. Nonogaki K, Fuller GM, Fuentes NL, Moser AH, Staprans I, Grunfeld C, et al. Interleukin-6 stimulates hepatic triglyceride secretion in rats. Endocrinology. 1995;136(5):2143-9.

255. Klover PJ, Zimmers TA, Koniaris LG, Mooney RA. Chronic exposure to interleukin-6 causes hepatic insulin resistance in mice. Diabetes. 2003;52(11):2784-9.

256. Kim HJ, Higashimori T, Park SY, Choi H, Dong J, Kim YJ, et al. Differential effects of interleukin-6 and -10 on skeletal muscle and liver insulin action in vivo. Diabetes. 2004;53(4):1060-7.

257. Franckhauser S, Elias I, Rotter Sopasakis V, Ferre T, Nagaev I, Andersson CX, et al. Overexpression of Il6 leads to hyperinsulinaemia, liver inflammation and reduced body weight in mice. Diabetologia. 2008;51(7):1306-16.

258. Klover PJ, Clementi AH, Mooney RA. Interleukin-6 depletion selectively improves hepatic insulin action in obesity. Endocrinology. 2005;146(8):3417-27.

259. Rotter Sopasakis V, Larsson BM, Johansson A, Holmang A, Smith U. Short-term infusion of interleukin-6 does not induce insulin resistance in vivo or impair insulin signalling in rats. Diabetologia. 2004;47(11):1879-87.

260. Weigert C, Hennige AM, Brodbeck K, Haring HU, Schleicher ED. Interleukin-6 acts as insulin sensitizer on glycogen synthesis in human skeletal muscle cells by phosphorylation of Ser473 of Akt. Am J Physiol Endocrinol Metab. 2005;289(2):E251-7.

261. Weigert C, Hennige AM, Lehmann R, Brodbeck K, Baumgartner F, Schauble M, et al. Direct cross-talk of interleukin-6 and insulin signal transduction via insulin receptor substrate-1 in skeletal muscle cells. J Biol Chem. 2006;281(11):7060-7.

262. Wallenius K, Wallenius V, Sunter D, Dickson SL, Jansson JO. Intracerebroventricular interleukin-6 treatment decreases body fat in rats. Biochem Biophys Res Commun. 2002;293(1):560-5.

263. Wallenius V, Wallenius K, Ahren B, Rudling M, Carlsten H, Dickson SL, et al. Interleukin-6-deficient mice develop mature-onset obesity. Nat Med. 2002;8(1):75-9.

264. Faldt J, Wernstedt I, Fitzgerald SM, Wallenius K, Bergstrom G, Jansson JO. Reduced exercise endurance in interleukin-6-deficient mice. Endocrinology. 2004;145(6):2680-6.

265. Fritsche L, Hoene M, Lehmann R, Ellingsgaard H, Hennige AM, Pohl AK, et al. IL-6 deficiency in mice neither impairs induction of metabolic genes in the liver nor affects blood glucose levels during fasting and moderately intense exercise. Diabetologia. 2010;53(8):1732-42. 266. Di Gregorio GB, Hensley L, Lu T, Ranganathan G, Kern PA. Lipid and carbohydrate metabolism in mice with a targeted mutation in the IL-6 gene: absence of development of age-related obesity. Am J Physiol Endocrinol Metab. 2004;287(1):E182-7.

267. Wernstedt I, Edgley A, Berndtsson A, Faldt J, Bergstrom G, Wallenius V, et al. Reduced stress- and cold-induced increase in energy expenditure in interleukin-6deficient mice. Am J Physiol Regul Integr Comp Physiol. 2006;291(3):R551-7.

268. Wunderlich FT, Strohle P, Konner AC, Gruber S, Tovar S, Bronneke HS, et al. Interleukin-6 signaling in liver-parenchymal cells suppresses hepatic inflammation and improves systemic insulin action. Cell Metab. 2010;12(3):237-49.

269. Sadagurski M, Norquay L, Farhang J, D'Aquino K, Copps K, White MF. Human IL6 enhances leptin action in mice. Diabetologia. 2010;53(3):525-35.

270. Cadoudal T, Blouin JM, Collinet M, Fouque F, Tan GD, Loizon E, et al. Acute and selective regulation of glyceroneogenesis and cytosolic phosphoenolpyruvate carboxykinase in adipose tissue by thiazolidinediones in type 2 diabetes. Diabetologia. 2007;50(3):666-75.

271. Pedersen BK, Febbraio MA. Muscle as an endocrine organ: focus on musclederived interleukin-6. Physiol Rev. 2008;88(4):1379-406.

272. Allen TL, Febbraio MA. IL6 as a mediator of insulin resistance: fat or fiction? Diabetologia. 2010;53(3):399-402.

273. Matthews VB, Allen TL, Risis S, Chan MH, Henstridge DC, Watson N, et al. Interleukin-6-deficient mice develop hepatic inflammation and systemic insulin resistance. Diabetologia. 2010;53(11):2431-41.

# Chapter 3 Epinephrine Mediated Regulation of PDK4 mRNA in Rat Adipose Tissue<sup>1</sup>

# **3.1 Introduction**

Adipose is a vibrant metabolic tissue secreting a wide range of hormones and metabolites that influence whole body fuel metabolism (1). The primary metabolic function of adipose tissue is to serve as a storage depot for lipids. In conditions of nutrient excess, glucose and fatty acids are stored in adipose tissue as triglycerides (TGs). When blood glucose levels become limiting, such as during fasting or prolonged exercise, fatty acids are released from TGs via lipolysis. While a large percentage of these fatty acids are released into the circulation, a significant amount are re-esterified back to TG (2). Re-esterification requires the provision of glycerol 3-phosphate (G-3-P) and, at least in rodent adipose tissue, the generation of G-3-P occurs primarily through *de novo* synthesis from sources such as lactate and pyruvate, in a process termed glyceroneogenesis (3-4). Since fluctuations in fatty acid levels have been linked to the development of systemic insulin resistance (5), an understanding of the factors which regulate glyceroneogenesis is vital in understanding the pathogenesis of type 2 diabetes.

Accumulating evidence suggests that pyruvate dehydrogenase kinase 4 (PDK4) is involved in the regulation of glyceroneogenesis in white adipose tissue (6). PDK4, along with the related enzymes PDK1, -2 and -3, phosphorylates and inhibits pyruvate dehydrogenase complexes (PDC) leading to decreases in the formation of acetyl CoA and

<sup>&</sup>lt;sup>1</sup> A version of this chapter has been published. Wan et al. Am J Physiol Cell Physiol. 2010. Nov;299(5):C1162-70.

a greater flux of pyruvate towards oxaloacetate and the glyceroneogenic pathway (6). Thiazolidinediones (TZDs), a commonly prescribed class of anti-diabetic drugs and potent peroxisome proliferator-activated receptor gamma (PPAR $\gamma$ ) agonists, increase PDK4 expression, lower fasting fatty acid levels and improve insulin sensitivity in insulin resistant Zucker *fa/fa* rats (6). Moreover, TZD treatment of adipocytes leads to increases in the expression of PDK4 concomitant with increased rates of glyceroneogenesis (6). PDK4 would appear to be essential for the up-regulation of glyceroneogenesis by TZDs as the attenuation of PDK4 activity pharmacologically with leelamine or reducing PDK4 content by short interfering RNA both result in decreased fatty acid re-esterification (6). Given the effects of TZDs on PDK4 expression and glyceroneogenesis (6-7) and owing to the unwanted side effects of these drugs (8) the identification of alternative approaches to induce PDK4 in white adipose tissue is warranted.

The regulation of PDK4 expression has been extensively studied in skeletal muscle and has been shown to be robustly induced by fasting (9) and exercise (10-12). The over-expression of the transcriptional coactivator  $1-\alpha$  (PGC- $1\alpha$ ) in rodent skeletal muscle (13) and C2C12 muscle cells (14) results in increases in PDK4 expression, thus lending credence to the supposition that this molecule is a key regulator of PDK4. Interestingly, we have recently made the novel discovery that exercise and epinephrine increase PGC- $1\alpha$  mRNA expression in rat epididymal adipose tissue (15). Moreover,  $\beta$ -adrenergic agonists have been reported to increase the activity of the 5'AMP activated protein kinase (AMPK) (16-18) and p38 mitogen-activated protein kinase (MAPK) pathways in adipocytes (19-20), signaling cascades which have been implicated in the regulation of PDK4 (10, 21-22). Taken in concert, these findings suggest a role for  $\beta$ -

adrenergic agonists in the regulation of PDK4 in white adipose tissue. In this context, we hypothesized that fasting, exercise and epinephrine would increase PDK4 mRNA expression in rat epididymal adipose tissue. We further surmised that epinephrine would increase p38 and AMPK signaling and these pathways would be involved in the regulation of PDK4 in rat adipose tissue. Lastly, since PPAR $\gamma$  agonists have been implicated in regulating PDK4 mRNA expression, we postulated that this transcription factor would also be involved in the pathway through which epinephrine increases PDK4 in adipose tissue.

#### **3.2 Methods and Materials**

#### **3.2.1 Materials**

RNeasy kit was purchased from Qiagen (Toronto, ON). SuperScript II Reverse Transcriptase, oligo(dT) and dNTP were products from Invitrogen (Burlington, ON). Taqman Gene Expression Assays for beta 2 microglobulin (β2MG), beta actin, PDK1, PDK2, PDK3, PDK4, pyruvate dehydrogenase phosphatase 1 (PDP1) and PDP2 were from Applied Biosystems (Foster City, CA). 5-Aminoimidazole-4-carboxyamide ribonucleoside (AICAR) was purchased from Toronto Research Chemicals (Toronto, ON) while rosiglitazone was obtained from Cayman (Ann Arbor, MI). Total and phosphorylation specific antibodies for the detection of AMPK, ACC acetyl coA carboxylase (ACC), p38 MAPK, MAPK activated protein kinase 2 (MK2) and hormone sensitive lipase (HSL) were from Cell Signaling (Danvers, MA). Phosphorylated pyruvate dehydrogenase (PDH) antibodies and SB202190/SB203580 were products of EMD Biosciences (San Diego, CA). All other chemicals were purchased from Sigma (Oakville, ON).

#### **3.2.2 Treatment of Rodents**

All protocols followed Canadian Council on Animal Care (CCAC) guidelines and were approved by the Animal Use and Welfare Committee at the University of Alberta. Male Wistar rats (Charles River, Wilmington MA) weighing ~200g were housed 2 per cage, with a 12/12-hour light/dark cycle, and were provided with water and standard rat chow ad libitum. The 12-hour light cycle was from 6am to 6pm and all experimental protocols were performed between 7am and 10am. Swim exercise and epinephrine injections were performed as described previously by our laboratory (15). Briefly, rats were acclimatized to the animal housing unit for 1 week prior to experiments. For the exercise studies, half of the animals swam for 2 hours while the remaining animals served as controls, which swam for 10 min to control for exposure to water. For the epinephrine experiment, rats were injected intraperitoneally (IP) with a weight adjusted bolus of epinephrine (20  $\mu$ g/100 gram body weight) or an equivalent volume of sterile saline. To determine the effects of fasting on PDK4 mRNA levels in adipose tissue, food was removed at ~ 5:00 pm the evening before the experiment. Immediately following exercise, 2 hours after epinephrine injections or ~18 hours following the removal of food animals were anesthesized with sodium pentabaritol (5mg/100g body weight), epididymal adipose tissue was dissected free of the testes, immediately weighed and then clamp frozen in tongs cooled to the temperature of liquid nitrogen and stored at -80  $^{\circ}$ C until further analysis.

#### 3.2.3 Adipose Tissue Organ Culture

Adipose tissue organ culture (ATOC) is a well characterized technique that has been used to determine changes in adipose tissue metabolism and gene expression (23). Epididymal adipose tissue was cultured as we have described in detail previously (15). Briefly, 250 mg of epididymal adipose tissue was minced into  $\sim$  5-10 mg pieces and placed into culture dishes containing 7.5 ml of M199 medium supplemented with 1% antibiotic/antimycotic, 50 µU insulin and 2.5 nM dexamethasone. The epididymal adipose tissue was then kept in an incubator at 37  $\,^{\circ}$ C to equilibrate for 24 h. To assess the effects of epinephrine (1  $\mu$ M), AICAR (1 mM) and metformin (1mM) on the expression of PDK4 mRNA, adipose tissue was cultured for 24 hours and then treated with these agents for 2 hours. To determine the effects of p38 inhibition on PDK4 expression, cultures were treated with 1 µM SB202190 or SB203580 for 30 minutes prior to and during the 2 hour epinephrine treatment. These compounds are cell permeable inhibitors of p38 MAPK that binds to the ATP pocket of the active kinase. We chose this concentration of SB202190/SB203580 since previous studies have demonstrated that higher concentrations of the pyridinyl imidazole compounds may possess some nonspecific effects (24). For the determination of changes in the phosphorylation of AMPK, ACC, p38, MK2 and HSL experiments were conducted as described above except cultures were treated with epinephrine for 30 minutes. At the end of the experiments adipose tissue cultures were rinsed in ice cold sterile PBS, strained and adipose tissue fragments snap frozen and stored at -80 ° C for further analysis.

### **3.2.4 Western Blot Analysis**

Protein was extracted from adipose tissue and changes in the phosphorylation status of AMPK, ACC, p38, MK2 and HSL determined by Western blotting, as described in detail by our laboratory previously (25). Briefly, adipose tissue samples were homogenized in 1.5 volumes of ice-cold cell lysis buffer supplemented with Protease Inhibitor Cocktail (Sigma) and phenylmethylsulfonyl fluoride. Homogenized samples were centrifuged for 10 min at 1500 X G at 4 °C. The fat cake was removed and the infranatant was collected and protein concentration determined using the BCA method (26). Equal amounts of protein were separated on 7.5% gels. Proteins were wet transferred to nitrocellulose membranes at 200mA/tank and subsequently blocked in Tris buffered saline/0.1% Tween 20 (TBST) supplemented with 5% (W/V) non-fat dry milk for 1 hour at room temperature with gentle agitation. Membranes were incubated in TBST/5% non-fat dry milk supplemented with appropriate primary antibodies overnight at 4°C with gentle agitation. The following morning, membranes were briefly washed in TBST and then incubated in TBST/1% non-fat dry milk supplemented with HRP conjugated secondary antibodies for 1 hour at room temperature. Bands were visualized using ECL plus and captured using a Typhoon Imaging system (GE Health Care, Baie d'Urfe Quebec). ImageQuant (GE Healthcare), software was used to quantify relative band intensities. Beta actin or tubulin were used as internal controls.

# **3.2.5 Lipolysis Assays**

Adipose tissue was cultured as described above in *section 3.2.3*. On the morning of the experiment media were removed and replaced with fresh M199 supplemented with 3% (W/V) fatty acid-free BSA. Epinephrine, SB202190 or dichloroacetate (DCA) were

added as described above. Media was collected at the end of the 2 hour epinephrine treatment and adipose tissue fragments weighed and frozen in liquid nitrogen. Glycerol release into the media was determined using a Free Glycerol Kit (Sigma, St. Louis, MO). The coefficient of variation for this assays in our laboratory is <10%.

# 3.2.6 Real Time RT-PCR

RNA was isolated from epididymal adipose tissue, complementary DNA (cDNA) synthesized and PCR analysis performed as we have described in detail previously (15, 25). Briefly, RNA was isolated from epididymal adipose tissue using an RNeasy kit according to the manufacturer's instructions. One microgram of RNA was used for the synthesis of cDNA using SuperScript II Reverse Transcriptase, oligo(dT) and dNTP. Real-time PCR was performed using a 7900HT Fast Real-Time PCR system (Applied Biosystems). A Relative difference in gene expression between groups was determined using the  $2^{-\Delta\Delta CT}$  method (27). The amplification efficiencies of the gene of interest and the housekeeping gene were equivalent and there was no effect of the experimental manipulation on the expression of housekeeping genes.

# **3.2.7 Statistical Analysis**

Data are presented as means  $\pm$  standard error of the mean (SEM). Comparisons between control and exercised rats were made using a one-way analysis of variance (ANOVA) followed by a post-hoc comparison using Fisher's LSD test. Comparisons between fed and fasted rats, vehicle and epinephrine, AICAR or metformin treated cultures were made using a Student's T-test. The effects of SB202190 on cell signaling and lipolysis were analyzed using a 2 X 2 ANOVA with LSD post-hoc comparisons. The effects of SB202190 or SB203580 on PDK4 mRNA expression were analyzed using a one way ANOVA on ranks followed by Dunn's test. Statistical significance was set at p<0.05.

#### **3.3 Results**

# 3.3.1 Exercise, Fasting and Epinephrine Increase PDK4 mRNA Expression in Epididymal Adipose Tissue

PDK4 mRNA expression was increased  $\sim 6$  fold immediately following 2 hours of swim exercise (*Figure 3-1A*) and remained elevated 2 ( $3.83 \pm 0.32$  fold increase versus control, p < 0.05) and 4 hours (3.34 ± 0.70 fold increase versus control, p < 0.05) following exercise cessation. Similarly, an overnight fast led to a  $\sim 3$  fold increase in PDK4 mRNA levels. PDK4 mRNA levels were increased in epididymal adipose tissue 2 hours following a bolus injection of epinephrine (20  $\mu$ g/100 gram body weight) and were also increased in cultured epididymal adipose tissue following a 2 hour epinephrine (1 µM epinephrine) treatment. Exercise, fasting and epinephrine did not increase mRNA levels of PDK1, 2 or 3 (Figure 3-1A). Similarly the mRNA expression of PDP1 and 2, enzymes which dephosphorylate PDH complexes, were not altered by any of our experimental manipulations (data not shown). Increases in PDK4 mRNA levels after an overnight fast or 4 hours following exercise cessation were associated with increases in the phosphorylation of PDH on serine 293 and serine 300, PDK4 phosphorylation sites (28) (Figure 3-1B and C). Confirming the importance of PDK4 in regulating fatty acid release (6), we found that DCA, a PDK4 inhibitor (28), increased the ratio (µmol/g tissue fatty acid to  $\mu$ mol/g tissue glycerol) of fatty acid to glycerol release in cultured adipose tissue (1.76 ±0.10 control, 2.67 ±0.45 DCA, *p* =0.05).


Figure 3-1 Effects of exercise, fasting and epinephrine on PDK4 mRNA and the phosphorylation of PDH protein expression. A) Exercise, fasting and epinephrine increase the mRNA expression of PDK4, but not PDK1, PDK2 or PDK3 in rat epididymal adipose tissue. Increases in PDK4 mRNA levels are associated with increases in the phosphorylation of PDH on serine 293 and 300 B) after an overnight fast and C) 4 hours following exercise cessation. Data are presented as means + SEM for 5-11 animals or tissue cultures per group. The mRNA data is normalized to beta actin and expressed as fold differences compared to sedentary controls, fed controls, saline injected animals or vehicle treated tissue cultures. Representative Western blot images are given to the right of the quantified data in B and C. \* p < 0.05 versus control.

# 3.3.2 Epinephrine Increases AMPK and P38 MAPK Signaling in Cultured

## **Epididymal Adipose Tissue**

As illustrated in *Figure 3-2*, epinephrine treatment (1  $\mu$ M, 30 minutes) led to increases in the phosphorylation of AMPK and its downstream substrate ACC (29). Similarly, epinephrine also increased the activity of the p38 MAPK signaling pathway as evidenced by increases in the phosphorylation of p38 and MK2. p38 MAPK phosphorylates MK2 on threonine-222 and thus MK2 phosphorylation can be used as a marker of p38 MAPK activity (30).



Figure 3-2 Effects of *ex vivo* epinephrine on p38 and AMPK signalling pathways. *Ex vivo* epinephrine treatment (30 minutes, 1µM) increases the phosphorylation of p38, MK2, AMPK and ACC in cultured rat epididymal adipose tissue. Data are presented as means + SEM for 5-8 tissue cultures per group. Representative Western blot images are given to the right of the quantified data. \*p < 0.05 versus control.

# 3.3.3 AMPK Agonists Do Not Increase PDK4 mRNA Expression in Rat Epididymal Adipose Tissue

As illustrated in *Figure 3-3A and B*, AICAR (1 mM, 1 hour) and metformin (1 mM, 1 hour) both increased AMPK and ACC phosphorylation while having no effect on p38 MAPK phosphorylation. These compounds decreased PDK4 mRNA expression (*Figure 3-3C*).



Figure 3-3 Effects of *ex vivo* AICAR and metformin on AMPK and p38 signalling pathways and PDK4 mRNA. The AMPK agonists A) AICAR (1 mM) and B) metformin (1 mM) increase AMPK signaling (1 hour treatment) in cultured epididymal adipose tissue but do not increase the expression of C) PDK4 mRNA levels (2 hour treatment). Data are presented as means + SEM for 5-11 samples per group. RT-PCR data was normalized to beta actin. Representative Western blot images are given to the right of the quantified Western blot data. \*p < 0.05 versus control. (For figure 3-3C, the relative PDK4 mRNA expression in vehicle treated group for the *ex vivo* AICAR and metformin treatment experiment is 1.10 ± 0.49.)

# 3.3.4 SB202190 Does Not Attenuate AMPK or PKA Signaling in Epididymal

# Adipose Tissue Cultures

Treating cultured epididymal adipose tissue with 1  $\mu$ M SB202190 for 30 minutes prior to and during epinephrine treatment led to reductions in MK2 phosphorylation (*Figure 3-4A*). SB202190 did not inhibit the phosphorylation of HSL on serine 563 and 660 (*Figure 3-4B*) which are protein kinase A (PKA) phosphorylation sites (31). Similarly AMPK phosphorylation was not inhibited by SB202190 (*Figure 3-4C*). Epinephrine-induced increases in lipolysis, as determined by glycerol release into the culture media, was not reduced by prior treatment with SB202190 (*Figure 3-4D*).

A p-MK2



Figure 3-4 Effects of ex vivo SB on the phosphorylation of MK2, AMPK, HSL, lipolysis and PDK4 mRNA. SB202190 inhibits epinephrine induced increases in A) MK2 phosphorylation but not B) HSL phosphorylation (serine 563 and 660) C) AMPK phosphorylation or D) lipolysis. E) Epinephrine mediated increases in PDK4 mRNA expression are abrogated by SB202190 and SB203580. Data are presented as means + SEM for 5-15 tissue cultures per group. The RT-PCR data is normalized to beta actin and expressed as fold differences compared to controls. Representative Western blot images are given above

the quantified data. \* p < 0.05 versus vehicle treated control, <sup>#</sup> p < 0.05 versus SB202190 treated epinephrine condition. (For figure 3-4E, the relative PDK4 mRNA expression in vehicle treated group is  $1.31\pm0.64$ .)

# 3.3.5 SB202190 Attenuates Epinephrine Induced Increases in PDK4 mRNA

Epinephrine-induced increases in PDK4 mRNA were blunted by SB202190 (*Figure 3-4E*, as shown above). Similarly the related compound SB203580 (1  $\mu$ M) also blocked the effects of epinephrine on PDK4 mRNA levels. In the absence of epinephrine these compounds had no effect on PDK4 mRNA (data not shown).

## 3.3.6 Epinephrine Mediated Increases in PDK4 Are Inhibited by PPARy Antagonist

The PPAR $\gamma$  agonist rosiglitazone (2  $\mu$ M) increased PDK4 mRNA levels in cultured adipose tissue and this effect was abrogated by GW9662 (20  $\mu$ M), a PPAR $\gamma$  antagonist (*Figure 3-5A*). These reductions did not appear to be secondary to decreases in  $\beta$  adrenergic signaling as GW9662 did not affect epinephrine mediated increases in the phosphorylation of p38 MAPK, AMPK or HSL (*Figure 3-5B*).



Β.



Figure 3-5 Effects of *ex vivo* GW9662 on rosiglitazone and epinephrine mediated increases in PDK4 mRNA and signalling. GW9662 (20 $\mu$ M), a PPAR $\gamma$  antagonist, inhibits A) rosiglitazone (2 $\mu$ M) and epinephrine (1  $\mu$ M) mediated increases in PDK4 mRNA levels while having no effect on epinephrine mediated increases in the phosphorylation of B) p38, AMPK, or HSL. Data are presented as means + SEM for 6-8 cultures per group. The RT-PCR data is normalized to beta actin and expressed as fold differences compared to control. Representative Western blot images are given to the right of the quantified data. \* *p* 

Α.

<0.05 versus control, <sup>#</sup>p <0.05 versus epinephrine + GW9662. (For figure 3-5A, the relative PDK4 mRNA expression in vehicle treated group for *ex vivo* rosiglitazone and epinephrine treatment is 1.01 ± 0.07 and 1.09 ± 0.09 respectively.)

# **3.4 Discussion**

Adipose tissue is an active metabolic organ that is intimately involved in the regulation of systemic carbohydrate and lipid metabolism (1). Accumulating evidence suggests that the modulation of white adipose tissue metabolism is a beneficial approach in the treatment of type 2 diabetes. In this regard it has been demonstrated that TZDs increase fatty acid re-esterification and limit fatty acid release from white adipocytes (6). Cadoudal and collegues (6) have reported that the inhibition or knockdown of PDK4 attenuates the effects of TZDs on glyceroneogenesis thus demonstrating the essential role of this enzyme in fatty acid recycling. Given the key role of PDK4 in the regulation of white adipose tissue metabolism, we sought to explore the regulation of this kinase by physiological stimuli such as exercise, fasting and epinephrine. We have made the novel discovery that exercise leads to rapid increases in PDK4 mRNA levels that persist for upwards of 4 hours following exercise cessation. These findings are similar in magnitude and time course to the reported effects of exercise on PDK4 mRNA levels in skeletal muscle (10-12) and further demonstrate (15) the plasticity of adipose tissue in response to exercise. In addition to exercise, and consistent with one previous report (32), we found that PDK4 mRNA levels were also increased by fasting. In contrast to PDK4, PDK1, 2 and 3, were not responsive to these perturbations. These findings are congruent with previous studies which demonstrated negligible changes in PDK2 levels in skeletal muscle following fasting (33). In addition to increasing PDK4 mRNA levels, we found that the phosphorylation of PDH on serine 293 and 300 was increased after an overnight fast or 4 hours following exercise cessation. Since PDK4 phosphorylates these residues (28), these findings suggest an increase in PDK4 activity.

Swim exercise (34) and fasting (35) increase circulating epinephrine levels. To determine if epinephrine could be a signal which initiates increases in PDK4 mRNA levels, we examined PDK4 in adipose tissue harvested from rats two hours following epinephrine injections or in adipose tissue cultures treated with epinephrine. Similar to what has previously been reported in mouse skeletal muscle following treatment with the  $\beta_2$ -adrenergic agonist formoterol (36), we found that epinephrine increased PDK4 mRNA levels both *in vivo* and *ex vivo*.

In isolated adipocytes  $\beta$ -adrenergic agonists increase the activity of AMPK (16-17, 37). Since AMPK has been postulated to regulate PDK4 mRNA expression in skeletal muscle (10, 21), we reasoned that this kinase may serve a similar role in white adipose tissue. Consistent with this line of thinking we found that *ex vivo* epinephrine treatment increased the phosphorylation of AMPK and ACC concomitant with increases in PDK4 mRNA expression. To determine if the direct activation of AMPK could increase PDK4 mRNA levels we treated adipose tissue cultures with AICAR or metformin. AICAR is an adenosine analogue which is converted to ZMP and mimics the effects of AMP on AMPK activation (38) while metformin activates AMPK independent of changes in the AMP:ATP ratio (38). Despite increasing AMPK and ACC phosphorylation we demonstrate that these mechanistically distinct AMPK agonists did not increase, and in fact reduced, PDK4 mRNA expression in cultured epididymal adipose tissue. In adipocytes the activation of AMPK results in the phosphorylation of HSL (39-41) on serine 565, leading to decreases in HSL activity and reductions in lipolysis (42-44). Since rates of fatty acid re-esterification change in proportion to lipolysis (39, 45) it is not entirely surprising that AMPK agonists would down regulate the expression of PDK4 given reductions in lipolysis, and presumably by extension, decreases in the absolute rate of fatty acid re-esterification. Regardless of the mechanisms involved these findings clearly demonstrate tissue specific differences in the mechanisms regulating PDK4.

In addition to AMPK, catecholamines have also been shown to activate p38 MAPK signaling in adipocytes (19-20). Since this pathway has been implicated in modulating the expression of PDK4 in cardiomyocytes (22), we postulated that p38 MAPK would regulate PDK4 in white adipose tissue. Consistent with this hypothesis we found that the p38 MAPK inhibitor SB202190 attenuated the epinephrine-induced increases in MK2 phosphorylation and PDK4 mRNA expression. While the pyridinyl imidazole compounds are potent inhibitors of p38 MAPK (46), these agents have also been shown to possess non-specific effects (24, 47). In this light we thought it vital to determine if SB202190 impaired epinephrine signaling, independent of reductions in p38 MAPK activity. Under the conditions utilized we found that SB202190 did not reduce epinephrine-induced increases in the phosphorylation of HSL on serine 660 and 563 which are specific PKA phosphorylation sites (31) nor did the compound inhibit epinephrine-induced increases in lipolysis. Similarly, SB202190 did not attenuate epinephrine-induced increases in AMPK phosphorylation. Collectively, these results strongly suggest that inhibition of p38 MAPK, and not a general abrogation of  $\beta$ adrenergic signaling mediates the effects of SB202190 on epinephrine-induced increases in PDK4 mRNA levels in white adipose tissue.

In addition to providing novel information regarding the processes which regulate PDK4 mRNA expression in white adipose tissue our findings also lend insight into the relationship between AMPK and p38 MAPK signaling in this tissue. It has been suggested by some (48) but not all (49) that AMPK may lie upstream of p38 MAPK. Conversely, it has also been demonstrated that SB202190, at levels ~10 fold higher than used in the current study, attenuates AMPK activity in heart muscle (50). Contrary to these studies we did not find an interaction between p38 and AMPK signaling in cultured white adipose tissue. For instance, AICAR and metformin did not increase the phosphorylation of p38 MAPK while the inhibition of p38 MAPK signaling was not associated with reductions in AMPK phosphorylation. Collectively these results imply that epinephrine increases the activation of AMPK and p38 MAPK through parallel, not serial pathways in white adipose tissue.

Previous work has demonstrated that rosiglitazone causes a rapid increase in PDK4 mRNA levels in rodent adipose tissue (6). Since the time-course and magnitude of change in PDK4 mRNA with rosiglitizone are similar to what we see following epinephrine treatment we reasoned that PPAR $\gamma$  may be involved in the pathway through which epinephrine-induces PDK4 in white adipose tissue. Consistent with this hypothesis, and in line with work from Caudoudal and associates (6), we found that rosiglitazone, a potent PPAR $\gamma$  agonist, increased the expression of PDK4. Conversely, GW9662, a PPAR $\gamma$  antagonist prevented rosiglitazone and epinephrine-mediated increases in PDK4. The reduction in epinephrine-induced PDK4 was not secondary to attenuated  $\beta$ -adrenergic signaling as epinephrine-induced increases in p38, AMPK and HSL phosphorylation were unaffected by GW9662. In combination with the p38

inhibition experiments, these findings provide evidence suggesting the involvement of p38 MAPK and PPAR $\gamma$  in regulating epinephrine mediated increases in PDK4. What is not clear at this juncture is whether there is a direct link between p38 activation, PPAR $\gamma$  and increases in PDK4 or whether both p38 MAPK and PPAR $\gamma$  are required for the induction of PDK4 by epinephrine, but do so through distinct pathways. While speculative, the activation of p38 MAPK may activate PGC-1 $\alpha$  (51), which in turn may bind to and co-activate PPAR $\gamma$  thus leading to the induction of PDK4 in white adipose tissue. Clearly, this is a question that needs to be explored in greater detail in the future.

In summary we have made the novel observation that exercise, fasting and epinephrine increase the mRNA expression of PDK4 in rat epididymal adipose tissue. We have further demonstrated a disconnect between the activation of AMPK and the induction of PDK4 mRNA and provide data suggesting that p38 MAPK and PPAR $\gamma$  are involved in the regulation of PDK4 mRNA levels in adipose tissue. These findings provide novel information regarding the biochemical processes regulating PDK4 and importantly highlight alternative, non-pharmaceutical based approaches that can be used to induce PDK4 mRNA expression.

# **3.5 Literature Cited**

1. Rosen ED, Spiegelman BM. Adipocytes as regulators of energy balance and glucose homeostasis. Nature. 2006;444(7121):847-53.

2. Wang T, Zang Y, Ling W, Corkey BE, Guo W. Metabolic partitioning of endogenous fatty acid in adipocytes. Obes Res. 2003;11(7):880-7.

3. Nye C, Kim J, Kalhan SC, Hanson RW. Reassessing triglyceride synthesis in adipose tissue. Trends Endocrinol Metab. 2008;19(10):356-61.

 Nye CK, Hanson RW, Kalhan SC. Glyceroneogenesis is the dominant pathway for triglyceride glycerol synthesis in vivo in the rat. J Biol Chem. 2008;283(41):27565-74.

5. Boden G, Shulman GI. Free fatty acids in obesity and type 2 diabetes: defining their role in the development of insulin resistance and beta-cell dysfunction. Eur J Clin Invest. 2002;32 Suppl 3:14-23.

6. Cadoudal T, Distel E, Durant S, Fouque F, Blouin JM, Collinet M, et al. Pyruvate dehydrogenase kinase 4: regulation by thiazolidinediones and implication in glyceroneogenesis in adipose tissue. Diabetes. 2008;57(9):2272-9.

7. Sears DD, Hsiao A, Ofrecio JM, Chapman J, He W, Olefsky JM. Selective modulation of promoter recruitment and transcriptional activity of PPARgamma. Biochem Biophys Res Commun. 2007;364(3):515-21.

8. Stafylas PC, Sarafidis PA, Lasaridis AN. The controversial effects of thiazolidinediones on cardiovascular morbidity and mortality. Int J Cardiol. 2009;131(3):298-304.

104

9. Sanchez J, Palou A, Pico C. Response to carbohydrate and fat refeeding in the expression of genes involved in nutrient partitioning and metabolism: striking effects on fibroblast growth factor-21 induction. Endocrinology. 2009;150(12):5341-50.

10. Jorgensen SB, Wojtaszewski JF, Viollet B, Andreelli F, Birk JB, Hellsten Y, et al. Effects of alpha-AMPK knockout on exercise-induced gene activation in mouse skeletal muscle. FASEB J. 2005;19(9):1146-8.

11. Pilegaard H, Keller C, Steensberg A, Helge JW, Pedersen BK, Saltin B, et al. Influence of pre-exercise muscle glycogen content on exercise-induced transcriptional regulation of metabolic genes. J Physiol. 2002;541(Pt 1):261-71.

12. Pilegaard H, Ordway GA, Saltin B, Neufer PD. Transcriptional regulation of gene expression in human skeletal muscle during recovery from exercise. Am J Physiol Endocrinol Metab. 2000;279(4):E806-14.

13. Wende AR, Huss JM, Schaeffer PJ, Giguere V, Kelly DP. PGC-1alpha coactivates PDK4 gene expression via the orphan nuclear receptor ERRalpha: a mechanism for transcriptional control of muscle glucose metabolism. Mol Cell Biol. 2005;25(24):10684-94.

14. Wende AR, Schaeffer PJ, Parker GJ, Zechner C, Han DH, Chen MM, et al. A role for the transcriptional coactivator PGC-1alpha in muscle refueling. J Biol Chem. 2007;282(50):36642-51.

15. Sutherland LN, Bomhof MR, Capozzi LC, Basaraba SA, Wright DC. Exercise and adrenaline increase PGC-1{alpha} mRNA expression in rat adipose tissue. J Physiol. 2009;587(Pt 7):1607-17.

105

16. Gauthier MS, Miyoshi H, Souza SC, Cacicedo JM, Saha AK, Greenberg AS, et al. AMP-activated protein kinase is activated as a consequence of lipolysis in the adipocyte: potential mechanism and physiological relevance. J Biol Chem. 2008;283(24):16514-24.

17. Koh HJ, Hirshman MF, He H, Li Y, Manabe Y, Balschi JA, et al. Adrenaline is a critical mediator of acute exercise-induced AMP-activated protein kinase activation in adipocytes. Biochem J. 2007;403(3):473-81.

18. Park H, Kaushik VK, Constant S, Prentki M, Przybytkowski E, Ruderman NB, et al. Coordinate regulation of malonyl-CoA decarboxylase, sn-glycerol-3-phosphate acyltransferase, and acetyl-CoA carboxylase by AMP-activated protein kinase in rat tissues in response to exercise. J Biol Chem. 2002;277(36):32571-7.

19. Cao W, Medvedev AV, Daniel KW, Collins S. beta-Adrenergic activation of p38 MAP kinase in adipocytes: cAMP induction of the uncoupling protein 1 (UCP1) gene requires p38 MAP kinase. J Biol Chem. 2001;276(29):27077-82.

20. Moule SK, Denton RM. The activation of p38 MAPK by the beta-adrenergic agonist isoproterenol in rat epididymal fat cells. FEBS Lett. 1998;439(3):287-90.

21. McAinch AJ, Cameron-Smith D. Adiponectin decreases pyruvate dehydrogenase kinase 4 gene expression in obese- and diabetic-derived myotubes. Diabetes Obes Metab. 2009;11(7):721-8.

22. Puthanveetil P, Wang Y, Wang F, Kim MS, Abrahani A, Rodrigues B. The increase in cardiac pyruvate dehydrogenase kinase-4 after short-term dexamethasone is controlled by an Akt-p38-forkhead box other factor-1 signaling axis. Endocrinology. 2010;151(5):2306-18.

Fried SK, Moustaid-Moussa N. Culture of adipose tissue and isolated adipocytes.
Methods Mol Biol. 2001;155:197-212.

24. Antonescu CN, Huang C, Niu W, Liu Z, Eyers PA, Heidenreich KA, et al. Reduction of insulin-stimulated glucose uptake in L6 myotubes by the protein kinase inhibitor SB203580 is independent of p38MAPK activity. Endocrinology. 2005;146(9):3773-81.

25. Frier BC, Williams DB, Wright DC. The effects of apelin treatment on skeletal muscle mitochondrial content. Am J Physiol Regul Integr Comp Physiol. 2009;297(6):R1761-8.

 Smith PK, Krohn RI, Hermanson GT, Mallia AK, Gartner FH, Provenzano MD,
et al. Measurement of protein using bicinchoninic acid. Anal Biochem. 1985;150(1):76-85.

27. Livak KJ, Schmittgen TD. Analysis of relative gene expression data using realtime quantitative PCR and the 2(-Delta Delta C(T)) Method. Methods. 2001;25(4):402-8.

28. Rardin MJ, Wiley SE, Naviaux RK, Murphy AN, Dixon JE. Monitoring phosphorylation of the pyruvate dehydrogenase complex. Anal Biochem. 2009;389(2):157-64.

29. Ha J, Daniel S, Broyles SS, Kim KH. Critical phosphorylation sites for acetyl-CoA carboxylase activity. J Biol Chem. 1994;269(35):22162-8.

Ben-Levy R, Leighton IA, Doza YN, Attwood P, Morrice N, Marshall CJ, et al.
Identification of novel phosphorylation sites required for activation of MAPKAP kinase EMBO J. 1995;14(23):5920-30.

31. Anthonsen MW, Ronnstrand L, Wernstedt C, Degerman E, Holm C. Identification of novel phosphorylation sites in hormone-sensitive lipase that are phosphorylated in response to isoproterenol and govern activation properties in vitro. J Biol Chem. 1998;273(1):215-21.

32. Li RY, Zhang QH, Liu Z, Qiao J, Zhao SX, Shao L, et al. Effect of short-term and long-term fasting on transcriptional regulation of metabolic genes in rat tissues. Biochem Biophys Res Commun. 2006;344(2):562-70.

33. Peters SJ, Harris RA, Heigenhauser GJ, Spriet LL. Muscle fiber type comparison of PDH kinase activity and isoform expression in fed and fasted rats. Am J Physiol Regul Integr Comp Physiol. 2001;280(3):R661-8.

34. Higashida K, Higuchi M, Terada S. Potential role of lipin-1 in exercise-induced mitochondrial biogenesis. Biochem Biophys Res Commun. 2008;374(3):587-91.

35. Frier BC, Jacobs RL, Wright DC. Interactions between the consumption of a highfat diet and fasting in the regulation of fatty acid oxidation enzyme gene expression: an evaluation of potential mechanisms. Am J Physiol Regul Integr Comp Physiol. 2011;300(2):R212-21.

36. Pearen MA, Myers SA, Raichur S, Ryall JG, Lynch GS, Muscat GE. The orphan nuclear receptor, NOR-1, a target of beta-adrenergic signaling, regulates gene expression that controls oxidative metabolism in skeletal muscle. Endocrinology. 2008;149(6):2853-65.

37. Gaidhu MP, Anthony NM, Patel P, Hawke TJ, Ceddia RB. Dysregulation of lipolysis and lipid metabolism in visceral and subcutaneous adipocytes by high-fat diet: role of ATGL, HSL, and AMPK. Am J Physiol Cell Physiol. 2010;298(4):C961-71.

38. Fryer LG, Parbu-Patel A, Carling D. The Anti-diabetic drugs rosiglitazone and metformin stimulate AMP-activated protein kinase through distinct signaling pathways. J Biol Chem. 2002;277(28):25226-32.

39. Brooks B, Arch JR, Newsholme EA. Effects of hormones on the rate of the triacylglycerol/fatty acid substrate cycle in adipocytes and epididymal fat pads. FEBS Lett. 1982;146(2):327-30.

40. Daval M, Diot-Dupuy F, Bazin R, Hainault I, Viollet B, Vaulont S, et al. Antilipolytic action of AMP-activated protein kinase in rodent adipocytes. J Biol Chem. 2005;280(26):25250-7.

41. Garton AJ, Campbell DG, Carling D, Hardie DG, Colbran RJ, Yeaman SJ. Phosphorylation of bovine hormone-sensitive lipase by the AMP-activated protein kinase. A possible antilipolytic mechanism. Eur J Biochem. 1989;179(1):249-54.

42. Corton JM, Gillespie JG, Hawley SA, Hardie DG. 5-aminoimidazole-4carboxamide ribonucleoside. A specific method for activating AMP-activated protein kinase in intact cells? Eur J Biochem. 1995;229(2):558-65.

43. Djouder N, Tuerk RD, Suter M, Salvioni P, Thali RF, Scholz R, et al. PKA phosphorylates and inactivates AMPKalpha to promote efficient lipolysis. EMBO J. 2010;29(2):469-81.

44. Sullivan JE, Brocklehurst KJ, Marley AE, Carey F, Carling D, Beri RK. Inhibition of lipolysis and lipogenesis in isolated rat adipocytes with AICAR, a cellpermeable activator of AMP-activated protein kinase. FEBS Lett. 1994;353(1):33-6.

45. Vaughan M. The production and release of glycerol by adipose tissue incubated in vitro. J Biol Chem. 1962;237:3354-8.

46. Davies SP, Reddy H, Caivano M, Cohen P. Specificity and mechanism of action of some commonly used protein kinase inhibitors. Biochem J. 2000;351(Pt 1):95-105.

47. Huang M, Wang Y, Collins M, Gu JJ, Mitchell BS, Graves LM. Inhibition of nucleoside transport by p38 MAPK inhibitors. J Biol Chem. 2002;277(32):28364-7.

48. Lemieux K, Konrad D, Klip A, Marette A. The AMP-activated protein kinase activator AICAR does not induce GLUT4 translocation to transverse tubules but stimulates glucose uptake and p38 mitogen-activated protein kinases alpha and beta in skeletal muscle. FASEB J. 2003;17(12):1658-65.

49. Ho RC, Fujii N, Witters LA, Hirshman MF, Goodyear LJ. Dissociation of AMPactivated protein kinase and p38 mitogen-activated protein kinase signaling in skeletal muscle. Biochem Biophys Res Commun. 2007;362(2):354-9.

50. Jaswal JS, Gandhi M, Finegan BA, Dyck JR, Clanachan AS. p38 mitogenactivated protein kinase mediates adenosine-induced alterations in myocardial glucose utilization via 5'-AMP-activated protein kinase. Am J Physiol Heart Circ Physiol. 2007;292(4):H1978-85.

51. Puigserver P, Rhee J, Lin J, Wu Z, Yoon JC, Zhang CY, et al. Cytokine stimulation of energy expenditure through p38 MAP kinase activation of PPARgamma coactivator-1. Mol Cell. 2001;8(5):971-82.

# Chapter 4 Epinephrine Induces PDK4 mRNA Expression in Adipose Tissue from Obese, Insulin Resistant Rats<sup>2</sup>

# 4.1 Introduction

Thiazolidinediones (TZDs) are a commonly prescribed medication used in the treatment of type 2 diabetes (T2DM). These PPAR $\gamma$  agonists mediate a large portion of their beneficial effects on insulin sensitivity through increasing fatty acid re-esterification in adipose tissue, reducing fatty acid efflux and limiting ectopic lipid deposition (1). Increases in fatty acid re-esterification require the formation of glycerol-3-phosphate (G3P) and this occurs primarily through the conversion of non-glucose molecules such as pyruvate in a process termed glyceroneogenesis (2).

Pyruvate dehydrogenase kinase 4 (PDK4) is up-regulated in skeletal muscle in insulin-resistant states and is postulated to be involved in the etiology of insulin resistance (3-4). In adipocytes, increases in PDK4 are associated with an inhibition of glucose uptake (4), yet despite this, this enzyme is thought to mediate a portion of the beneficial effects of TZDs on adipose tissue lipid metabolism. PDK4 inactivates pyruvate dehydrogenase complexes resulting in a shunting of pyruvate to oxaloacetate and ultimately the generation of glycerol-3-phosphate (G3P) (1). TZDs increase the expression of PDK4, concomitant with improvements in insulin sensitivity in Zucker fa/fa rats (1). The inhibition of this enzyme attenuates TZD mediated increases in glyceroneogenesis and leads to increases in fatty acid release (1).

<sup>&</sup>lt;sup>2</sup> A version of this chapter has been published. Wan et al. Obesity (Silver Spring). 2012 Feb;20(2):453-6.

While effective, TZDs possess many deleterious side effects (5) and thus the identification of alternative approaches to mimic the beneficial effects of these compounds is needed. In this regard we have previously demonstrated that epinephrine increases the expression of PDK4 through PPARy and p38 MAPK (p38) dependent pathways (6). While these findings would suggest that the activation of beta adrenergic signaling may serve as an effective means with which to mimic one of the beneficial effects of TZDs, it should be noted that this study was completed using adipose tissue from lean, insulin-sensitive animals. Adipocytes from high fat fed rodents display a blunted response to the effects of catecholamines (7) and thus it is not clear if acute elevations in beta adrenergic signaling would be an effective means with which to induce the expression of PDK4 in adipose tissue from obese, insulin resistant animals. Therefore, the purpose of the present investigation was to determine if acute increases in epinephrine could induce PDK4 in adipose tissue from insulin resistant rats and if the signaling pathways mediating this effect were intact. As exercise increases plasma catecholamine levels, we further wanted to determine if this perturbation would have similar beneficial effects on PDK4 expression in adipose tissue from obese rodents.

# 4.2 Methods and Materials

#### **4.2.1 Methods and Procedures**

All protocols followed Canadian Council on Animal Care (CCAC) guidelines and were approved by the Animal Use and Welfare Committee at the University of Alberta. Male Wistar rats (Charles River, Wilmington, MA) were housed 2 per cage, with a 12/12hour light/dark cycle and were provided with water and either standard rat chow a high fat diet *ad libitum* for 6 weeks. The chow diet contained 28.5% protein, 13.5% fat, and 58.0% carbohydrates, expressed as percentage of total energy. The high fat diet contained 18.4% protein, 21.3% carbohydrate and 60.3% fat (Harlan, Madison, WI). Rats were injected intraperitoneally (IP) with a weight-adjusted bolus of epinephrine (20  $\mu$ g/100 gram body weight) or an equivalent volume of sterile saline (8) or swam for 2 hours with the remaining animals serving as controls (6). Immediately following exercise, or 30 minutes or 2 hours after epinephrine injections, animals were anesthesized with sodium pentobarbitol (5mg/100g body weight) and epididymal adipose tissue was harvested and stored at -80 °C until further analysis.

As we have described previously (6) and detailed in chapter 3, the protein content and/or phosphorylation of AMPK (5'AMP-activated protein kinase), p38, MK-2 (MAPKactivated protein kinase 2) and ACC (acetyl CoA carboxylase) were determined by Western blot analysis, while changes in PDK4 mRNA expression were determined using real time RT-PCR with beta actin as a housekeeping gene (6). Relative differences in gene expression between groups were determined using the  $2^{-\Delta\Delta CT}$  method.

#### 4.2.2 Statistical analysis

Data are presented as means  $\pm$  standard error of the mean (SEM). Data were compared with two-way analysis of variance (ANOVA) followed by a post-hoc comparison using Fisher's LSD test. Statistical significance was set at p < 0.05.

#### **4.3 Results and Discussion**

Although TZDs increase the expression of PDK4 in adipose tissue from obese, insulin resistant rodents (1), these compounds possesses many unwanted side effects (5)

and thus it was of great interest to determine if acute increases in epinephrine induce PDK4 mRNA expression in adipose tissue from obese, insulin resistant animals. Rats fed a HFD weighed more (chow 448  $\pm$  6, HFD 519  $\pm$  8 g, *p*<0.05), had heavier epididymal fat pads (chow 5.7  $\pm$  0.3, HFD 15.4  $\pm$  1.2 g, *p*<0.05) and were hyperglycemic (chow 98.6  $\pm$  3.3, HFD 115.9  $\pm$  5.7 mg/dl, *p*<0.05). As seen in *Figure 4-1A below*, we made the novel observation that epinephrine-induced PDK4 mRNA levels to a similar extent in adipose tissue from chow and HFD rats. It should be stressed that these were mRNA measurements and longer duration studies are needed to determine if equivalent changes occur in PDK4 protein.

We have previously shown that p38 plays a role in the pathway(s) through which epinephrine induces PDK4 mRNA in adipose tissue from lean rats (6). The activation of p38 has been reported to be impaired in skeletal muscle from insulin resistant mice (9) and thus it was of interest to determine if p38 signaling was intact in adipose tissue from HFD rats. As the activation of p38 would occur prior to the induction of PDK4 gene expression, we harvested tissue 30 minutes following a bolus injection of epinephrine. To our surprise we found that the phosphorylation of p38, and its downstream substrate, MK2, were markedly blunted in adipose tissue from HFD compared to chow fed rats (*Figure 4-1 B, C*). This is particularly surprising given the robust induction of PDK4 in epididymal adipose tissue from these animals. To determine if the blunted effect of the HFD, we measured changes in AMPK signaling. Consistent with previous reports we found that epinephrine increased the phosphorylation of AMPK and its substrate ACC in chow fed rats (10). In contrast to p38, the phosphorylation of AMPK and ACC were not blunted in adipose tissue from HFD rats (*Figure 4-1 D, E*). These findings are in contrast to a recent report from Ceddia's laboratory (7) demonstrating a marked reduction in AMPK protein content and signaling in isolated adipocytes from HFD fed mice. This discrepancy could be explained by species-related differences, the mode of epinephrine administration or the use of intact fat pads vs. isolated adipocytes.

A. PDK4 mRNA







B. p-p38



D. p- AMPK







**Figure 4-1 Effects of epinephrine injection on PDK4, p38 and AMPK signalling pathways from chow and HFD rats.** A) Epinephrine (epi) increases PDK4 mRNA levels in adipose tissue from HFD rats. p38 MAPK signaling is attenuated in HFD rats (B and C) while AMPK signaling is intact (D and E). Data are presented as means + SEM for 8-10 animals per group. Real time RT-PCR data is presented as a fold

change compared to saline injected chow fed rats. Representative Western blots are displayed in panel F. \* p<0.05 compared to saline treated condition within the same diet group. (For figure 4-1A, the relative PDK4 gene expression in chow saline injection group is  $1.07\pm0.14$ .)

The induction of PDK4, despite the blunted activation of p38 in adipose tissue from HFD rats suggests that alternative mechanisms mediate the effects of epinephrine on the induction of PDK4. In skeletal muscle AMPK has been implicated in the control of this enzyme (11). However, in cultured adipose tissue we have shown that AMPK agonists such as metformin and AICAR decrease the expression of PDK4 (6). As absolute rates of fatty-acid re-esterification parallel lipolysis (12), and given the effect of AMPK on inhibiting lipolysis (13) this finding is not entirely unexpected. Collectively these findings suggest that AMPK- and p38- independent pathways are involved in the epinephrine-mediated induction of PDK4 *in vivo* in adipose tissue from HFD rats. TZDs have been shown to induce the expression of PPAR $\gamma$  dependent genes, in part, through the inhibition of cyclin dependent kinase 5-mediated phosphorylation of this transcription factor (14). While speculative, it is possible epinephrine may mediate the induction of PDK4 in adipose tissue from HFD rats through a similar mechanism.

Regardless of the specific pathways(s) involved our data clearly highlights the beneficial effect of acute increases in epinephrine on the induction of PDK4 in adipose tissue from HFD rats. The procedure utilized in our experiments raises plasma epinephrine levels to ~190 nmol/l, a concentration much higher than what is seen under normal physiological circumstances (8). Given this we wanted to determine if exercise, a condition that robustly increases plasma epinephrine levels, but to levels lower than what is seen with epinephrine injections (~15 nmol/l (15)), would be a beneficial approach to induce PDK4 in adipose tissue. Two hours of swim exercise increased PDK4 mRNA

levels to an equivalent or even greater extent in adipose tissue from HFD rats (chow exercise  $5.49 \pm 0.73$ , HFD sedentary  $1.36 \pm 0.17$ , HFD exercise  $9.36 \pm 1.05$  fold increase versus chow sedentary).

In summary our data is the very first to demonstrate that epinephrine and exercise, similar to TZDs, are potent stimuli that can be used to induce PDK4 in adipose tissue from obese, insulin resistant rodents. We have made the further novel observation that the pathways regulating the effects of epinephrine on PDK4 in adipose tissue from HFD rats may be distinct from those in lean animals. Our findings suggest that interventions that acutely increase epinephrine levels, such as exercise, may serve as an effective TZD independent approach with which to modulate adipose tissue metabolism.

# 4.4 Literature Cited

1. Cadoudal T, Distel E, Durant S, Fouque F, Blouin JM, Collinet M, et al. Pyruvate dehydrogenase kinase 4: regulation by thiazolidinediones and implication in glyceroneogenesis in adipose tissue. Diabetes. 2008 Sep;57(9):2272-9.

2. Forest C, Tordjman J, Glorian M, Duplus E, Chauvet G, Quette J, et al. Fatty acid recycling in adipocytes: a role for glyceroneogenesis and phosphoenolpyruvate carboxykinase. Biochem Soc Trans. 2003 Dec;31(Pt 6):1125-9.

3. Holness MJ, Bulmer K, Smith ND, Sugden MC. Investigation of potential mechanisms regulating protein expression of hepatic pyruvate dehydrogenase kinase isoforms 2 and 4 by fatty acids and thyroid hormone. Biochem J. 2003 Feb 1;369(Pt 3):687-95.

4. White UA, Coulter AA, Miles TK, Stephens JM. The STAT5A-mediated induction of pyruvate dehydrogenase kinase 4 expression by prolactin or growth hormone in adipocytes. Diabetes. 2007 Jun;56(6):1623-9.

5. Stafylas PC, Sarafidis PA, Lasaridis AN. The controversial effects of thiazolidinediones on cardiovascular morbidity and mortality. Int J Cardiol. 2009 Jan 24;131(3):298-304.

6. Wan Z, Thrush AB, Legare M, Frier BC, Sutherland LN, Williams DB, et al. Epinephrine-mediated regulation of PDK4 mRNA in rat adipose tissue. Am J Physiol Cell Physiol. 2010 Nov;299(5):C1162-70.

7. Gaidhu MP, Anthony NM, Patel P, Hawke TJ, Ceddia RB. Dysregulation of lipolysis and lipid metabolism in visceral and subcutaneous adipocytes by high-fat diet: role of ATGL, HSL, and AMPK. Am J Physiol Cell Physiol. 2010 Apr;298(4):C961-71.

8. Nolte LA, Gulve EA, Holloszy JO. Epinephrine-induced in vivo muscle glycogen depletion enhances insulin sensitivity of glucose transport. J Appl Physiol. 1994 May;76(5):2054-8.

9. Roth RJ, Le AM, Zhang L, Kahn M, Samuel VT, Shulman GI, et al. MAPK phosphatase-1 facilitates the loss of oxidative myofibers associated with obesity in mice. J Clin Invest. 2009 Dec;119(12):3817-29.

10. Koh HJ, Hirshman MF, He H, Li Y, Manabe Y, Balschi JA, et al. Adrenaline is a critical mediator of acute exercise-induced AMP-activated protein kinase activation in adipocytes. Biochem J. 2007 May 1;403(3):473-81.

11. Houten SM, Chegary M, Te Brinke H, Wijnen WJ, Glatz JF, Luiken JJ, et al. Pyruvate dehydrogenase kinase 4 expression is synergistically induced by AMP-activated protein kinase and fatty acids. Cell Mol Life Sci. 2009 Apr;66(7):1283-94.

12. Brooks B, Arch JR, Newsholme EA. Effects of hormones on the rate of the triacylglycerol/fatty acid substrate cycle in adipocytes and epididymal fat pads. FEBS Lett. 1982 Sep 20;146(2):327-30.

13. Corton JM, Gillespie JG, Hawley SA, Hardie DG. 5-aminoimidazole-4carboxamide ribonucleoside. A specific method for activating AMP-activated protein kinase in intact cells? Eur J Biochem. 1995 Apr 15;229(2):558-65.

14. Choi JH, Banks AS, Estall JL, Kajimura S, Bostrom P, Laznik D, et al. Antidiabetic drugs inhibit obesity-linked phosphorylation of PPARgamma by Cdk5. Nature. 2010 Jul 22;466(7305):451-6.

15. Higashida K, Higuchi M, Terada S. Potential role of lipin-1 in exercise-induced mitochondrial biogenesis. Biochem Biophys Res Commun. 2008 Sep 26;374(3):587-91.

# Chapter 5 IL-6 Indirectly Modulates the Induction of Glyceroneogenic Enzymes in Adipose Tissue during Exercise<sup>3</sup>

# **5.1 Introduction**

When blood glucose levels are limiting such as during exercise, the breakdown of triglyceride (TG) molecules within fat cells is accelerated. While the majority (~65-75%) of liberated fatty acids following lipolysis are released into the circulation to be used as a fuel source, a significant amount are retained in the fat cell and are re-esterified back to TG (1). The re-esterification of fatty acids requires the provision of glycerol 3-phosphate (G-3-P), and in rodent adipose tissue, the generation of G-3-P occurs primarily through *de novo* synthesis from sources such as lactate and pyruvate, in a process termed glyceroneogenesis (2-3). Phosphoenolpyruvate carboxykinase (PEPCK) and pyruvate dehydrogenase kinase 4 (PDK4) have been identified as essential components of the glyceroneogenic enzymatic machinery (4-5).

Similar to what has been reported in skeletal muscle (6), we have found that exercise leads to a robust induction of PDK4 in white adipose tissue (7). We demonstrated that this effect was recapitulated by epinephrine and could involve p38 mitogen activated protein kinase (MAPK) (7). On the other hand, 5-aminoimidazole-4carboxyamide ribonucleoside (AICAR) and metformin, which are 5' AMP activated protein kinase (AMPK) agonists, decreased PDK4 mRNA expression in rat adipose tissue (7). AMPK is an energy-sensing enzyme that inhibits energy-consuming processes (8).

<sup>&</sup>lt;sup>3</sup> A version of this chapter has been published. Wan et al. PLoS One. 2012;7(7):e41719. Epub 2012 Jul 23.

As fatty acid re-esterification is one of the primary drains on ATP levels in adipocytes (9) it is not entirely surprising that AMPK agonists would decrease the expression of enzymes involved in this process.

Given the vital function of adipose tissue in the provision of fatty acids, it is likely that there are multiple systemic factors involved in the regulation of genes involved in fatty acid handling. In this light recent work has highlighted the potential involvement of skeletal muscle derived interleukin 6 (IL-6) as a mediator of adipose tissue metabolism during exercise (10). For example, IL-6 has been reported to stimulate adipose tissue lipolysis (11) and to activate AMPK (12) in adipocytes. Moreover, the activation of AMPK during exercise is blunted in adipose tissue from IL-6 knockout mice (12). Of interest, previous work has demonstrated that long-term increases in inflammation (13-14), and in particular IL-6 (15), attenuate PEPCK expression in fat cells. These findings suggest that increases in IL-6 may serve to dampen the exercise-mediated induction of glyceroneogenic enzymes in adipose tissue.

The long-held view that muscle derived IL-6 signals to adipose tissue during exercise is increasingly being challenged. For instance, it has been reported that IL-6 infusions do not increase lipolysis or activate IL-6 signalling in adipose tissue from healthy humans (16). Likewise, *in vivo* AMPK activity was recently shown to be similar in subcutaneous adipose tissue from wild type and IL-6 deficient mice (17). Moreover, and somewhat surprisingly, to the best of our knowledge it is not known if IL-6 signalling is activated in adipose tissue during exercise.

With the aforementioned points in mind the purpose of the present investigation was to determine the role of IL-6 in the induction of glyceroneogenic enzymes during exercise in white adipose tissue. Using a combination of *ex vivo* adipose tissue culture approaches and whole body IL-6 deficient mice (IL- $6^{-/-}$ ) we hypothesized that a) IL-6 would directly and rapidly attenuate PEPCK and PDK4 mRNA expression in mouse adipose tissue, b) IL-6 signalling would be activated in adipose tissue during exercise and c) the induction of glyceroneogenic enzymes following exercise would be enhanced in adipose tissue from IL- $6^{-/-}$  mice.

# **5.2 Methods and Materials**

### 5.2.1 Materials

Reagents, molecular weight marker and nitrocellulose membranes for SDS-PAGE were purchased from Bio-Rad (Mississauga, ON). ECL Plus was a product of Amersham Pharmacia Biotech (Arlington Heights, IL). Antibodies against p-cAMP response element binding protein (p-CREB) (CAT# 4276), p-p38 MAPK (CAT #9211), p38 MAPK (CAT#9212), p-MAP kinase-activated protein kinase 2 (p-MK-2) (CAT#3044), MK2 (CAT#3042), p-AMPK (CAT#2531), AMPK $\alpha$  (CAT#2793), pacetyl-CoA carboxylase (p-ACC) (CAT#3661), ACC (CAT#3662), p- signal transducer and activator of transcription 3 (STAT3) tyrosine 705 (CAT#4113), STAT3 (CAT#9132), p-hormone sensitive lipase (p-HSL) (Ser563) (CAT#4139), HSL (CAT#4107), protein phosphatase type 2A (PP2A) (CAT# 2041), liver kinase B1 (LKB-1) (CAT#3050) and PPAR $\gamma$  (CAT#2430) were from Cell Signaling (Danvers, MA). Antibodies against tubulin (CAT#ab7291), the beta 3 adrenergic receptor ( $\beta$ 3-AR) (CAT#ab94506), PDK4 (CAT#ab38242), and PP2C (CAT# ab27267) were purchased from Abcam (Cambridge, MA) Anti PEPCK antibody (CAT# 10004943) was obtained from Cayman Chemicals (Ann Arbor, MI). Horseradish peroxidase-conjugated donkey anti-rabbit and goat antimouse IgG secondary antibodies were purchased from Jackson ImmunoResearch Laboratories (West Grove, PA). Fatty acid-free bovine serum albumin (FA-free BSA) (CAT# 152401) was from MP Biomedical (Solon, OH). Free glycerol reagent (CAT#F6428) was from Sigma (Oakville, ON). Non-esterified fatty acids assay kits (NEFA-HR kit) were purchased from Wako Chemicals (Richmond, VA). Recombinant mouse IL-6 (RMIL6I) was obtained from Thermo Scientific (Rockford, IL) while a mouse IL-6 ELISA was from Biolegend (San Diego, CA). AICAR was purchased from Toronto Research Chemicals (Toronto, ON). SuperScript II Reverse Transcriptase, oligo(dT) and dNTP were products from Invitrogen (Burlington, ON). Taqman Gene Expression Assays for mouse  $\beta$  actin (4352933E), eukaryotic 18S rRNA (4352930E), PDK4 (Mm01166879\_m1), PEPCK (Mm01247058\_m1) and suppressor of cytokine signaling3 (SOCS3) (Mm00545913\_s1) were from Applied Biosystems (Foster City, CA). All other chemicals were purchased from Sigma (Oakville, ON).

## **5.2.2 Treatment of Animals**

All protocols followed Canadian Council on Animal Care (CCAC) guidelines and were approved by the University of Guelph Animal Care Committee. 3 month old male IL-6<sup>-/-</sup> mice (Jackson Laboratories B6.12952-IL6<sup>tmlkopf/</sup>J) on a C57BL/6J background and age-matched C57BL/6J wild-type (WT) mice were housed 3 per cage, with a 12/12-hour light/dark cycle, and were provided with water and standard rodent chow *ad libitum*. Mice were acclimated to the animal housing facility for 1 month prior to the start of the exercise experiment. Before the experiment, all mice were acclimated to running on a motor-driven treadmill during a 3-day period with 15 min of running at 15 m/min, 5% grade per day. On the day of the experiment, one-half of the mice in each group (IL-6<sup>-/-</sup> and WT) were sedentary (KOSED and WTSED), and the other half subjected to running on a motorized treadmill at 15 m/min, with an incline of 5% for 90 min (KOEX and WTEX). This intensity of exercise is well tolerated by IL-6 deficient mice (18). Immediately following exercise cessation, mice were killed by cervical dislocation and epididymal white adipose tissue (eWAT) removed, immediately weighed and then clamp-frozen in tongs cooled to the temperature of liquid nitrogen and stored at -80  $\$  until further analysis. In a separate set of experiments blood was collected from the saphenous vein prior to, and immediately following 90 minutes of treadmill exercise. Blood was spun down and the plasma collected for the determination of glucose (YSI Glucose Analyzer), fatty acids, TGs and IL-6.

# 5.2.3 Adipose Tissue Organ Culture

Adipose tissue organ culture is a well characterized technique that has been used to determine changes in adipose tissue metabolism and gene expression (19). Epididymal adipose tissue was cultured as we have described in detail previously (7, 20) and detailed in chapter 3. To assess the effects of IL-6 or AICAR on the mRNA expression of PEPCK and PDK4, adipose tissue was cultured for 24 h and then treated with mouse IL-6 (150 ng/ml) or AICAR (1 mM) for 2, 6 or 12 h. To determine the effects of IL-6 on the activation of the AMPK and p38MAPK signalling pathways, experiments were conducted as described above except cultures were treated with mouse IL-6 (150 ng/ml) for 30 min. To determine the effects of IL-6 on lipolysis, adipose tissue was cultured as described above. On the morning of the experiment, media was replaced with fresh M199 supplemented with 2.5% FA-free BSA. IL-6 (150 ng/ml) or epinephrine (1 µM) was

added and media were collected after a 2h treatment. To determine the effects of IL-6 on fatty acid re-esterification, adipose tissue was cultured as described above and treated with mouse IL-6 (150 ng/ml) for 24h. Thereafter, the media was replaced by fresh M199 supplemented with 2.5% FA-free BSA and pyruvate (4 mM), media collected 4h later. At the end of all the experiments adipose tissue cultures were rinsed in ice-cold sterile PBS, strained and adipose tissue fragments snap frozen and stored at -80 ° C for further analysis.

### **5.2.4 Lipolysis in Adipose Tissue Explants**

To determine the effects of *ex vivo* treatment of epinephrine on lipolysis, epididymal adipose tissue minces (~50 mg cut into 10-20 mg fragments) from 4 month old male WT and IL-6<sup>-/-</sup> mice were incubated in 1.0 ml of oxygenated Krebs-Ringer Buffer (KRB) (118 mM NaCl, 4.8 mM KCl, 1.25 mM CaCl<sub>2</sub>, 1.2 mM KH<sub>2</sub>PO<sub>4</sub>, 1.2 mM MgSO<sub>4</sub>, 25 mM NaHCO<sub>3</sub>, 5 mM glucose, pH 7.4) with 2.5% FA-free BSA supplemented with or without 1  $\mu$ M epinephrine at 37° C in a shaking water bath. Buffer samples were collected after 2 hours. NEFA and glycerol were analyzed by colormetric assays according to the manufacturer's instructions. NEFA and glycerol concentrations were corrected for tissue weight and reported as µmol released per g tissue.

# **5.2.5 Western Blot Analysis**

Protein was extracted from adipose tissue and changes in the phosphorylation status or content of AMPK, p38, MK2, STAT3, CREB, HSL,  $\beta$ 3-AR, LKB1, PP2A, PP2C and PPAR $\gamma$  were determined by Western blotting, as described in detail by our laboratory previously (7, 20). Briefly, adipose tissue samples were homogenized in 2
volumes of ice-cold cell lysis buffer supplemented with Protease Inhibitor Cocktail (Sigma) and phenylmethylsulfonyl fluoride. Homogenized samples were centrifuged for 10 min at 1500 X G at 4 °C. The fat cake was removed and the infranatant was collected and protein concentration determined using the BCA method. Equal amounts of protein were separated on 7.5% or 10% SDS-PAGE gels. Proteins were wet transferred to nitrocellulose membranes at 200mA/tank and subsequently blocked in Tris buffered saline/0.1% Tween 20 (TBST) supplemented with 5% non-fat dry milk for 1 h at room temperature with gentle agitation. Membranes were incubated in appropriate primary antibodies diluted in TBST/5% non-fat dry milk overnight at 4° C with gentle agitation. The following morning, membranes were briefly washed in TBST and then incubated in HRP-conjugated secondary antibodies diluted in TBST/1% non-fat dry milk for 1 h at room temperature. Signals were detected using enhanced chemiluminesence and were subsequently quantified by densitometry by Gene Tool according to the manufacturer's instructions (SynGene, ChemiGenius2, PerkinElmer).

#### 5.2.6 Real Time PCR

RNA was isolated from adipose tissue using an RNeasy kit according to the manufacturer's directions. Total RNA (1  $\mu$ g) was used for the synthesis of complementary DNA (cDNA) using SuperScript II Reverse Transcriptase, oligo(dT) and dNTP. Real time PCR was performed using a 7500 Fast Real-Time PCR system (Applied Biosystems). Samples were run in duplicate in a 96-well plate format. Each assay (20  $\mu$ l total volume) contained 1  $\mu$ l gene expression assay, 1  $\mu$ l cDNA template, 10  $\mu$ l Taqman Fast Universal PCR Master Mix and 8  $\mu$ l RNase-free water. For  $\beta$ -actin or 18S, each 50  $\mu$ l reaction contained 25  $\mu$ l PCR Master mix, 2.5  $\mu$ l each of gene expression assay, 1  $\mu$ l

cDNA template, and 21.5  $\mu$ l RNase-free water. Results were normalized to the mRNA expression of  $\beta$ -actin or 18S. Relative differences in gene expression between groups were determined using the 2<sup>- $\Delta\Delta$ CT</sup> method (21). The amplification efficiencies of the gene of interest and the housekeeping gene were equivalent.

#### **5.2.7 Statistical Analysis**

Data are presented as means  $\pm$  standard error of the mean (SEM). Comparisons between vehicle and IL-6 signalling, and differences between the exercise-induced increases in PEPCK and PDK4 mRNA expression in WT and IL-6<sup>-/-</sup> were made using a Student's T-Test. Comparisons between vehicle and IL-6 or AICAR treated cultures on gene expression, and the effects of IL-6 and epinephrine on lipolysis in cultured adipose tissue were made using a one-way ANOVA followed by a post hoc comparison using Fisher's LSD test. Comparisons between WT and IL-6 KO mice in regards to signalling, *ex vivo* lipolysis and plasma glucose and fatty acid levels were completed using a 2 X 2 ANOVA with LSD post hoc analysis. Statistical significance was set at *p*<0.05.

#### **5.3 Results**

#### 5.3.1 Effects of ex vivo IL-6 Treatment on Signalling and Gene expression in eWAT

IL-6 treatment (150 ng/ml, 30 min) of cultured eWAT led to an ~ 3-fold increase in the tyrosine 705 phosphorylation of STAT3. Similarly, the phosphorylation of AMPK and its downstream substrate ACC were also increased by IL-6. The phosphorylation of p38MAPK and its substrate MK-2, was not increased by IL-6 (*Figure 5-1*). IL-6 led to a rapid induction of SOCS3 (2 hours  $4.76 \pm 1.26$  fold increase, 6 hours  $5.48 \pm 1.49$  fold increase, 12 hours  $4.24 \pm 1.33$  fold increase, p<0.05 vs control) a transcriptional target of IL-6 while decreasing PEPCK and PDK4 mRNA expression (*Figure 5-2A*). Reductions in PEPCK and PDK4 mRNA expression were mirrored by decreases in the protein content of these enzymes (*Figure 5-2B*). To determine if decreases in PEPCK and PDK4 resulted in a functional impairment in fatty acid handling we treated cultured adipose tissue with IL-6 (24 hours, 150 ng/ml) and then measured the ratio of fatty acid to glycerol released into the media 4 hours following the removal of IL-6. As seen in *figure 5-2C*, IL-6 treatment resulted in increases in the FFA/glycerol which is indicative of reductions in fatty acid re-esterification. As seen in *Table 5-1*, AICAR (1 mM) treatment reduced the mRNA expression of PEPCK and PDK4.



Figure 5-1 IL-6 signalling in cultured adipose tissue. *Ex vivo* IL-6 treatment (150 ng/ml, 30mins) increases the phosphorylation of STAT3, AMPK and ACC but not the p38MAPK signalling pathways in cultured eWAT. Data are presented as means + SEM for 7 cultures per group. Representative Western blot images are given to the right of the quantified data. \* p<0.05 versus vehicle control.



**Figure 5-2 Effects of** *ex vivo* **IL-6 on PEPCK and PDK4**. *Ex vivo* **IL-6** treatment (150 ng/ml) decreases A) PEPCK and PDK4 mRNA expression and B) protein content in cultured eWAT (150 ng/ml, 12h). These changes are associated with increases C) in fatty acids to glycerol ratio following long-term (150 ng/ml, 24 h) IL-6 treatment. Data are presented as means + SEM for 7 cultures per group. The mRNA data are normalized to beta actin mRNA and expressed as fold differences compared to vehicle controls in A. Representative Western blot images are given to the right of the quantified data in B. \* p<0.05 versus vehicle control. (For figure 5-2A, the relative PEPCK and PDK4 mRNA expression in vehicle treated groups is 1.45 ± 0.34 and 1.39 ± 0.52, respectively.)

Table 5-1 Effects of *ex vivo* AICAR Treatment on PEPCK, PDK4 mRNA Expression in eWAT.

Group	РЕРСК	PDK4
AICAR 6h	$0.49 \pm 0.13^{a}$	$0.90 \pm 0.21$
AICAR 12h	$0.44 \pm 0.12$ <sup>a</sup>	$0.64 \pm 0.09^{a}$

Data are presented as means  $\pm$  SEM for 6 cultures per group. The mRNA data are normalized to 18S mRNA and expressed as relative mRNA differences compared to vehicle controls (vehicle =1.0). <sup>a</sup> p<0.05 versus vehicle control.

#### 5.3.2 The IL-6 Mediated Activation of AMPK Occurs Independent of Lipolysis

As presented in *Figure 5-3*, short-term IL-6 treatment (150 ng/ml, 2 h) did not stimulate lipolysis as determined by the release of glycerol (*Figure 5-3A*) and fatty acids (*Figure 5-3B*) into the culture media. Epinephrine (1  $\mu$ M) was included in these experiments as a positive control, and resulted in marked increases in glycerol and fatty acid release. In contrast to short term incubations, IL-6 treatment for 12 hours moderately enhanced glycerol release (control 5.1 ± 0.7, IL-6 6.8 ± 0.8  $\mu$ mol/gram tissue/12 hours, n=6 per group, *p*=0.053).



Figure 5-3 Effects of *ex vivo* IL-6 on adipose tissue lipolysis. Short term (2 hours) IL-6 (150 ng/ml) treatment does not increase A) glycerol or B) fatty acid release from cultured eWAT. Epinephrine (2 hours, 1  $\mu$ M) treated cultures were included as a positive control. Data are presented as means + SEM for 7 cultures per group. \* *p*<0.05 compared to vehicle and IL-6 groups.

#### 5.3.3 Markers of IL-6 Signalling Are Not Activated in Adipose Tissue during

#### Exercise

SOCS3 mRNA expression in eWAT did not increase in response to the exercise protocol we utilized (*Figure 5-4A*). Meanwhile, a significant decrease in the phosphorylation status of STAT3 (Tyr705) was observed immediately after exercise (*Figure 5-4B*). Plasma IL-6 levels tended to be higher following exercise (13.4  $\pm$  6.1 sedentary, 25.3  $\pm$  5.4 exercise pg/ml, *p*=0.085).



Figure 5-4 Markers of IL-6 signalling in adipose tissue following exercise. An acute bout of exercise does not activate IL-6 signalling in eWAT as demonstrated by A) unchanged SOCS3 mRNA levels and B) decreases in the phosphorylation of STAT3 (Tyr705). Data are presented as means + SEM for 10 animals per group. The mRNA data is normalized to 18S and expressed as fold differences compared to the WT sedentary (SED) group in A. Representative Western blot images are given to the right of the quantified data in B. \* p<0.05 versus WT SED.

## 5.3.4 Exercise-Mediated Increases in PEPCK and PDK4 Expression Are Attenuated in IL-6<sup>-/-</sup> Mice

The expression of PEPCK and PDK4 in eWAT from WT and IL-6<sup>-/-</sup> sedentary mice were similar. Consequently, we compared the exercise-mediated induction of PEPCK and PDK4 between WT and IL-6<sup>-/-</sup> mice. As seen in *Figure 5.5*, the induction of glyceroneogenic enzymes, especially PDK4, was blunted in adipose tissue from IL-6 deficient mice. There were no differences in PPAR $\gamma$  protein content in adipose tissue from WT and KO mice (WT 1.00 ± 0.11, IL-6<sup>-/-</sup> 1.03 ± 0.08, arbitrary units N= 8-10/group, *p*>0.05).



Figure 5-5 Exercise-mediated induction of PEPCK and PDK4 from WT and IL-6<sup>-/-</sup> mice. The exercise-mediated induction of PEPCK and PDK4 is attenuated in eWAT from IL-6<sup>-/-</sup> deficient mice. Data are presented as means + SEM for 10-12 animals per group and are expressed as fold increases in gene expression compared to the sedentary control group of the same genotype. <sup>#</sup>p <0.05 compared to WT EX.

### 5.3.5 AMPK Phosphorylation Is Altered in eWAT from IL-6<sup>-/-</sup> Mice

In sedentary IL-6<sup>-/-</sup> mice, phosphorylated AMPK (p-AMPK) in eWAT was diminished by ~40% compared to WT sedentary mice. A single bout of treadmill running did not lead to an increase in p-AMPK from WT mice while exercise significantly increased p-AMPK in IL-6<sup>-/-</sup> mice (*Figure 5-6A*). The phosphorylation of p38 MAPK was not changed following exercise in either genotype (*Figure 5-6B*). Total AMPK and p38 MAPK protein were similar in both genotypes (data not shown). Alterations in AMPK phosphorylation were not associated with changes in the protein content of LKB-1, PP2A or PP2C (*Figure 5-6C*).







**Figure 5-6 Exercise mediated AMPK signaling from WT and IL-6**<sup>-/-</sup> **mice.** A) P-AMPK is reduced in eWAT from sedentary IL-6<sup>-/-</sup> compared to WT mice and increased following an acute bout of exercise in IL-6<sup>-/-</sup> mice only. B) Exercise did not change the phosphorylation of p38 in either genotype. C) There is no differences for the protein expression of LKB1, PP2A and PP2C between WT and KO SED group. Data are presented as means + SEM for 10-12 animals per group. Representative Western blot images are shown to the right of the quantified Western blot data. \* p<0.05 versus WT SED. \* versus IL-6<sup>-/-</sup> KOSED in A.

#### 5.3.6 There Are No Apparent Reductions in Beta Adrenergic Signalling in eWAT

#### from IL-6<sup>-/-</sup> Mice

As beta-adrenergic agonists have been shown to induce glyceroneogenic enzymes we wanted to determine if the attenuation of PDK4 and PEPCK during exercise was secondary to reductions in adrenergic signalling. In this regard we found that the phosphorylation of CREB (*Figure 5-7A*) and HSL (*Figure 5-7B*), proteins that are phosphorylated via beta adrenergic dependent pathways (22-24) were similar in adipose tissue from WT and IL-6 deficient mice following exercise. Likewise, the ability of epinephrine to stimulate lipolysis *ex vivo* was similar in adipose tissue from WT and IL-6 <sup>/-</sup> mice (*Figure 5-7C, D*). There were no differences in the protein content of the beta 3 adrenergic receptor in eWAT from WT and IL-6<sup>-/-</sup> mice (WT 1.00  $\pm$  0.27, IL-6<sup>-/-</sup> 1.03  $\pm$ 0.36 arbitrary units N= 8-10/group, p>0.05). Plasma fatty acid levels were similar at rest between genotypes and increased to a greater extent in IL-6<sup>-/-</sup> mice. Exercise decreased plasma TG levels in both WT and KO mice. There was no effect of genotype or exercise on plasma glucose or glycerol levels (*Table 5-2*).



**Figure 5-7 Beta adrenergic signalling in adipose tissue from WT and IL-6<sup>-/-</sup> mice.** Exercise-induced increases in A) CREB and B) HSL phosphorylation are not different in eWAT from WT and IL-6<sup>-/-</sup> mice. Epinephrine stimulated C) glycerol and D) fatty acid release are not different in eWAT from either WT or IL-6<sup>-/-</sup> mice. Data are presented as means + SEM for 6-12 samples/animals per group. \* p<0.05 compared to sedentary control group in A and B or vehicle treated control group in C and D.

Table 5-2 Effects of Exercise on Plasma Glucose, NEFA, Glycerol and Triglyceride levels from WT and IL-6<sup>-/-</sup> Mice.

Group	Glucose	NEFA	Glycerol	TG
	(mmol/L)	(mmol/L)		(mmol/L)
WT-SED	9.50±0.82	0.51±0.03	0.27±0.02	0.73±0.11
WT-EX	12.59±0.74	$0.74\pm\!\!0.05^{a}$	0.32±0.01	$0.44 \pm 0.06^{a}$
KO-SED	9.86±1.49	0.49±0.05	0.30±0.02	0.73±0.07
KO-EX	13.86±0.46	0.92±0.06 <sup>a,b</sup>	0.28±0.01	$0.45 \pm 0.02^{a}$

Values are means  $\pm$ SEM for 4–7 animals per group. NEFA, nonesterified fatty acid; Glucose and NEFA in sedentary group were measured in the fed state <sup>a</sup> P <0.05 vs. sedentary group within genotype. <sup>b</sup> *p*<0.05 compared to WT-EX group.

#### **5.4 Discussion**

In previous work we have shown that exercise increases the expression of PDK4 in rat adipose tissue (7). While this effect is likely mediated, at least in part, by catecholamines, it is not clear if additional systemic factors modulate the expression of enzymes involved in fatty acid re-esterification. As it has been proposed that exercise increases IL-6 secretion from skeletal muscle and stimulates adipose tissue lipolysis (11), it was of interest to determine if this myokine modulated the expression of PDK4 and PEPCK in mouse white adipose tissue during exercise. As a first step in addressing this question we treated cultured epididymal mouse adipose tissue with IL-6 and found a rapid activation of IL-6 signalling and reductions in the expression of glyceroneogenic enzymes.

The activation of AMPK in fat cells has been suggested to occur as a consequence of lipolysis and subsequent increases in fatty acid re-esterification (25). To determine if this was the case in our model we measured the ability of IL-6 to stimulate fatty acid and glycerol release in cultured eWAT. In contrast to a previous report demonstrating small increases (~5-10%) in glycerol release in 3T3 adipocytes with a similar IL-6 treatment (11), we were unable to detect an effect of IL-6. We included epinephrine-treated cultures as a positive control and clearly showed a robust increase in glycerol and fatty acid release with this hormone, demonstrating the suitability of our preparation for the measurement of lipolysis. The discrepancy between previous work done in cultured adipocytes (26-27), and the current results are likely due to the duration of IL-6 exposure. In this regard we found that longer incubations (i.e. 12 hours) increased lipolysis in cultured adipose tissue. As increases in lipolysis occurred at a much later time point than the activation of AMPK this suggests that the modulation of AMPK signalling by IL-6 occurred independent of lipolysis, our findings would question the role of IL-6 as a stimulator of lipolysis in adipose tissue during exercise.

Having shown that IL-6 directly and rapidly reduces the expression of enzymes involved in fatty acid re-esterification we next sought to determine if the deletion of IL-6 would potentiate the effects of exercise on the induction of PDK4 and PEPCK in mouse adipose tissue. As an initial approach we analyzed the effects of exercise on the activation of reputed markers of IL-6 signalling in adipose tissue, i.e. STAT3 phosphorylation (28) and the induction of SOCS3 mRNA (29). While we did not detect changes in these parameters, perhaps these findings are not entirely unexpected. In this regard it has been reported that interstitial concentrations of IL-6 (30) are several orders of magnitude higher than levels in the circulation, even after extremely large volumes of exercise such as a marathon (31). Thus, it seems unlikely that an increase in plasma IL-6 in the low

pg/ml range would be a sufficient enough stimulus to activate IL-6 signalling in a tissue that is bathed by much higher concentrations at rest. With these points in mind we can not rule out the possibility that longer durations, and/or a greater intensity of exercise could lead to increases in IL-6 signalling in adipose tissue.

Despite the fact that IL-6 signalling did not appear to be activated in adipose tissue during exercise, we found that the exercised-mediated induction of PEPCK and PDK4 was attenuated in adipose tissue from  $IL-6^{-/-}$  mice. We interpret these findings as suggesting that the attenuated effect of exercise on these genes was indirect. In an effort to elucidate the mechanisms that could, at least in part, explain the attenuation in glyceroneogenic enzymes we assessed changes in AMPK signalling. Consistent with previous work from Kelly et al. (12) we found that the phosphorylation of AMPK was decreased in adipose tissue from IL-6 deficient mice at rest. Although exercise did not appreciably increase AMPK phosphorylation in adipose tissue from WT mice, a finding similar to recent work from Pilegaard's group (17), exercise lead to an  $\sim 2$  fold increase in AMPK phosphorylation in adipose from IL-6 deficient mice. Alterations in AMPK phsophorylation could not be explained by differences in the total protein content of LKB-1, an upstream AMPK kinase (32), or PP2C/PP2A, protein phosphatises which dephosphorylate AMPK (33-34). As we have shown that AICAR treatment reduces PEPCK and PDK4 mRNA expression, these results provide evidence suggesting that the greater relative increase in AMPK activation in adipose tissue from IL-6<sup>-/-</sup> mice may be associated with the attenuated induction of PEPCK and PDK4. At this point it is not clear why exercise increased AMPK phosphorylation to a greater extent in eWAT from IL-6 -/versus WT mice.

In addition to AMPK, we have previously demonstrated a role for p38 MAPK and PPAR $\gamma$  in the regulation of PDK4 mRNA in rat adipose tissue (7). Thus, we thought it plausible that genotypic differences in these parameters could explain, at least in part, the blunted induction of PDK4 in adipose tissue from IL-6 deficient following exercise. As the protein content of PPAR $\gamma$  was not different between groups and p38 MAPK phosphorylation not increased in either genotype post-exercise, it seems unlikely that differences in these pathways account for the attenuated induction of PDK4.

We (7, 35), and others (36), have demonstrated a role for beta-adrenergic agonists in the control of glyceroneogenic enzymes in adipose tissue. Therefore, it is plausible that alterations in beta-adrenergic signalling could account for the observed changes in the exercise-mediated induction of PEPCK and PDK4 in adipose tissue from IL-6<sup>-/-</sup> mice. We found that the exercise-induced phosphorylation of HSL and CREB, proteins that are phosphorylated via beta-adrenergic dependent pathways (22-24), were similar between genotypes. Likewise, the ability of epinephrine to stimulate lipolysis in adipose tissue *ex vivo* was nearly identical in adipose tissue from WT and IL-6 deficient mice. Lastly, the exercise-mediated increase in plasma fatty acid levels was, if anything, slightly elevated in IL-6 deficient mice. Collectively these results provide evidence that reductions in the exercise-induced expression of glyceroneogenic enzymes in adipose from IL-6<sup>-/-</sup> mice occurred independent of decreases in beta adrenergic signalling.

If IL-6 modulates the expression of glyceroneogenic enzymes indirectly and apparently independent of alterations in beta adrenergic signalling, the question arises as to the specific mechanisms which could be mediating this effect. In this regard, a growing body of literature has demonstrated the existence of complex tissue-to-tissue communication during exercise. For instance it has recently been shown that exercise stimulates the secretion of chemokine (C-X-C motif) ligand 1 (CXCL1) from the liver through an IL-6 dependent mechanism (37). Increases in circulating CXCL1 levels via the over-expression of this cytokine in skeletal muscle leads to an up-regulation of enzymes involved in fatty acid oxidation (38). Thus, while speculative, it could be argued that increases in skeletal muscle-derived IL-6 signal to the liver during exercise resulting in increases in CXCL1, which in turn acts as a signal involved in the regulation of PEPCK and PDK4 in adipose tissue. Clearly, this is an area of research that requires further attention.

In summary we have found that IL-6 signalling is not activated in adipose tissue during exercise, nor does IL-6, at least in the short-term, increase adipose tissue lipolysis. Despite these results we have evidence to suggest that IL-6 plays a role, albeit most likely indirect, in mediating the effects of exercise on the induction of glyceroneogenic enzymes. These intriguing results shed insight into the complex regulatory pathways governing the expression of genes involved in fatty acid re-esterification and challenge the long held paradigm that IL-6 is a direct mediator of exercise-induced changes in adipose tissue metabolism and gene expression.

#### **5.5 Literature Cited**

1. Wang T, Zang Y, Ling W, Corkey BE, Guo W. Metabolic partitioning of endogenous fatty acid in adipocytes. Obes Res. 2003;11(7):880-7.

2. Reshef L, Hanson RW, Ballard FJ. A possible physiological role for glyceroneogenesis in rat adipose tissue. J Biol Chem. 1970;245(22):5979-84.

3. Reshef L, Olswang Y, Cassuto H, Blum B, Croniger CM, Kalhan SC, et al. Glyceroneogenesis and the triglyceride/fatty acid cycle. J Biol Chem. 2003;278(33):30413-6.

4. Cadoudal T, Blouin JM, Collinet M, Fouque F, Tan GD, Loizon E, et al. Acute and selective regulation of glyceroneogenesis and cytosolic phosphoenolpyruvate carboxykinase in adipose tissue by thiazolidinediones in type 2 diabetes. Diabetologia. 2007;50(3):666-75.

5. Cadoudal T, Distel E, Durant S, Fouque F, Blouin JM, Collinet M, et al. Pyruvate dehydrogenase kinase 4: regulation by thiazolidinediones and implication in glyceroneogenesis in adipose tissue. Diabetes. 2008;57(9):2272-9.

6. Pilegaard H, Ordway GA, Saltin B, Neufer PD. Transcriptional regulation of gene expression in human skeletal muscle during recovery from exercise. Am J Physiol Endocrinol Metab. 2000;279(4):E806-14.

7. Wan Z, Thrush AB, Legare M, Frier BC, Sutherland LN, Williams DB, et al. Epinephrine-mediated regulation of PDK4 mRNA in rat adipose tissue. Am J Physiol Cell Physiol. 2010;299(5):C1162-70.

8. Richter EA, Ruderman NB. AMPK and the biochemistry of exercise: implications for human health and disease. Biochem J. 2009;418(2):261-75.

9. Rognstad R, Katz J. The balance of pyridine nucleotides and ATP in adipose tissue. Proc Natl Acad Sci U S A. 1966;55(5):1148-56.

10. Pedersen BK, Febbraio MA. Muscle as an endocrine organ: focus on musclederived interleukin-6. Physiol Rev. 2008;88(4):1379-406.

11. Petersen EW, Carey AL, Sacchetti M, Steinberg GR, Macaulay SL, Febbraio MA, et al. Acute IL-6 treatment increases fatty acid turnover in elderly humans in vivo and in tissue culture in vitro. Am J Physiol Endocrinol Metab. 2005;288(1):E155-62.

12. Kelly M, Keller C, Avilucea PR, Keller P, Luo Z, Xiang X, et al. AMPK activity is diminished in tissues of IL-6 knockout mice: the effect of exercise. Biochem Biophys Res Commun. 2004;320(2):449-54.

13. Zhang J, Henagan TM, Gao Z, Ye J. Inhibition of glyceroneogenesis by histone deacetylase 3 contributes to lipodystrophy in mice with adipose tissue inflammation. Endocrinology. 2011;152(5):1829-38.

14. Leroyer S, Vatier C, Kadiri S, Quette J, Chapron C, Capeau J, et al. Glyceroneogenesis is inhibited through HIV protease inhibitor-induced inflammation in human subcutaneous but not visceral adipose tissue. J Lipid Res. 2011;52(2):207-20.

15. Feingold KR, Moser A, Shigenaga JK, Grunfeld C. Inflammation inhibits the expression of phosphoenolpyruvate carboxykinase in liver and adipose tissue. Innate Immun. 2011.

16. Wolsk E, Mygind H, Grondahl TS, Pedersen BK, van Hall G. IL-6 selectively stimulates fat metabolism in human skeletal muscle. Am J Physiol Endocrinol Metab. 2010;299(5):E832-40.

17. Brandt C, Jakobsen AH, Adser H, Olesen J, Iversen N, Kristensen JM, et al. IL-6 regulates exercise and training-induced adaptations in subcutaneous adipose tissue in mice. Acta Physiol (Oxf). 2012;205(2):224-35.

18. Fritsche L, Hoene M, Lehmann R, Ellingsgaard H, Hennige AM, Pohl AK, et al. IL-6 deficiency in mice neither impairs induction of metabolic genes in the liver nor affects blood glucose levels during fasting and moderately intense exercise. Diabetologia. 2010;53(8):1732-42.

Fried SK, Moustaid-Moussa N. Culture of adipose tissue and isolated adipocytes.
Methods Mol Biol. 2001;155:197-212.

20. Sutherland LN, Bomhof MR, Capozzi LC, Basaraba SA, Wright DC. Exercise and adrenaline increase PGC-1{alpha} mRNA expression in rat adipose tissue. J Physiol. 2009;587(Pt 7):1607-17.

21. Livak KJ, Schmittgen TD. Analysis of relative gene expression data using realtime quantitative PCR and the 2(-Delta Delta C(T)) Method. Methods. 2001;25(4):402-8.

22. Greenberg AS, Shen WJ, Muliro K, Patel S, Souza SC, Roth RA, et al. Stimulation of lipolysis and hormone-sensitive lipase via the extracellular signal-regulated kinase pathway. J Biol Chem. 2001;276(48):45456-61.

23. Thonberg H, Fredriksson JM, Nedergaard J, Cannon B. A novel pathway for adrenergic stimulation of cAMP-response-element-binding protein (CREB) phosphorylation: mediation via alpha1-adrenoceptors and protein kinase C activation. Biochem J. 2002;364(Pt 1):73-9.

146

24. Tchivileva IE, Tan KS, Gambarian M, Nackley AG, Medvedev AV, Romanov S, et al. Signaling pathways mediating beta3-adrenergic receptor-induced production of interleukin-6 in adipocytes. Mol Immunol. 2009;46(11-12):2256-66.

25. Gauthier MS, Miyoshi H, Souza SC, Cacicedo JM, Saha AK, Greenberg AS, et al. AMP-activated protein kinase is activated as a consequence of lipolysis in the adipocyte: potential mechanism and physiological relevance. J Biol Chem. 2008;283(24):16514-24.

26. Ji C, Chen X, Gao C, Jiao L, Wang J, Xu G, et al. IL-6 induces lipolysis and mitochondrial dysfunction, but does not affect insulin-mediated glucose transport in 3T3-L1 adipocytes. J Bioenerg Biomembr. 2011;43(4):367-75.

27. Yang Y, Yang G. Rosiglitazone regulates IL-6-stimulated lipolysis in porcine adipocytes. Biochem Cell Biol. 2010;88(5):853-60.

28. Heinrich PC, Behrmann I, Haan S, Hermanns HM, Muller-Newen G, Schaper F. Principles of interleukin (IL)-6-type cytokine signalling and its regulation. Biochem J. 2003;374(Pt 1):1-20.

29. Fujimoto M, Naka T. Regulation of cytokine signaling by SOCS family molecules. Trends Immunol. 2003;24(12):659-66.

30. Nielsen NB, Hojbjerre L, Sonne MP, Alibegovic AC, Vaag A, Dela F, et al. Interstitial concentrations of adipokines in subcutaneous abdominal and femoral adipose tissue. Regul Pept. 2009;155(1-3):39-45.

31. Ostrowski K, Hermann C, Bangash A, Schjerling P, Nielsen JN, Pedersen BK. A trauma-like elevation of plasma cytokines in humans in response to treadmill running. J Physiol. 1998;513 (Pt 3):889-94.

32. Thomson DM, Brown JD, Fillmore N, Condon BM, Kim HJ, Barrow JR, et al. LKB1 and the regulation of malonyl-CoA and fatty acid oxidation in muscle. Am J Physiol Endocrinol Metab. 2007;293(6):E1572-9.

33. Wu Y, Song P, Xu J, Zhang M, Zou MH. Activation of protein phosphatase 2A by palmitate inhibits AMP-activated protein kinase. J Biol Chem. 2007;282(13):9777-88.

34. Marley AE, Sullivan JE, Carling D, Abbott WM, Smith GJ, Taylor IW, et al. Biochemical characterization and deletion analysis of recombinant human protein phosphatase 2C alpha. Biochem J. 1996;320 (Pt 3):801-6.

35. Wan Z, Frier BC, Williams DB, Wright DC. Epinephrine induces PDK4 mRNA expression in adipose tissue from obese, insulin resistant rats. Obesity (Silver Spring). 2012;20(2):453-6.

36. Franckhauser-Vogel S, Antras-Ferry J, Robin D, Robin P, Forest C. Transcriptional and posttranscriptional mechanisms of glucocorticoid-mediated repression of phosphoenolpyruvate carboxykinase gene expression in adipocytes. J Cell Biochem. 1997;66(3):386-93.

37. Pedersen L, Pilegaard H, Hansen J, Brandt C, Adser H, Hidalgo J, et al. Exerciseinduced liver chemokine CXCL-1 expression is linked to muscle-derived interleukin-6 expression. J Physiol. 2011;589(Pt 6):1409-20.

38. Pedersen L, Olsen CH, Pedersen BK, Hojman P. Muscle-derived expression of the chemokine CXCL1 attenuates diet-induced obesity and improves fatty acid oxidation in the muscle. Am J Physiol Endocrinol Metab. 2012;302(7):E831-40.

# Chapter 6 IL-6 Deficient Mice Fed a High Fat Diet Display Alterations in Adipose Tissue Metabolism

#### **6.1 Introduction**

Interleukin 6 (IL-6) has traditionally been viewed as a pro-inflammatory cytokine that is linked to the etiology of insulin resistance (1-3). For instance IL-6 in both *in vivo* and *in vitro* models has been shown to cause hepatocyte insulin resistance (4-5). However, accumulating evidence has begun to challenge this long held view. Carey et al. (6) reported that the infusion of IL-6 in humans increased basal and insulin stimulated glucose uptake in skeletal muscle. These results are particularly interesting given the recent work by Matthews et al. (7) demonstrating that IL-6 deficient mice develop insulin resistance and, when fed a high fat diet (HFD), display signs of hepatic steatosis and liver inflammation. Collectively these findings imply that IL-6 possesses insulin-sensitizing properties in liver and muscle.

Owing to the central role of skeletal muscle and liver in the regulation of carbohydrate metabolism, studies investigating the role of IL-6 in glucose homeostasis have focused primarily on these tissues (7-8). However interstitial concentrations of IL-6 surrounding adipose tissue are orders of magnitude greater than levels in the circulation (9) suggesting that adipose tissue-derived IL-6 could play a pivotal role in the maintenance of adipose tissue metabolism and function in an autocrine and/or paracrine manner. Given the growing appreciation for adipose tissue in the control of systemic carbohydrate and lipid metabolism (10) it is somewhat surprising that effects of IL-6 ablation on adipose tissue metabolism have not been examined.

While some studies have reported that IL-6 causes adipose tissue insulin resistance (3), others have provided evidence that IL-6 favorably regulates adipose tissue function. For instance, IL-6 activates 5'AMP activated protein kinase (AMPK) in adipocytes (11). AMPK is a master regulator of cellular metabolism and recent work suggests that AMPK regulates mitochondrial biogenesis in fat cells (12). The induction of mitochondrial biogenesis in adipocytes is mirrored by improvements in insulin action (13), whereas reductions in mitochondrial content/function are associated with adipocyte insulin resistance (14). In addition to mitochondrial content, it has been suggested that AMPK controls the expression of the insulin sensitizing adipokine, adiponectin either directly (15) or secondary to increases in adipose tissue mitochondrial biogenesis (16). Increased expression of adiponectin has been shown to have favorable effects on adipose tissue metabolism such as increasing beta-adrenergic responsiveness and mitochondrial density (17).

Given the increasingly well-recognized role of adipose tissue in regulating glucose homeostasis (10) and the apparent controversy regarding the role of IL-6 in the pathogenesis of insulin resistance, the purpose of the present study was to examine the effects of IL-6 deficiency on adipose tissue metabolism. Given the possibility that a clear manifestation of IL-6 deficiency might not be evident under the condition of a standard chow diet we examined the metabolic outcomes of IL-6 deficiency under both chow and HFD conditions. We hypothesized that IL-6<sup>-/-</sup> mice would display reductions in adipose tissue insulin action, would have blunted beta adrenergic responsiveness and decreases in adipose tissue mitochondrial content and respiration, and adiponectin expression and secretion.

#### **6.2 Methods and Materials**

#### 6.2.1 Materials

Reagents, molecular weight marker and nitrocellulose membranes for SDS-PAGE were purchased from Bio-Rad (Mississauga, ON). Antibodies against ubiquinone: cytochrome c oxidoreductase core I subunit (CORE1) (CAT# MS303) were obtained from Mitosciences (Eugene, OR) while heat shock protein 60 (HSP60) antibodies (CAT# SPA-807) were from Enzo Life Sciences (Ann Arbor, MI). Antibodies against pthreonine 308 protein kinase B (PKB) (CAT#9275), p-serine 473 PKB (CAT#9271), PKB (CAT#4685), p-p38 mitogen-activated protein kinase (MAPK) (CAT #9211), p38 MAPK (CAT#9212), p-stress-activated protein kinase/Jun-amino-terminal kinase (SAPK/JNK) (CAT#4671), p-extracellular regulated kinase (ERK) 1/2 (CAT#9101), ERK1/2 (CAT#4695), p-AMPK (CAT#2531), AMPKa (CAT#2793), p-serine563 hormone sensitive lipase (HSL) (CAT#4139), HSL (CAT#4107) and peroxisome proliferator-activated receptor gamma (PPARy) (CAT#4685) were from Cell Signaling (Danvers, MA). Antibodies against tubulin (CAT#ab7291) were from Abcam (Cambridge, MA). Horseradish peroxidase-conjugated donkey anti-rabbit and goat antimouse IgG secondary antibodies were purchased from Jackson ImmunoResearch Laboratories (West Grove, PA). Fatty acid-free bovine serum albumin (BSA) (CAT# 152401) was from MP Biomedicals (Solon, OH). Free glycerol reagent (CAT#F6428) was from Sigma (Oakville, ON). A kit for the determination of non-esterified fatty acids (NEFA-HR kit) was purchased from Wako Chemicals (Richmond, VA). An ELISA for the detection of mouse adiponectin (CAT# EZMADP-60K) was from Millipore (Billerica, MA). SuperScript II Reverse Transcriptase, oligo(dT),dNTP and SYBR Green qPCR

SuperMix was products of Invitrogen (Burlington, ON). Tagman Gene Expression Assays for mouse  $\beta$  actin (4352933E), eukaryotic 18S rRNA, TNFalpha (CAT# Mm99999068 m1), adiponectin (CAT# Mm00456425 m1), F4/80 (Emr1) (CAT# Mm00802529 m1), IL-1-beta (CAT# Mm00434228 m1), CxC11 (CAT#Mm00433859\_m1), (CAT# Mm00475988 m1), Arg1 iNOS (CAT# Mm00440502\_m1), Cc12 (CAT#Mm00441242), CD206 (Mm00440502\_m1) and CD11c (CAT# Mm00498698 m1) were from Applied Biosystem (Foster City, CA). All other chemicals were purchased from Sigma (Oakville, ON).

#### **6.2.2 Treatment of Animals**

All protocols followed Canadian Council on Animal Care (CCAC) guidelines and were approved by the University of Guelph Animal Care Committee. Twelve week old male IL-6<sup>-/-</sup> mice (Jackson Laboratories B6.12952-IL6<sup>tmlkopf</sup>/J) and age-matched C57BL/6J wild-type (WT) mice were housed 1 per cage, with a 12/12-hour light/dark cycle. Mice were acclimated to the animal housing facility for 1 week prior to the start of the diet manipulation. At 13 weeks of age, mice were fed a high-fat diet (HFD) *ad libitum* for 10 week. The high-fat diet contained 18.4% protein (casein), 60.3% fat, and 21.3% carbohydrate, and was purchased from Harlan Laboratories Inc. (CAT# TD06414, Madison, WI). Body weight and food intake were determined weekly for the duration of the study. Pre-weighed food was placed in the food hoppers and measured on a per-cage basis. Food intake was determined as the average grams consumed per week over the course of the study.

#### 6.2.3 Glucose and Insulin Tolerance Tests

Intraperitoneal glucose (GTT) and insulin tolerance tests (ITT) were performed at the end of the 10 weeks of HFD. For the GTT, mice were fasted for 6h prior to an intraperitoneal (I.P.) injection of glucose (2g/kg body weight). For the ITT, mice had free access to HFD prior to an I.P. injection of insulin (0.75U/kg body weight). Blood glucose levels were determined by tail vein sampling at the indicated intervals using a glucometer. Changes in glucose over time were plotted, and the area under the curve (AUC) was calculated for each.

#### 6.2.4 In Vivo Insulin Signaling

Mice were anesthetized with sodium pentobarbitol (5mg/100g body weight) and a small incision (~1-2 cm) was made in the abdominal cavity allowing for the removal of the distal tip of one epididymal fat pad. Minimal blood loss occurred during the removal of the pre-insulin sample. Following this, mice were then injected I.P with a weight adjusted (3.75 U/kg body weight) bolus injection of insulin. 15 minutes post injection the contra lateral epididymal fat depot was removed. Adipose tissue was immediately clamp frozen in tongs cooled to the temperature of liquid nitrogen and then stored at -80°C until further analysis. The insulin injection protocol in these experiments allowed for the comparison of the effects of insulin on PKB phosphorylation within the same animal.

#### 6.2.5 Adipose Tissue Explants

Adipose tissue minces (~50 mg cut into 10-20 mg fragments) from WT or IL-6<sup>-/-</sup> mice were incubated in 2.0 ml of oxygenated Krebs-Ringer buffer (KRB) (118 mM NaCl, 4.8 mM KCl, 1.25 mM CaCl<sub>2</sub>, 1.2 mM KH<sub>2</sub>PO<sub>4</sub>, 1.2 mM MgSO<sub>4</sub>, 25 mM NaHCO<sub>3</sub>, 4 mM pyruvate, pH 7.4) with 2.5% fatty acid-free BSA at 37° C in a shaking water bath. Explants were treated with epinephrine (1  $\mu$ M) and IL-6 (150 ng/ml) for either 30 minutes (signaling) or 2 hours (lipolysis). For the determination of insulin signaling explants were incubated in KRB supplemented with vehicle, 0.25 or 1 mU/ml of insulin for 30 minutes. Lipolytic rates are linear during this time period. Media and tissue minces were collected for the determination of fatty acid and glycerol release and the activation of the lipolytic and insulin-signaling pathway respectively. At the end of the experiments adipose tissue minces were rinsed in ice-cold sterile PBS, strained and adipose tissue fragments snap frozen and stored at -80 °C until further analysis.

#### 6.2.6 Plasma Adiponectin and Adiponectin Secretion

The adiponectin levels in plasma and adiponectin secretion in KRB from vehicle treated adipose tissue explants were measured using a mouse specific adiponectin ELISA. Samples were run in duplicate.

#### 6.2.7 Mitochondrial Respiration in Adipose Tissue

High-resolution O<sub>2</sub> consumption measurements were conducted in 2 mL of Miro5 using the OROBOROS Oxygraph-2k (OROBOROS INSTRUMENTS, Corp., Innsbruck, AT) with stirring at 750 rpm at 37 °C. Epididymal adipose tissue was removed and ~ 40-50 mg were minced with scissors (~30 seconds) and placed in an Oxygraph-2k reaction chamber. In agreement with a previous report (18), preliminary experiments demonstrated no additional effect of standard membrane permeabilization detergents (digitonin, saponin) on respirometric assessments. This suggests the surgical and mincing procedures are sufficient for maximal respiration in adipose tissue.

Adipose samples were allowed to incubate for ~10 min before initiating the respirometric protocols. 10 mM glutamate (G) and 5 mM malate (M) were then added as complex I substrates (State IV respiration) followed by 5mM ADP (D; State III respiration). Additional electron entry into complex I and II were achieved with 10mM pyruvate (P) and 20mM succinate (S) respectively. Electron transport system (ETS) capacity was determined via titration of FCCP to induce maximal uncoupled respiration (~typically occurred with 0.5  $\mu$ M FCCP). All experiments were completed before oxygraph chamber [O<sub>2</sub>] reached 100  $\mu$ M, which is well above the threshold of O<sub>2</sub> limitation we have observed in adipose tissue (~60-80  $\mu$ M). Polarographic oxygen measurements were acquired in 2-second intervals, with the rate of respiration derived from 40 data points, and expressed as pmol • s<sup>-1</sup> •  $\mu$ g<sup>-1</sup> protein at 37 °C. Cytochrome *c* was added to test for mitochondrial membrane integrity and respiration was never increased by >10%.

#### 6.2.8 Western Blot Analysis

Protein was extracted from adipose tissue and the content and/or phosophorylation of PKB, CORE1, HSP60, HSL, PPARγ, ERK1/2, JNK, p38MAPK, AMPK and the beta 3 adrenergic receptor determined by Western blotting, as described in detail by our laboratory previously (19-20) and detailed in chapter 3. Signals were detected using enhanced chemiluminesence (PerkinElmer Life Science) and were subsequently quantified by densitometry by Gene Tool according to the manufacturer's instructions (SynGene, ChemiGenius2, PerkinElmer).

#### **6.2.9 Mitochondrial DNA**

Total DNA was isolated using DNeasy blood and tissue kit (Qiagen) and relative mitochondrial DNA copy number was determined as described in detail previously by Smith et al (21). Briefly, mtDNA primers were designed using the mouse mitochondrial genome sequence (Genbank accession number NC\_001665) within the NADH dehydrogenase subunit 5 (ND5) gene: forward, 5'-GCAGCCACAGGAAAATCCG-3'; and reverse, 5'-GTAGGGCAGAGACGGGGAGTTG-3'. Primers measuring genomic content were designed within the solute carrier family 16, member1 gene sequence on chromosome 2 (Genbank accession number NC 005101): forward, 5'-TAGCTGGATCCCTGATGCGA-3'; and reverse, 5'-GCATCAGACTTCCCAGCTTCC-3'. Each assay (25 µl total volume) contained 2.5 µl forward and reverse primers, 5 µl of DNA template, 10 µl of SYBR Green qPCR SuperMix and 5 µl of RNase free water. Relative mtDNA content was calculated by the  $2^{-\Delta\Delta CT}$  method using genomic DNA content as an internal standard with 7500 System SDS Software Version 1.2.1.22 (Applied Biosystems).

#### 6.2.10 Real Time PCR

RNA was isolated from adipose tissue using an RNeasy kit according to the manufacturer's instructions. 1  $\mu$ g of RNA was used for the synthesis of complementary DNA (cDNA) using SuperScript II Reverse Transcriptase, oligo(dT) and dNTP. Real time PCR was performed using a 7500 Fast Real-Time PCR system (Applied Biosytems) or Rotorgene 6000 (Qiagen), as described previously (22). Each assay (20  $\mu$ l total volume) contained 1  $\mu$ l gene expression assay, 1  $\mu$ l of cDNA template, 10  $\mu$ l of Taqman Fast Universal PCR Master Mix and 8  $\mu$ l of RNase free water. For  $\beta$ -actin or 18S, each

50 μl reaction contained 25 μl of PCR Master mix, 2.5 μl each of gene expression assay, 1 μl of cDNA template, and 21.5 μl of RNase free water. Results were normalized to the mRNA expression of β-actin or 18S. Relative differences in gene expression between groups were determined using the  $2^{-\Delta\Delta CT}$  method (23). The amplification efficiencies of the gene of interest and the housekeeping gene were equivalent.

#### **6.2.11 Statistical Analysis**

Data are presented as means  $\pm$  standard error of the mean (SEM). The effects of epinephrine and insulin *ex vivo* on lipolysis and insulin signaling pathways in chow groups between WT and IL-6<sup>-/-</sup> mice were made using a two way ANOVA followed by a LSD post hoc analysis. Comparisons between two groups were made using a Student's T-Test. Statistical significance was established at a *p* level <0.05.

#### 6.3 Results

# 6.3.1 No Phenotypic Differences in Adipose Tissue Metabolism in Chow Fed IL-6<sup>-/-</sup> Mice

*Ex vivo* insulin-induced PKB phosphorylation on both threonine-308 and serine-473 phosphorylation sites were similar in epididymal adipose tissue explants from WT and IL-6 deficient mice (*Figure 6-1A*). Similarly the ability of epinephrine to stimulate lipolysis was not different between genotypes (*Figure 6-1B*). Markers of mitochondrial content such as CORE1 and HSP60 (*Figure 6-1C*) and mitochondrial respiration in response to successive additions of electrons into complex I (GM, GMD, GMDP) and complex II (GMDP-S) and mitochondrial uncoupler (FCCP) were not different between groups (*Figure 6-1D*). There was no difference in adiponectin expression in epididymal adipose tissue between genotypes ( $1.33 \pm 0.18$  fold increase in IL-6<sup>-/-</sup> mice versus WT).



Figure 6-1 Adipose tissue metabolism in chow fed WT and IL-6<sup>-/-</sup> mice. A) *Ex vivo* insulin (0.25 and 1 mU) signaling, B) *ex vivo* epinephrine (1 $\mu$ M, 2hour) induced glycerol and FFA release, C) mitochondrial proteins and D) mitochondrial respiration in epididymal adipose tissue from WT and IL-6<sup>-/-</sup> mice. Data are presented as means + SEM for 6-10 samples per group. Representative Western blots are shown above in A and C. Mitochondrial respiration was measured by high-resolution O<sub>2</sub> consumption. GM (glutamate, malate), GMD (glutamate, malate, ADP), GMDP (glutamate, malate, ADP, pyruvate), GMDPS (glutamate, malate, ADP, pyruvate, succinate).\*p<0.05 compared to vehicle treated condition.

### 6.3.2 Food Intake, Body Weight and Fat Pad Mass in WT and IL-6<sup>-/-</sup> Fed a HFD

There were no differences in body weight, cumulative food intake or epididymal

fat pad mass in WT or IL-6<sup>-/-</sup> mice fed a HFD for 10 weeks (*Table 6-1*).

Table 6-1 Descriptive Characteristics of WT and IL-6<sup>-/-</sup> Mice Fed a High Fat Diet

	WT	КО	р
Initial body weight (g)	$27.6 \pm 0.5$	$27.4 \pm 0.4$	0.80
Final body weight (g)	$41.8 \pm 1.4$	$44.2 \pm 1.0$	0.09
Epididymal fat pad weight (g)	$1.85 \pm 0.41$	$2.14 \pm 0.15$	0.44
Average food intake (g per week)	$18.90\ \pm 0.4$	$18.7 \pm 0.7$	0.78

Data are presented as means  $\pm$  SEM for 9-10 animals per group.

## 6.3.3 IL-6<sup>-/-</sup> Mice Have Impaired Glucose Homeostasis Compared to WT Fed a HFD

Following 10 weeks of a HFD WT and IL-6 deficient mice underwent an I.P. GTT. As seen in *Figure 6-2A and B*, IL-6<sup>-/-</sup> mice had impaired glucose tolerance compared to WT controls. Similarly, IL-6<sup>-/-</sup> mice were insulin intolerant compared to WT as shown by a blunted reduction in blood glucose levels (*Figure 6-2C and D*).



**Figure 6-2 Glucose and insulin tolerance in WT and IL-6**<sup>-/-</sup> **mice fed a HFD.** Glucose curves and the quantified area under the curves for glucose (A, B) and insulin (C, D) tolerance tests. Data are presented as means +SEM for 8-10 mice per group. \* p<0.05 versus WT group at the same time point in A and C. \* p<0.05 versus WT group in B and D.

### 6.3.4 Insulin Signaling Is Attenuated in Adipose Tissue from IL-6<sup>-/-</sup> Mice Fed a HFD

To ascertain if systemic glucose and insulin intolerance were reflected in defects in insulin signaling in adipose tissue we injected mice *in vivo* with insulin and assessed changes in the serine and threonine phosphorylation of PKB. As adipose tissue was procured from the same animal prior to, and following insulin injection the data is expressed as a fold increase by insulin within the same animal. There were no differences in PKB phosphorylation between genotypes in samples taken prior to the injection of insulin (data not shown). As shown in *Figure 6-3A*, the insulin-induced phosphorylation of PKB was markedly attenuated in adipose tissue from IL-6 deficient compared to WT mice. Similar effects were found when insulin signaling was measured in adipose tissue explants *ex vivo* (data not shown). The reductions in PKB phosphorylation were associated with increases in the phosphorylation of ERK and JNK, but not p38 (*Figure 6-3B*).



**Figure 6-3** *In vivo* **insulin signaling in adipose tissue from WT and IL-6<sup>-/-</sup> mice fed a HFD.** A) *In vivo* insulin-mediated (3.75U/kg BW, 15 mins) phosphorylation of PKB on threonine 308 and serine 473. B) The attenuated phosphorylation of PKB is associated with increases in the phosphorylation of ERK and JNK in adipose tissue from IL-6<sup>-/-</sup> mice. Data are presented as means + SEM for 8-10 animals per group. In A) data are presented as fold increases in PKB phosphorylation relative to the control sample from the same animal. Representative Western blot images are given to the right of the quantified data. \* p<0.05 versus WT group.
## 6.3.5 Beta-Adrenergic Signaling Is Attenuated in Adipose Tissue from IL-6<sup>-/-</sup> Mice

To determine if catecholamine signaling is altered in adipose tissue from IL-6 deficient mice fed a HFD we treated epididymal adipose tissue explants with epinephrine (1µM) and measured changes in glycerol and fatty acid release *ex vivo*. Epinephrine significantly increased glycerol and fatty acid release in eWAT from WT (*Figure 6-4A*) but not IL-6<sup>-/-</sup> (*Figure 6-4B*) mice. There were no differences in vehicle treated HSL phosphorylation between groups (data not shown). The attenuation of lipolysis in explants from IL-6<sup>-/-</sup> mice was associated with reductions in the epinephrine-induced phosphorylation of HSL (*Figure 6-4C*). Reductions in beta-adrenergic signaling were not explained by decreases in the protein content of the beta 3 adrenergic receptor (WT 1.00  $\pm$  0.23, KO 1.20  $\pm$  0.18, arbitrary units, N=10 per groups).



Figure 6-4 *Ex vivo* epinephrine-stimulated lipolysis in adipose tissue from WT and IL-6<sup>-/-</sup> mice fed a HFD. Epinephrine-induced (1 $\mu$ M, 2h) glycerol and FFA release in epididymal adipose tissue from A) WT and B) IL-6<sup>-/-</sup> mice fed a HFD. C) Epinephrine-induced (1 $\mu$ M, 30 minutes) increases in the phosphorylation of HSL at ser563. Increases in phosphorylation are expressed relative to the vehicle control condition from the same animal. Data are presented as means + SEM for 8-10 animals per group. Representative Western blot images are given to the right of the quantified data in C. \* *p*<0.05 versus vehicle control within the same genotype in A and versus WT in C.

#### 6.3.6 IL-6 Does Not Potentiate Epinephrine-Stimulated Lipolysis

IL-6 (150 ng/ml) did not increase or potentiate the effects of epinephrine (1  $\mu$ M)

on glycerol release in adipose tissue explants from WT mice (Table 6-2).

Table 6-2 Effects of IL-6 alone, or in Combination with Epinephrine, on GlycerolRelease in Epididymal Adipose Tissue Explants from Male C57BL/6J Mice FedChow.

Experimental Condition	Glycerol Release	
	(µmol/gram tissue/2hours)	
vehicle	$1.31 \pm 0.24$	
IL-6 (150 ng/ml)	$1.45 \pm 0.35$	
epinephrine (1 µM)	$2.89 \pm 0.54*$	
IL-6 + epinephrine	$2.88 \pm 0.4*$	

Data are means  $\pm$  SEM for 5-8 explants per group. \* p<0.05, significant effects compared to vehicle or IL-6 alone determined by 2-way ANOVA.

# 6.3.7 Reductions in Adiponectin mRNA Expression and Secretion in Adipose Tissue from IL-6<sup>-/-</sup> Mice

As seen in *Figure 6-5A*, adiponectin mRNA expression was reduced ~ 50% in adipose tissue from HFD-fed IL-6<sup>-/-</sup> mice concomitant with reductions in the protein content of PPAR $\gamma$  (*Figure 6-5B*). Despite plasma adiponectin levels being similar between genotypes (*Figure 6-5C*), adiponectin secretion *ex vivo* was reduced in eWAT from IL-6 deficient mice (*Figure 6-5D*).



**Figure 6-5 Adiponectin expression, secretion and plasma levels in WT and IL-6**<sup>-/-</sup> **mice fed a HFD.** A) Adiponectin mRNA expression and B) PPAR $\gamma$  protein expression in epididymal adipose tissue from WT IL-6<sup>-/-</sup> mice. C) Plasma adiponectin levels and D) adiponectin secretion from epididymal adipose tissue from WT and IL-6<sup>-/-</sup> mice fed a HFD. Data are presented as means + SEM for 8-10 animals per group. The mRNA data is normalized to 18S and expressed as fold differences compared to the WT group in A. Western blot images are given above the quantified data in B. \* *p*<0.05 versus WT group.

# 6.3.8 Markers of Inflammation and Macrophage Infiltration in Adipose Tissue from IL-6<sup>-/-</sup> Mice

As shown in *Figure 6-6*, there were modest increases in markers of macrophage infiltration such as EmR1 (F4/80) in eWAT. Interestingly the expression of markers of pro-inflammatory macrophages such as CD11c was not increased, while CD206, a marker of macrophages in the M2 state was significantly increased. The inducible nitric oxide synthase (iNOS) to arginase ratio was reduced in adipose tissue from IL-6<sup>-/-</sup> mice.



**Figure 6-6 mRNA markers of inflammation in eWAT from WT and IL-6**<sup>-/-</sup> **fed a HFD.** Markers of macrophage infiltration are increased while the iNOS/arginase is reduced in adipose tissue from IL-6 deficient fed a high fat diet. Data are presented as means + SEM for 8-10 animals per group. The mRNA data is normalized to beta actin and expressed as fold differences compared to the WT group. \* p<0.05 versus WT group. (For figure 6-6, the relative mRNA expression of EmR1, CD206, CD11c, TNF- $\alpha$ , IL-1 $\beta$ , CXCL1, CCL2, iNOS and arginase in WT HFD group is 1.19 ±0.21, 1.03 ±0.08, 1.15 ±0.20, 1.14 ±0.20, 1.16 ±0.23, 1.24 ±0.31, 1.10 ±0.15, 1.13 ±0.18 and 1.31 ±0.52, respectively.)

#### 6.3.9 Mitochondrial Content Is Not Different in Adipose Tissue from WT and IL-6<sup>-/-</sup>

#### Mice fed a HFD

The phosphorylation of AMPK, a reputed mediator of mitochondrial biogenesis in adipose tissue, was reduced in adipose tissue from IL-6<sup>-/-</sup> mice (*Figure 6-7A*). Despite the

reduction in AMPK, mitochondrial respiration was not altered in adipose tissue from IL- $6^{-/-}$  compared to WT mice fed a HFD as demonstrated by similar levels of respiration in response to successive additions of electrons into complex I (GM, GMD) and complex II (GMDP-S) and FCCP (*Figure 6-7B*). Data were similar whether expressed relative to tissue weight (data not shown) or total protein content. The content of mitochondrial marker proteins (*Figure 6-7C*) and mitochondrial DNA copy number (*Figure 6-7D*) were not different between genotypes.



Figure 6-7 AMPK and mitochondrial content and respiration in adipose tissue from WT and IL-6<sup>-/-</sup> fed a HFD. A) AMPK phosphorylation, B) mitochondrial respiration, C) mitochondrial marker proteins and D) mtDNA content in epididymal adipose tissue from WT and IL-6<sup>-/-</sup> mice fed a HFD. Data are presented as means + SEM for 8-10 animals per group. Western blot images are given above the quantified data in A and C. Mitochondrial respiration was measured by high-resolution  $O_2$  consumption. GM (glutamate, malate), GMD (glutamate, malate, ADP), GMDPS (glutamate, malate, ADP, pyruvate, succinate). Relative mtDNA content is normalized to genomic MCT1 DNA content and expressed as fold differences compared to the WT group in D. \* p<0.05 versus WT group in A.

#### **6.4 Discussion**

Elevations in IL-6 have traditionally been thought of as a causal event in the development of insulin resistance. For instance IL-6 administration both *in vivo* and *in vitro* has been shown to cause insulin resistance in hepatocytes (4-5). In contrast to these studies, previous work has clearly shown that IL-6 deficient mice develop impaired glucose tolerance and insulin resistance when fed a chow diet (7). When challenged with a HFD, IL-6 deficient mice display signs of liver damage, hepatocyte inflammation and reductions in mitochondrial content in the liver (7). While these are not universal findings (24) they would suggest that a re-evaluation of the tissue specific functions of IL-6 is needed.

Adipose tissue is a key contributor to the regulation of systemic fuel metabolism through the release of fatty acids and various adipokines, such as adiponectin (10). Increases in fatty acid release and reductions in adiponectin secretion are hallmark features of insulin resistance (10). Given the surprising phenotype of IL-6 deficient mice we sought to explore the effects of systemic IL-6 deficiency on adipose tissue metabolism. In our initial experiments we examined insulin signaling, lipolysis and mitochondrial content in adipose tissue from WT and IL-6<sup>-/-</sup> mice fed a low fat chow diet. Surprisingly, despite the reported impaired glucose homeostasis in chow fed IL-6<sup>-/-</sup> mice (7-8), we found that there were no phenotypic differences in epididymal adipose between WT and IL-6 deficient mice. Given these unexpected results we reasoned that the provision of a metabolic challenge to IL-6 deficient mice, in the form of a HFD, could uncover genotypic differences in adipose tissue metabolism.

Similar to some (7), but not all (24) previous studies, we found that IL-6<sup>-/-</sup> mice fed a HFD were more insulin and glucose intolerant than WT mice given a HFD. The attenuation in glucose homeostasis was associated with reductions in insulin-stimulated PKB phosphorylation in eWAT *in vivo*. The activation of stress kinases such as ERK (25) and JNK (26) have been shown to inhibit insulin signaling and thus the greater activation of these kinases in adipose tissue from IL-6 deficient mice could explain the abrogated activation of PKB. These results are similar to what has been reported in liver from IL-6<sup>-/-</sup> mice (7).

Elevations in fatty acid levels have been causally linked to the pathogenesis of insulin resistance (27-28) and thus it was of interest to determine if fatty acid release was perturbed in adipose tissue from IL- $6^{-/-}$  mice. Similar to the attenuated insulin-mediated activation of PKB, we found that the ability of epinephrine to stimulate glycerol and FFA release and to increase the phosphorylation of HSL was blunted in epididymal adipose tissue from IL-6 deficient mice. These results suggest that increases in lipolysis do not account for the observed differences in systemic glucose homeostasis between genotypes, a finding that is consistent with previous work reporting similar plasma fatty acid levels in WT and IL- $6^{-/-}$  mice (7).

The role of IL-6 in stimulating lipolysis is controversial. While some studies have demonstrated a direct effect of IL-6 on lipolysis in adipocytes (29-30), others have shown that IL-6 infusions do not stimulate adipose tissue lipolysis in humans (31). Based on the attenuated catecholamine-induced lipolysis in IL-6<sup>-/-</sup> mice, it is tempting to speculate that IL-6 may play a synergistic role with epinephrine in mediating lipolysis. However, arguing against this premise we found that IL-6 treatment did not potentiate the effects of

epinephrine on lipolysis *ex vivo*. Moreover, as catecholamine mediated lipolysis was similar in adipose tissue from chow fed WT and IL-6<sup>-/-</sup> mice it would suggest that the blunted effects of epinephrine in adipose from IL-6 deficient mice were not the direct result of the absence of IL-6. Rather our findings would point towards an indirect role of IL-6 in mediating lipolysis under conditions of chronic nutrient excess.

Adiponectin is an adipose tissue-derived hormone, or adipokine, that possesses noted insulin sensitizing effects in a wide range of tissues including adipose (32). Moreover, recent work from Scherer's group has provided evidence that adiponectin plays a role in sensitizing adipocytes to beta-adrenergic stimulation (17). Given these findings it would seem reasonable to assume that alterations in adiponectin may account for some of the observed phenotypic differences between WT and IL-6 deficient mice. Despite reductions in adiponectin mRNA levels, PPAR $\gamma$  protein content and *ex vivo* adiponectin secretion in adipose tissue from IL-6<sup>-/-</sup> mice, plasma adiponectin levels were not different between groups. These results suggest that there are differences in the clearance/degradation of adiponectin, or an upregulation of adiponectin secretion from another fat depot in IL-6<sup>-/-</sup> mice. Regardless, our results would imply that reductions in adiponectin could be modulating adipose tissue beta-adrenergic responsiveness through an autocrine/paracrine mechanism, a concept that has previously been put forth by others (33).

Adiponectin increases the activity of AMPK (34), a reputed mediator of mitochondrial biogenesis in adipose tissue (12). Moreover, AICAR, which activates AMPK, can induce adiponectin mRNA expression in adipocytes (15). Similarly, it has been suggested that a disruption of mitochondrial function or decreases in mitochondrial

content lead to reductions in adiponectin synthesis (16). Collectively these findings point towards a complex interplay between adiponectin, AMPK and mitochondrial biogenesis. In our model, despite seeing reductions in the phosphorylation of AMPK, markers of mitochondrial content and respiration were not different in adipose tissue from WT and IL-6 deficient mice. These findings are somewhat surprising as they demonstrate a lack of association between not only AMPK and mitochondrial biogenesis, but mitochondrial content/function and adiponectin as well. As AICAR has been shown to induce adiponectin expression (15), it is tempting to speculate that reductions in AMPK phosphorylation could be linked to decreases in adiponectin secretion/expression. However, in adipose tissue from chow fed IL-6<sup>-/-</sup> mice, despite reductions in AMPK activity (35), the expression of adiponectin mRNA was not reduced. These findings demonstrate that decreases in AMPK alone are not sufficient to lead to reductions in adiponectin expression and suggest the involvement of additional, yet to be resolved mechanisms.

Emerging evidence has suggested that inflammation can decrease the activity of AMPK (36). Given this we were surprised to find only marginal alterations in markers of inflammation in adipose tissue from IL-6<sup>-/-</sup> mice. If anything, our results, particularly the reduction in the iNOS to arginase ratio, demonstrates a reduction in specific markers of inflammation in adipose tissue from IL-6 deficient mice. Although these findings do not fit with previous work from Febbraio's group (7) who elegantly demonstrated increases in hepatic inflammation in IL-6 knockout mice, our results are not entirely unexpected given the pronounced effect of IL-6 on the induction of pro-inflammatory processes in adipose tissue.

The current study, for the first time, clearly demonstrates phenotypic differences in adipose tissue from WT and IL-6<sup>-/-</sup> mice when fed a HFD. Although we cannot ascertain with certainty if these effects are directly due to a loss of IL-6 signaling, or secondary to changes in other circulating factors, they point towards a unique role for IL-6 in the regulation of adipose tissue metabolic flexibility. In contrast to mice fed a HFD we did not see differences in adipose tissue insulin signaling or beta-adrenergic responsiveness in WT or IL-6 deficient mice fed a low fat diet. As chow fed IL-6<sup>-/-</sup> mice from the same commercial distributor as used in the current study are insulin resistant compared to WT controls (8), this would suggest that perturbations in adipose tissue metabolism are not a causal event in the etiology of impaired glucose homeostasis in IL-6 knockout mice and likely implicates skeletal muscle and/or liver as the primary site of impairment. Although speculative, our results would further suggest that the increase in IL-6 during the development of obesity and insulin resistance (37-38) might serve as a protective mechanism in the maintenance of adipose tissue metabolism.

#### **6.5 Literature Cited**

1. Carey AL, Febbraio MA. Interleukin-6 and insulin sensitivity: friend or foe? Diabetologia. 2004;47(7):1135-42.

2. Pickup JC, Mattock MB, Chusney GD, Burt D. NIDDM as a disease of the innate immune system: association of acute-phase reactants and interleukin-6 with metabolic syndrome X. Diabetologia. 1997;40(11):1286-92.

3. Rotter V, Nagaev I, Smith U. Interleukin-6 (IL-6) induces insulin resistance in 3T3-L1 adipocytes and is, like IL-8 and tumor necrosis factor-alpha, overexpressed in human fat cells from insulin-resistant subjects. J Biol Chem. 2003;278(46):45777-84.

4. Klover PJ, Zimmers TA, Koniaris LG, Mooney RA. Chronic exposure to interleukin-6 causes hepatic insulin resistance in mice. Diabetes. 2003;52(11):2784-9.

5. Senn JJ, Klover PJ, Nowak IA, Mooney RA. Interleukin-6 induces cellular insulin resistance in hepatocytes. Diabetes. 2002;51(12):3391-9.

6. Carey AL, Steinberg GR, Macaulay SL, Thomas WG, Holmes AG, Ramm G, et al. Interleukin-6 increases insulin-stimulated glucose disposal in humans and glucose uptake and fatty acid oxidation in vitro via AMP-activated protein kinase. Diabetes. 2006;55(10):2688-97.

7. Matthews VB, Allen TL, Risis S, Chan MH, Henstridge DC, Watson N, et al. Interleukin-6-deficient mice develop hepatic inflammation and systemic insulin resistance. Diabetologia. 2010;53(11):2431-41.

175

8. Fritsche L, Hoene M, Lehmann R, Ellingsgaard H, Hennige AM, Pohl AK, et al. IL-6 deficiency in mice neither impairs induction of metabolic genes in the liver nor affects blood glucose levels during fasting and moderately intense exercise. Diabetologia. 2010;53(8):1732-42.

9. Nielsen NB, Hojbjerre L, Sonne MP, Alibegovic AC, Vaag A, Dela F, et al. Interstitial concentrations of adipokines in subcutaneous abdominal and femoral adipose tissue. Regul Pept. 2009;155(1-3):39-45.

10. Rosen ED, Spiegelman BM. Adipocytes as regulators of energy balance and glucose homeostasis. Nature. 2006;444(7121):847-53.

11. Kelly M, Keller C, Avilucea PR, Keller P, Luo Z, Xiang X, et al. AMPK activity is diminished in tissues of IL-6 knockout mice: the effect of exercise. Biochem Biophys Res Commun. 2004;320(2):449-54.

12. Gaidhu MP, Fediuc S, Anthony NM, So M, Mirpourian M, Perry RL, et al. Prolonged AICAR-induced AMP-kinase activation promotes energy dissipation in white adipocytes: novel mechanisms integrating HSL and ATGL. J Lipid Res. 2009;50(4):704-15.

13. Deng T, Sieglaff DH, Zhang A, Lyon CJ, Ayers SD, Cvoro A, et al. A peroxisome proliferator-activated receptor gamma (PPARgamma)/PPARgamma coactivator 1beta autoregulatory loop in adipocyte mitochondrial function. J Biol Chem. 2011;286(35):30723-31.

176

14. Chen XH, Zhao YP, Xue M, Ji CB, Gao CL, Zhu JG, et al. TNF-alpha induces mitochondrial dysfunction in 3T3-L1 adipocytes. Mol Cell Endocrinol. 2010;328(1-2):63-9.

15. Lihn AS, Jessen N, Pedersen SB, Lund S, Richelsen B. AICAR stimulates adiponectin and inhibits cytokines in adipose tissue. Biochem Biophys Res Commun. 2004;316(3):853-8.

16. Koh EH, Park JY, Park HS, Jeon MJ, Ryu JW, Kim M, et al. Essential role of mitochondrial function in adiponectin synthesis in adipocytes. Diabetes. 2007;56(12):2973-81.

17. Asterholm IW, Scherer PE. Enhanced metabolic flexibility associated with elevated adiponectin levels. Am J Pathol. 2010;176(3):1364-76.

18. Kraunsoe R, Boushel R, Hansen CN, Schjerling P, Qvortrup K, Stockel M, et al. Mitochondrial respiration in subcutaneous and visceral adipose tissue from patients with morbid obesity. J Physiol. 2010;588(Pt 12):2023-32.

19. Sutherland LN, Bomhof MR, Capozzi LC, Basaraba SA, Wright DC. Exercise and adrenaline increase PGC-1{alpha} mRNA expression in rat adipose tissue. J Physiol. 2009;587(Pt 7):1607-17.

20. Wan Z, Thrush AB, Legare M, Frier BC, Sutherland LN, Williams DB, et al. Epinephrine-mediated regulation of PDK4 mRNA in rat adipose tissue. Am J Physiol Cell Physiol. 2010;299(5):C1162-70.

21. Smith BK, Jain SS, Rimbaud S, Dam A, Quadrilatero J, Ventura-Clapier R, et al. FAT/CD36 is located on the outer mitochondrial membrane, upstream of long-chain acyl-CoA synthetase, and regulates palmitate oxidation. Biochem J. 2011;437(1):125-34.

22. Schertzer JD, Tamrakar AK, Magalhaes JG, Pereira S, Bilan PJ, Fullerton MD, et al. NOD1 activators link innate immunity to insulin resistance. Diabetes. 2011;60(9):2206-15.

23. Livak KJ, Schmittgen TD. Analysis of relative gene expression data using realtime quantitative PCR and the 2(-Delta Delta C(T)) Method. Methods. 2001;25(4):402-8.

24. Di Gregorio GB, Hensley L, Lu T, Ranganathan G, Kern PA. Lipid and carbohydrate metabolism in mice with a targeted mutation in the IL-6 gene: absence of development of age-related obesity. Am J Physiol Endocrinol Metab. 2004;287(1):E182-7.

25. Fujishiro M, Gotoh Y, Katagiri H, Sakoda H, Ogihara T, Anai M, et al. Three mitogen-activated protein kinases inhibit insulin signaling by different mechanisms in 3T3-L1 adipocytes. Mol Endocrinol. 2003;17(3):487-97.

26. Hirosumi J, Tuncman G, Chang L, Gorgun CZ, Uysal KT, Maeda K, et al. A central role for JNK in obesity and insulin resistance. Nature. 2002;420(6913):333-6.

27. Kim JK, Fillmore JJ, Sunshine MJ, Albrecht B, Higashimori T, Kim DW, et al. PKC-theta knockout mice are protected from fat-induced insulin resistance. J Clin Invest. 2004;114(6):823-7.

28. Yu C, Chen Y, Cline GW, Zhang D, Zong H, Wang Y, et al. Mechanism by which fatty acids inhibit insulin activation of insulin receptor substrate-1 (IRS-1)-associated phosphatidylinositol 3-kinase activity in muscle. J Biol Chem. 2002;277(52):50230-6.

29. Petersen EW, Carey AL, Sacchetti M, Steinberg GR, Macaulay SL, Febbraio MA, et al. Acute IL-6 treatment increases fatty acid turnover in elderly humans in vivo and in tissue culture in vitro. Am J Physiol Endocrinol Metab. 2005;288(1):E155-62.

30. Yang Y, Yang G. Rosiglitazone regulates IL-6-stimulated lipolysis in porcine adipocytes. Biochem Cell Biol. 2010;88(5):853-60.

31. Wolsk E, Mygind H, Grondahl TS, Pedersen BK, van Hall G. IL-6 selectively stimulates fat metabolism in human skeletal muscle. Am J Physiol Endocrinol Metab. 2010;299(5):E832-40.

32. Sevillano J, de Castro J, Bocos C, Herrera E, Ramos MP. Role of insulin receptor substrate-1 serine 307 phosphorylation and adiponectin in adipose tissue insulin resistance in late pregnancy. Endocrinology. 2007;148(12):5933-42.

33. Dietze-Schroeder D, Sell H, Uhlig M, Koenen M, Eckel J. Autocrine action of adiponectin on human fat cells prevents the release of insulin resistance-inducing factors. Diabetes. 2005;54(7):2003-11.

34. Liu Q, Gauthier MS, Sun L, Ruderman N, Lodish H. Activation of AMP-activated protein kinase signaling pathway by adiponectin and insulin in mouse adipocytes:

requirement of acyl-CoA synthetases FATP1 and Acsl1 and association with an elevation in AMP/ATP ratio. FASEB J. 2010;24(11):4229-39.

35. Wan Z, Ritchie I, Beaudoin MS, Castellani L, Chan CB, Wright DC. IL-6 Indirectly Modulates the Induction of Glyceroneogenic Enzymes in Adipose Tissue during Exercise. PLoS One. 2012;7(7):e41719.

36. Steinberg GR. Inflammation in obesity is the common link between defects in fatty acid metabolism and insulin resistance. Cell Cycle. 2007;6(8):888-94.

37. Fisette A, Munkonda MN, Oikonomopoulou K, Paglialunga S, Lambris JD, Cianflone K. C5L2 receptor disruption enhances the development of diet-induced insulin resistance in mice. Immunobiology. 2012.

38. Nara N, Nakayama Y, Okamoto S, Tamura H, Kiyono M, Muraoka M, et al. Disruption of CXC motif chemokine ligand-14 in mice ameliorates obesity-induced insulin resistance. J Biol Chem. 2007;282(42):30794-803

### **Chapter 7 General Discussion**

#### 7.1 Summary of Main Findings

The aim of this thesis is 1) to understand exercise and the role of hormonal factors (epinephrine and IL-6) that mediate the regulation of pyruvate dehydrogenase kinase 4 (PDK4) and phosphoenolpyruvate carboxykinase (PEPCK) gene expression in rodent white adipose tissue (WAT) and the potential signalling pathways therein; and 2) to understand the metabolic action of IL-6 on rodent WAT metabolism by utilizing IL-6<sup>-/-</sup> mice under both chow and a high fat diet (HFD) conditions.

First of all, we determined if an acute bout of exercise and epinephrine treatment can induce PDK4 mRNA expression in epididymal adipose tissue (eWAT) from lean rats and explored the potential signalling pathways that might be involved in. We found that exercise, *in* or *ex vivo* epinephrine treatment increased PDK4 mRNA and p38 mitogenactivated protein kinase (MAPK) and peroxisome proliferator-activated receptor gamma (PPAR $\gamma$ ) could be involved in epinephrine-mediated regulation of PDK4 mRNA in eWAT from lean and insulin sensitive rats. Whereas, 5' AMP activated protein kinase (AMPK) agonists including 5-aminoimidazole-4-carboxyamide ribonucleoside (AICAR) and metformin decreased PDK4 mRNA expression in cultured eWAT.

Second, we determined the regulation of PDK4 mRNA expression in eWAT from obese and insulin resistant rats by an acute bout of exercise and acute increases in epinephrine *in vivo*. We demonstrated that the abilities of exercise and epinephrine to induce PDK4 mRNA were maintained in eWAT from obese and insulin resistant rats, while the pathways mediating the effects of epinephrine on PDK4 mRNA in obese rats may be distinct from those in lean animals.

Third, we determined the effects of *ex vivo* IL-6 treatment on PEPCK and PDK4 expression in cultured mouse eWAT, and the effects of an acute bout of treadmill running on IL-6 signaling and PEPCK and PDK4 mRNA expression in mouse eWAT. We also elucidated the role of IL-6 in modulating the effects of exercise on PEPCK and PDK4 mRNA expression and the putative signaling pathways therein by utilizing an IL-6<sup>-/-</sup> mouse model. We found that *ex vivo* IL-6 treatment decreased PEPCK and PDK4 mRNA and protein expression, and increased the phosphorylation of AMPK, acetyl coA carboxylase (ACC) and signal transducer and activator of transcription 3 (STAT3) and induced suppressor of cytokine signaling 3 (SOCS3) mRNA expression. The activation of AMPK by IL-6 was independent of increases in lipolysis. IL-6 signaling was not activated immediately following an acute bout of treadmill running (15 meters/min, 5% incline, 90 mins). Exercise-induced increases in PEPCK and PDK4 mRNA were attenuated in mouse eWAT from IL-6<sup>-/-</sup> mice.

Finally, we characterized the adipose tissue phenotypes between age matched wild type (WT) and IL-6<sup>-/-</sup> mice under both chow and HFD (10 weeks) conditions. In IL-6<sup>-/-</sup> mice fed a normal chow diet, there were no differences in eWAT insulin action, beta-adrenergic responsiveness or mitochondrial content/function compared to WT controls. IL-6<sup>-/-</sup> mice fed a HFD, were more glucose and insulin intolerant than WT controls fed the same diet and this was mirrored by reductions in the insulin-stimulated activation of protein kinase B (PKB) and increases in the phosphorylation of extracellular signal regulated kinases (ERK) and c-Jun N-terminal kinases (JNK) in eWAT. Epinephrine

stimulated lipolysis and hormone sensitive lipase (HSL) phosphorylation were blunted in eWAT from IL-6<sup>-/-</sup> mice. These changes were associated with decreased mRNA expression and secretion of adiponectin and AMPK phosphorylation. However, mitochondrial content and respiration in eWAT were similar between genotypes under HFD challenge.

In this chapter, the results we have obtained will be discussed and put into a broader perspective and important issues for future research will be addressed. First, methodological considerations for the specific models that I used will be discussed.

#### 7.2 Methodological Considerations

#### 7.2.1 Adipose tissue organ culture

One of the main techniques used in this thesis was adipose tissue organ culture (ATOC). Compared with primary adipocyte culture or *in vitro* cell lines, it has advantages and disadvantages. The major advantages of ATOC is 1) the paracrine and/or autocrine interactions between adipocytes and other cell types in adipose tissue remain intact and more representative of what is seen *in vivo* compared to isolated adipocyte preparations; and 2) depot-specific differences in adipose tissue remain during the culture procedure (1). The primary disadvantage of using ATOC is that it can't be determined whether the observed effects are owing to direct effects on adipocytes *per se* or on other types of cells resident within the adipose tissue.

Primary culture of adipocytes could be a complementary approach to differentiate whether the effect is on adipocytes themsleves. However, it has been reported that the isolation of adipocytes could result in the induction of the inflammatory response and decrease key adipose-specific genes such as glucose transporter type 4 (GLUT4) and PPAR $\gamma$  (2). Isolation of adipocytes is labor intensive (1), and the crosstalk between adipocytes and nonadipose cells within adipose tissue is lost. Meanwhile, the primary culture of adipocytes is unable to completely recapitulate the whole adipocyte population since it is limited by the specific selected adipocytes that are recovered. For instance, owing to the fragile nature of adipocytes, the larger fat cells are particularly easy to be disrupted and lost during the isolation procedure (3).

## 7.2.2 Selection of adipose depots

As has been detailed in chapter 1, there are fundamental differences among different adipose depots with regards to their influence on adipose tissue metabolism. An increased delivery of non-esterified fatty acids (NEFAs) from visceral adipose tissue (VAT) to the liver via the portal vein, can cause liver insulin resistance, therefore visceral fat overaccumulation has a more negative impact on health than subcutaneous fat (4). Omental-mesenteric, gonadal and retroperitoneal adipose tissues are considered to be representative of VAT and are most commonly used in rodent models of metabolic studies. Omental-mesenteric fat drains into the portal vein directly (5), however, the isolation and acquisition of mesenteric adipose tissue is difficult (6). Although epididymal adipose tissue drains into the systemic circulation (7), it has been demonstrated that the removal of gonadal, but not mesenteric or retroperitoneal fat, improves insulin action and glucose tolerance in mice (8). Consequently, this thesis focused on epididymal adipose tissue.

#### 7.2.3 Exercise models

#### 7.2.3.1 Swimming

One of the major differences between swimming and running is their distinct differences in the sympathoadrenal function (9) with swimming inducing greater increase in circulating catecholamine levels (9-10). The major advantages of swimming include 1) less mechanical stress and/or damage (11); and 2) cost-effective as swimming requires less expensive apparatus than treadmill running. The main disadvantages of swimming are that the exercise intensity can't be quantitatively measured and accordingly, the applicability of the findings to exercise prescription in a clinical setting is limited. Meanwhile, researchers must be extremely alert to make sure animals swim continuously.

The water temperature is also an important factor for influencing swimming performance. If the water temperature is greater than 42  $\ \C$  or lower than 20  $\ \C$ , the exercise performance will be greatly affected (12). Most swimming studies are conducted at a water temperature between 32 and 36  $\ \C$  (13-16).

#### 7.2.3.2 Treadmill running

The main advantage of treadmill running is that researchers can precisely manipulate the duration and the intensity of exercise and thus can keep all the mice running at the same level in the study. The major disadvantage of using treadmill running is the use of external stimuli to keep animals running, such as continuous tail shock. It has been reported that shock can elicit significant increases in both plasma norepinephrine and epinephrine levels (9). Meanwhile, treadmill training requires a specialized device, which is more expensive than that required for swimming. Like swimming, researchers must also be vigilant to make sure all the animals finish the designed exercise protocol.

### 7.2.3.3 Gender differences in response to exercise

There are gender differences in response to exercise in rodent models. For instance, in contrast to male rats, female rats lose little or no weight during intensive training (17). Female mice of C57BL/6J strain have increased treadmill running ability and greater capacity to increase their cardiac mass when challenged with similar amounts of exercise compared with their male counterparts (18). From a mechanistic perspective, in skeletal muscle, estrogen has been shown to activate AMPK (19) and affect genes involved in glucose and lipid metabolism such as peroxisome proliferator-activated receptor gamma coactivator 1-alpha (PGC-1 $\alpha$ ), PPAR $\alpha$ , PPAR $\beta/\delta$ , carnitine palmitoyltransferase I (CPT1), sterol regulatory element-binding protein-1c (SREBP-1c) and GLUT4 (20). In adipose tissue, estradiol supplementation has been reported to affect adipose triglyceride lipase (ATGL) and perilipin expression in ovariectomized mice (21). Since AMPK, PGC-1 $\alpha$  and lipolytic signaling pathways are, or are closely related, to the endpoints of this thesis research, we focused on male rodents.

## 7.2.4 IL-6<sup>-/-</sup> mouse model

To elucidate the role of IL-6 in modulating the effects of exercise on PEPCK and PDK4 mRNA expression and in regulating adipose tissue metabolism under HFD, we utilized mice with a global deletion of IL-6. Both IL-6<sup>-/-</sup> mice (B6.12952-IL6<sup>tmlkopf/</sup>J) and their age matched WT mice (C57BL/6J) were purchased from the Jackson Laboratory. The IL-6 targeted mutation was originally created by Kopf et al (22). The Jackson

Laboratory mice database provides the following information for the IL- $6^{-/-}$  mice used in this thesis:

".....a targeting vector was designed to place a neomycin resistance cassette into the first coding exon (exon 2) of the targeted gene. This construct was electroporated into 129S2/SvPas-derived D3 embryonic stem (ES) cells. Correctly targeted ES cells were injected into C57BL/6J blastocysts and the resulting mice were sent to The Jackson Laboratory where the mouse line was bred to C57BL/6J (Stock No. 00664) mice for 11 generations".

It has been reported that 129 background genes can be reduced to 0.1% of all genes if back-crossing 10 generations (23). Thus, IL-6<sup>-/-</sup> and WT mice utilized in the current thesis have an almost identical strain background, i.e. C57BL/6J.

It should be mentioned that there are also some limitations for using an IL-6<sup>-/-</sup> mouse model. Like other transgenic or knockout mice, the interpretation of results generated from IL-6<sup>-/-</sup> mouse model should be viewed cautiously. The total absence of any factor may result in compensatory adaptations and alterations during development (24).

This thesis aimed to investigate the effects of IL-6 on adipose tissue metabolism. A systemic deletion of IL-6 might be unable to differentiate whether the observed effects are on adipose tissue *per se* or compensatory effects in other organs. Future studies via adipose-tissue-specific IL-6 knockout or transgenic mice may shed more light on the effects of IL-6 on adipose tissue metabolism.

#### 7.2.5 HFD

Over the past several decades, multiple rodent models of insulin resistance have been utilized in type 2 diabetes (T2DM) related research. These mainly include 1) genetic rodent models such as Zucker Diabetic Fatty (ZDF) rats, Otsuka Long Evans Tokushima Fatty (OLETF) rats, *ob/ob* and *db/db* mice; and 2) diet induced obese (DIO) rats and mice. Diabetes develops spontaneously in male ZDF rats at ages 8-10 weeks owing to leptin receptor mutation (25). OLETF rats develop impaired glucose regulation at  $\sim 8$  weeks of age and progress to diabetes by ~24 weeks of age (26) due to a spontaneous lack of cholecystokinin-1 receptor (27). Mice lacking the functional leptin (ob/ob) or leptin receptors (db/db) develop diabetes at a relatively early age ~10 weeks (28-29). Although these genetic rodent models are useful in exploring the molecular mechanisms in the pathogenesis of T2DM, the primary limitation is that they are unable to accurately model human situations as diabetes and T2DM in modern society is induced mainly by environmental manipulation (30). Alternatively, high fat feeding-induced obese rodent models may more closely mirror the development of obesity and insulin resistance in humans.

To determine if an acute bout of exercise and acute increases in epinephrine could induce PDK4 mRNA expression in adipose tissue from insulin resistant rats, male Wistar rats weighing 200 g were fed a chow or a high fat diet *ad libitum* for 6 weeks. Previous findings from our lab had shown that 6 weeks of HFD in rats resulted in abdominal obesity, fasting hyperinsulinemia, adipose tissue insulin resistance, and decreased adipose tissue mitochondrial content (31), thus it served as a suitable diet induced insulin resistant model for *objective 2*. The C57Bl/6J mouse strain is an obesity prone mouse line (32). HFD induced obesity in C57BL/6J mouse strain was first described by Surwit et al. (33) in 1988. When challenged with a HFD, C57BL/6J mice exhibit severe obesity, hyperglycemia, hyperinsulinemia, hypertension, progressive peripheral and then central leptin resistance (33-34). In contrast, C57BL/6J mice remain euglycemic and lean on low fat, chow diets (35). DIO in the C57BL/6J genetic background has been widely used for studying the mechanisms underlying the progression or resistance to T2DM (36-37). Meanwhile, HFD is commonly used in transgenic mouse models to uncover the importance of specific genes in the etiology of insulin resistance (38).

It should be noted, however, that there are some limitations to HFD-induced obesity in the current study: First, the high percentage of calories from fat (60% calories from fat) and the fatty acid profile in the diet (fat mainly from lard) is rarely seen in most populations. Second, not all of the phenotypes induced by HFD in C57BL/6J mice mimic the situations in humans. For instance, HFD resulted in an increase in plasma high density lipoprotein-cholesterol (HDL-C) in almost all mouse strains (39), whereas in humans one of the hallmarks of the metabolic syndrome is reduced levels in HDL-C (40). Thus, further studies are required to develop diets that will more closely represent most western diet: a diet with less energy from fat (20-35% from fat) (41), as well as a diet enriched in mono-unsaturated fatty acids and poly-unsaturated fatty acids (42).

#### 7.3 Exercise on Adipose Tissue Metabolism

Exercise is a fundamental tool in the management of T2DM. However, the biochemical and molecular mechanisms underlying the beneficial effects of exercise have remained elusive. Thiazolidinediones (TZDs) are a commonly prescribed medication

used in the treatment of T2DM. A large portion of TZD's beneficial effects on insulin sensitivity is through its upregulation on PEPCK and PDK4 activity in adipose tissue, resulting in increased fatty acid re-esterification through the glyceroneogenic pathway. This will consequently prevent ectopic lipid deposition (43-44). While effective, these drugs cause a wide range of side effects (45) and thus the identification of alternative approaches that mimic the beneficial effects of these compounds on adipose tissue function is needed. Within this broad context, a central effort of this thesis has been exploring the hormonal factors, underlying biochemical/cellular mechanisms, mediating the effects of exercise on glyceroneogenic enzymes in adipose tissue.

In this thesis, both swimming exercise in rats and treadmill running in mice led to rapid increase in PDK4 mRNA. Similarly, we also observed increased PEPCK mRNA expression in mice immediately after exercise. The exercise-induced PDK4 and PEPCK mRNA expression in rodent white adipose tissue not only further demonstrates the plasticity of adipose tissue in response to exercise (16), but also demonstrates that exercise, similar to TZDs, is a potent stimuli which can be used to induce adipose tissue glyceroneogenic enzymes.

The rate of fatty acid re-esterification is proportional to the rate of lipolysis (46-47). Exercise is a robust activator of lipolysis in adipose tissue (48-49). Thus, it is reasonable to assume that exercise would induce PDK4 and PEPCK mRNA expression, presumably favoring the glyceroneogenic pathway for fatty acid re-esterification. It should be mentioned that, during exercise, although the relative rate of fatty acid reesterification remains constant (~30-40%), the absolute amount of fatty acids that are reesterified back to triglycerides (TG) is greatly enhanced owing to exercise-induced lipolysis. For instance, Wolfe et al. (50) quantified the TG-fatty acid cycling in humans during 4 hour walking. The TG-fatty acid cycling in adipocytes increased from 1.3  $\mu$ mol/kg/min at rest to 3  $\mu$ mol/kg/min throughout the exercise (50). Similarly, the whole body TG-fatty acid cycling is about 4 times higher in athletes than untrained volunteers (51). This was also supported by *ex vivo* and *in vivo* evidence from rodent studies. Treatment of cultured epididymal adipose tissue with norepinephrine *ex vivo* led to about 2 fold increase in TG-fatty acid cycling (47). Likewise, epinephrine infusion in rats resulted in about 2 fold increase in the glyceroneogenic pathways for fatty acid re-esterification in adipose tissue (52).

#### 7.4 IL-6 and Adipose Tissue Metabolism

To date the role of IL-6 in mediating adipose tissue metabolism is still under debate (53). Although evidence from *in vitro* studies suggest that IL-6 causes insulin resistance via decreasing adiponectin mRNA expression and negatively affecting insulin signaling pathways (54-56), there is also evidence that IL-6 may favorably regulate adipose tissue function. For instance, IL-6 activates AMPK in adipocytes (57), a central player in remodeling adipose tissue metabolism (58). This thesis was interested in the regulation of IL-6 on adipose tissue metabolism under two conditions of metabolic stress: an acute bout of exercise and chronic nutrient excess.

We found that IL-6 treatment in adipose tissue *ex vivo* activated AMPK signaling pathways independently of lipolysis. AMPK has been reported to regulate mitochondrial biogenesis in fat cells (59). It has also been suggested that AMPK controls the expression of adiponectin either directly (60) or secondary to increases in adipose tissue mitochondrial biogenesis (61). Adiponectin is intimately involved in mediating adipose tissue metabolism through its auto/paracrine actions on adipose tissue (62-63). Interstitial concentrations of IL-6 surrounding adipose tissue are orders of magnitude greater than levels in the circulation (64). Collectively, our findings suggested that IL-6 plays a pivotal role in maintaining adipose tissue function.

We also found that IL-6 plays an indirect role in mediating the effects of exercise on the induction of glyceroneogenic enzymes. During exercise, a wide range of myokines, hepatokines and adipokines are secreted from skeletal muscle, liver and adipose tissue, such as myostatin, IL-7, leukemia inhibitory factor, follistatin and irisin (65-67). A growing body of evidence has also demonstrated the existence of complex tissue to tissue communication during exercise at least partly through myokines, adipokines and hepatokines (66, 68). For instance it has recently been shown that exercise stimulates the secretion of chemokine (C-X-C motif) ligand 1 (CXCL1) from the liver through a muscle derived-IL-6 dependent mechanism (68). The current findings argue a potential role of IL-6 in mediating adipose tissue metabolism during exercise indirectly through its interactions with other organs or myokines, adipokines and hepatokines such as CXCL-1.

We further demonstrated that under conditions of calorie excess, the ablation of endogenous IL-6 leads to marked alterations in adipose tissue metabolism. The dysregulated phenotypes mainly include attenuated sensitivity to both insulin and catecholamines, as well as decreased adiponectin mRNA expression and secretion in adipose tissue. These findings further point towards a unique role of IL-6 in modulating adipose tissue insulin signaling, beta adrenergic responsiveness and adiponectin secretion. Thus, it is reasonable to assume that the increased secretion of IL-6 from adipose tissue during the development of obesity is a protective mechanism for the maintenance of adipose tissue function. We propose that at least part of the salutary effects of IL-6 on adipose tissue metabolism is through its action on the AMPK signaling pathway. Meanwhile, similar to the role that IL-6 is playing in exercise induced glyceroneogenic enzymes, it is likely that the effects of IL-6 on adipose tissue action is through its interactions with other organs or other hormonal factors such as glucagon-like peptide-1 (GLP-1) (69). For instance, it has been reported that elevated systemic IL-6 could stimulate GLP-1 secretion from intestinal L cells and pancreatic alpha cells, resulting in improved insulin secretion and glycemia (69).

In summary, this thesis proposes that IL-6 is essential for the metabolic homeostasis in adipose tissue under conditions of metabolic stress such as exercise and chronic nutrient excess. Future studies are required to further elucidate the mechanisms for the beneficial effects of IL-6 on adipose tissue metabolism.

## 7.5 Relevance of the Current Findings to Human Health Research

A large portion of TZD's beneficial effects on insulin sensitivity is through their up-regulation of PEPCK and PDK4 activity in adipose tissue. Currently, there are reservations about the use of TZDs because they are associated with multiple unwanted side effects, such as worsening of the coronary heart diseases, increasing body weight and fluid retention (45). In the current thesis, we found that an acute bout of exercise can induce PEPCK and PDK4 mRNA in epididymal adipose tissue from rodents, consequently resulting in increased fatty acid re-esterification in adipose tissue. This is in consistent with the findings by Wolfe et al. (50) who reported increased TG-fatty acid cycling in humans during 4 hours of walking and suggests that rodent models recapitulate human physiology to the extent that they are useful models for future studies in this area of research. Future studies are required to elucidate 1) if exercise can also induce PEPCK and PDK4 mRNA in adipose tissue from lean and obese human subjects and 2) if depotspecific and gender-specific responses exist for exercise induced glyceroneogenic enzymes. It is noteworthy that gender-specific responses to exercise have been reported in human population studies. For instance, in regards to the normal cardiac response to exercise, men had a higher systolic blood pressure compared to women (70). Females oxidize more fat and less carbohydrate and amino acids during endurance exercise compared to males, and this was at least partially owing to the higher levels of  $17-\beta$ estradiol in females (71). Of relevance to this thesis, women have a higher percentage of body fat in the gluteal-femoral region, in contrast, men have more body fat in the abdominal region (72). Using stable isotope methods, it has been reported that the rate of glycerol appearance was higher in females compared to male during submaximal cycling exercise (73). A gender difference in sympathetic nervous system activity during exercise has also been reported with males having a greater increase in plasma epinephrine and norepinephrine levels (74). Through the utilization of in-situ microdialysis it has been shown that women have a decreased  $\alpha^2$ -mediated antilipolysis (75) and increased  $\beta$ adrenergic mediated lipolysis (76) in abdominal subcutaneous tissue compared to men.

The current thesis also suggested that, in mice, IL-6 is essential for the maintenance of adipose tissue function (mainly the responses to both catecholamines and insulin, as well as adiponectin mRNA expression and secretion) under chronic nutrient excess. Similarly, the lack of functional IL-6 in patients with rheumatoid arthritis through IL-6 receptor-inhibiting monoclonal antibody leads to increased plasma lipid and glucose

levels, arguing against IL-6 as an insulin resistance causing agent in humans *in vivo* (77-78). Further studies are required to re-evaluate the effects of IL-6 on adipose tissue metabolism in human population studies during the development of insulin resistance.

#### **7.6 Future Directions**

The studies in this thesis support 1) an increasingly recognized role that adipose tissue is playing in mediating exercise-induced health benefits and 2) a unique role of IL-6 in modulating adipose tissue metabolism under conditions of metabolic stress. Although this thesis has contributed significantly to our understanding of exercisemediated adipose tissue metabolism and in elucidating the role of IL-6 in the etiology of metabolic derangements in adipose tissue, some important questions are still unanswered:

1) Whether IL-6 signaling is activated post-exercise and how IL-6 is involved in regulating glucose and lipid metabolism in adipose tissue and interacts with other tissues during the post-exercise period?

2) How exercise-induced myokines, hepatokines and adipokines communicate with each other and what are their physiological roles in exercise mediated whole body energy metabolism?

3) Besides epinephrine and IL-6, how are the other hormonal factors such as glucagon involved in mediating exercise induced adipose tissue metabolism?

4) For the observed phenotype in adipose tissue of IL-6<sup>-/-</sup> mice under HFD from this thesis, what is the causal event that leads to metabolic derangements in adipose tissue? What is the role of IL-6 in the etiology of adipose tissue and systemic insulin resistance? If increased IL-6 in adipose tissue is protective, does it act on adipose tissue metabolism directly or through its interactions with other organs or other hormonal factors?

5) Broadly, what messages can be obtained from the first three studies for the personalized exercise prescriptions in the prevention and/or treatment of obesity and T2DM over the long term? What information can be taken from the last study for the therapeutic treatment of obesity related disorders considering the enigmatic roles that IL-6 is playing systematically?

#### 7.6 Literature Cited

 Fried SK, Moustaid-Moussa N. Culture of adipose tissue and isolated adipocytes. Methods Mol Biol. 2001;155:197-212.

2. Ruan H, Zarnowski MJ, Cushman SW, Lodish HF. Standard isolation of primary adipose cells from mouse epididymal fat pads induces inflammatory mediators and down-regulates adipocyte genes. J Biol Chem. 2003;278(48):47585-93.

3. Skurk T, Ecklebe S, Hauner H. A novel technique to propagate primary human preadipocytes without loss of differentiation capacity. Obesity (Silver Spring). 2007;15(12):2925-31.

4. Kabir M, Catalano KJ, Ananthnarayan S, Kim SP, Van Citters GW, Dea MK, et al. Molecular evidence supporting the portal theory: a causative link between visceral adiposity and hepatic insulin resistance. Am J Physiol Endocrinol Metab. 2005;288(2):E454-61.

5. Rytka JM, Wueest S, Schoenle EJ, Konrad D. The portal theory supported by venous drainage-selective fat transplantation. Diabetes. 2011;60(1):56-63.

 DiGirolamo M, Fine JB, Tagra K, Rossmanith R. Qualitative regional differences in adipose tissue growth and cellularity in male Wistar rats fed ad libitum. Am J Physiol. 1998;274(5 Pt 2):R1460-7.

7. Berthiaume M, Laplante M, Festuccia W, Gelinas Y, Poulin S, Lalonde J, et al. Depot-specific modulation of rat intraabdominal adipose tissue lipid metabolism by pharmacological inhibition of 11beta-hydroxysteroid dehydrogenase type 1. Endocrinology. 2007;148(5):2391-7.

197

8. Shi H, Strader AD, Woods SC, Seeley RJ. The effect of fat removal on glucose tolerance is depot specific in male and female mice. Am J Physiol Endocrinol Metab. 2007;293(4):E1012-20.

9. Geenen D, Buttrick P, Scheuer J. Cardiovascular and hormonal responses to swimming and running in the rat. J Appl Physiol. 1988;65(1):116-23.

10. Higashida K, Higuchi M, Terada S. Potential role of lipin-1 in exercise-induced mitochondrial biogenesis. Biochem Biophys Res Commun. 2008;374(3):587-91.

11. Ravi Kiran T, Subramanyam MV, Asha Devi S. Swim exercise training and adaptations in the antioxidant defense system of myocardium of old rats: relationship to swim intensity and duration. Comp Biochem Physiol B Biochem Mol Biol. 2004;137(2):187-96.

12. Baker MA, Horvath SM. Influence of Water Temperature on Oxygen Uptake by Swimming Rats. J Appl Physiol. 1964;19:1215-8.

13. Stallknecht B, Vinten J, Ploug T, Galbo H. Increased activities of mitochondrial enzymes in white adipose tissue in trained rats. Am J Physiol. 1991;261(3 Pt 1):E410-4.

14. Dawson CA, Nadel ER, Horvath SM. Cardiac output in the cold-stressed swimming rat. Am J Physiol. 1968;214(2):320-5.

 Kaplan ML, Cheslow Y, Vikstrom K, Malhotra A, Geenen DL, Nakouzi A, et al.
Cardiac adaptations to chronic exercise in mice. Am J Physiol. 1994;267(3 Pt 2):H1167-73.

 Sutherland LN, Bomhof MR, Capozzi LC, Basaraba SA, Wright DC. Exercise and adrenaline increase PGC-1{alpha} mRNA expression in rat adipose tissue. J Physiol. 2009;587(Pt 7):1607-17.
17. Bukowiecki L, Lupien J, Follea N, Paradis A, Richard D, LeBlanc J. Mechanism of enhanced lipolysis in adipose tissue of exercise-trained rats. Am J Physiol. 1980;239(6):E422-9.

18. Konhilas JP, Maass AH, Luckey SW, Stauffer BL, Olson EN, Leinwand LA. Sex modifies exercise and cardiac adaptation in mice. Am J Physiol Heart Circ Physiol. 2004;287(6):H2768-76.

19. Rogers NH, Witczak CA, Hirshman MF, Goodyear LJ, Greenberg AS. Estradiol stimulates Akt, AMP-activated protein kinase (AMPK) and TBC1D1/4, but not glucose uptake in rat soleus. Biochem Biophys Res Commun. 2009;382(4):646-50.

20. Fu MH, Maher AC, Hamadeh MJ, Ye C, Tarnopolsky MA. Exercise, sex, menstrual cycle phase, and 17beta-estradiol influence metabolism-related genes in human skeletal muscle. Physiol Genomics. 2009;40(1):34-47.

21. Wohlers LM, Spangenburg EE. 17beta-estradiol supplementation attenuates ovariectomy-induced increases in ATGL signaling and reduced perilipin expression in visceral adipose tissue. J Cell Biochem. 2010;110(2):420-7.

22. Kopf M, Baumann H, Freer G, Freudenberg M, Lamers M, Kishimoto T, et al. Impaired immune and acute-phase responses in interleukin-6-deficient mice. Nature. 1994;368(6469):339-42.

23. Wong GT. Speed congenics: applications for transgenic and knock-out mouse strains. Neuropeptides. 2002;36(2-3):230-6.

24. Glund S, Krook A. Role of interleukin-6 signalling in glucose and lipid metabolism. Acta Physiol (Oxf). 2008;192(1):37-48.

199

25. Finegood DT, McArthur MD, Kojwang D, Thomas MJ, Topp BG, Leonard T, et al. Beta-cell mass dynamics in Zucker diabetic fatty rats. Rosiglitazone prevents the rise in net cell death. Diabetes. 2001;50(5):1021-9.

26. Moran TH, Bi S. Hyperphagia and obesity of OLETF rats lacking CCK1 receptors: developmental aspects. Dev Psychobiol. 2006;48(5):360-7.

27. Schroeder M, Zagoory-Sharon O, Shbiro L, Marco A, Hyun J, Moran TH, et al. Development of obesity in the Otsuka Long-Evans Tokushima Fatty rat. Am J Physiol Regul Integr Comp Physiol. 2009;297(6):R1749-60.

28. Haluzik M, Colombo C, Gavrilova O, Chua S, Wolf N, Chen M, et al. Genetic background (C57BL/6J versus FVB/N) strongly influences the severity of diabetes and insulin resistance in ob/ob mice. Endocrinology. 2004;145(7):3258-64.

29. Kjorholt C, Akerfeldt MC, Biden TJ, Laybutt DR. Chronic hyperglycemia, independent of plasma lipid levels, is sufficient for the loss of beta-cell differentiation and secretory function in the db/db mouse model of diabetes. Diabetes. 2005;54(9):2755-63.

30. Joost HG. Pathogenesis, risk assessment and prevention of type 2 diabetes mellitus. Obes Facts. 2008;1(3):128-37.

31. Sutherland LN, Capozzi LC, Turchinsky NJ, Bell RC, Wright DC. Time course of high-fat diet-induced reductions in adipose tissue mitochondrial proteins: potential mechanisms and the relationship to glucose intolerance. Am J Physiol Endocrinol Metab. 2008;295(5):E1076-83.

32. Seldin MF, Mott D, Bhat D, Petro A, Kuhn CM, Kingsmore SF, et al. Glycogen synthase: a putative locus for diet-induced hyperglycemia. J Clin Invest. 1994;94(1):269-76.

33. Surwit RS, Kuhn CM, Cochrane C, McCubbin JA, Feinglos MN. Diet-induced type II diabetes in C57BL/6J mice. Diabetes. 1988;37(9):1163-7.

34. Lin S, Thomas TC, Storlien LH, Huang XF. Development of high fat diet-induced obesity and leptin resistance in C57Bl/6J mice. Int J Obes Relat Metab Disord. 2000;24(5):639-46.

35. Surwit RS, Feinglos MN, Rodin J, Sutherland A, Petro AE, Opara EC, et al. Differential effects of fat and sucrose on the development of obesity and diabetes in C57BL/6J and A/J mice. Metabolism. 1995;44(5):645-51.

36. El Akoum S, Lamontagne V, Cloutier I, Tanguay JF. Nature of fatty acids in high fat diets differentially delineates obesity-linked metabolic syndrome components in male and female C57BL/6J mice. Diabetol Metab Syndr. 2011;3:34.

37. Winzell MS, Ahren B. The high-fat diet-fed mouse: a model for studying mechanisms and treatment of impaired glucose tolerance and type 2 diabetes. Diabetes. 2004;53 Suppl 3:S215-9.

King AJ. The use of animal models in diabetes research. Br J Pharmacol.
 2012;166(3):877-94.

39. Paigen B. Genetics of responsiveness to high-fat and high-cholesterol diets in the mouse. Am J Clin Nutr. 1995;62(2):458S-62S.

40. Alberti KG, Zimmet P, Shaw J. The metabolic syndrome--a new worldwide definition. Lancet. 2005;366(9491):1059-62.

41. Bray GA, Popkin BM. Dietary fat intake does affect obesity! Am J Clin Nutr. 1998;68(6):1157-73.

42. Simopoulos AP. Essential fatty acids in health and chronic disease. Am J Clin Nutr. 1999;70(3 Suppl):560S-9S.

43. Cadoudal T, Distel E, Durant S, Fouque F, Blouin JM, Collinet M, et al. Pyruvate dehydrogenase kinase 4: regulation by thiazolidinediones and implication in glyceroneogenesis in adipose tissue. Diabetes. 2008;57(9):2272-9.

44. Cadoudal T, Blouin JM, Collinet M, Fouque F, Tan GD, Loizon E, et al. Acute and selective regulation of glyceroneogenesis and cytosolic phosphoenolpyruvate carboxykinase in adipose tissue by thiazolidinediones in type 2 diabetes. Diabetologia. 2007;50(3):666-75.

45. Stafylas PC, Sarafidis PA, Lasaridis AN. The controversial effects of thiazolidinediones on cardiovascular morbidity and mortality. Int J Cardiol. 2009;131(3):298-304.

46. Reshef L, Hanson RW, Ballard FJ. A possible physiological role for glyceroneogenesis in rat adipose tissue. J Biol Chem. 1970;245(22):5979-84.

47. Brooks B, Arch JR, Newsholme EA. Effects of hormones on the rate of the triacylglycerol/fatty acid substrate cycle in adipocytes and epididymal fat pads. FEBS Lett. 1982;146(2):327-30.

48. Horowitz JF. Fatty acid mobilization from adipose tissue during exercise. Trends Endocrinol Metab. 2003;14(8):386-92.

49. Fernandez C, Hansson O, Nevsten P, Holm C, Klint C. Hormone-sensitive lipase is necessary for normal mobilization of lipids during submaximal exercise. Am J Physiol Endocrinol Metab. 2008;295(1):E179-86.

50. Wolfe RR, Klein S, Carraro F, Weber JM. Role of triglyceride-fatty acid cycle in controlling fat metabolism in humans during and after exercise. Am J Physiol. 1990;258(2 Pt 1):E382-9.

51. Romijn JA, Klein S, Coyle EF, Sidossis LS, Wolfe RR. Strenuous endurance training increases lipolysis and triglyceride-fatty acid cycling at rest. J Appl Physiol. 1993;75(1):108-13.

52. Nye CK, Hanson RW, Kalhan SC. Glyceroneogenesis is the dominant pathway for triglyceride glycerol synthesis in vivo in the rat. J Biol Chem. 2008;283(41):27565-74.
53. Allen TL, Febbraio MA. IL6 as a mediator of insulin resistance: fat or fiction?

Diabetologia. 2010;53(3):399-402.

54. Fasshauer M, Kralisch S, Klier M, Lossner U, Bluher M, Klein J, et al. Adiponectin gene expression and secretion is inhibited by interleukin-6 in 3T3-L1 adipocytes. Biochem Biophys Res Commun. 2003;301(4):1045-50.

55. Rotter V, Nagaev I, Smith U. Interleukin-6 (IL-6) induces insulin resistance in 3T3-L1 adipocytes and is, like IL-8 and tumor necrosis factor-alpha, overexpressed in human fat cells from insulin-resistant subjects. J Biol Chem. 2003;278(46):45777-84.

56. Lagathu C, Bastard JP, Auclair M, Maachi M, Capeau J, Caron M. Chronic interleukin-6 (IL-6) treatment increased IL-6 secretion and induced insulin resistance in adipocyte: prevention by rosiglitazone. Biochem Biophys Res Commun. 2003;311(2):372-9.

57. Kelly M, Keller C, Avilucea PR, Keller P, Luo Z, Xiang X, et al. AMPK activity is diminished in tissues of IL-6 knockout mice: the effect of exercise. Biochem Biophys Res Commun. 2004;320(2):449-54.

58. Gaidhu MP, Ceddia RB. The role of adenosine monophosphate kinase in remodeling white adipose tissue metabolism. Exerc Sport Sci Rev. 2011;39(2):102-8.

59. Gaidhu MP, Fediuc S, Anthony NM, So M, Mirpourian M, Perry RL, et al. Prolonged AICAR-induced AMP-kinase activation promotes energy dissipation in white adipocytes: novel mechanisms integrating HSL and ATGL. J Lipid Res. 2009;50(4):704-15.

60. Lihn AS, Jessen N, Pedersen SB, Lund S, Richelsen B. AICAR stimulates adiponectin and inhibits cytokines in adipose tissue. Biochem Biophys Res Commun. 2004;316(3):853-8.

61. Koh EH, Park JY, Park HS, Jeon MJ, Ryu JW, Kim M, et al. Essential role of mitochondrial function in adiponectin synthesis in adipocytes. Diabetes. 2007;56(12):2973-81.

62. Asterholm IW, Scherer PE. Enhanced metabolic flexibility associated with elevated adiponectin levels. Am J Pathol. 2010;176(3):1364-76.

63. Dietze-Schroeder D, Sell H, Uhlig M, Koenen M, Eckel J. Autocrine action of adiponectin on human fat cells prevents the release of insulin resistance-inducing factors. Diabetes. 2005;54(7):2003-11.

64. Nielsen NB, Hojbjerre L, Sonne MP, Alibegovic AC, Vaag A, Dela F, et al. Interstitial concentrations of adipokines in subcutaneous abdominal and femoral adipose tissue. Regul Pept. 2009;155(1-3):39-45. 65. Hansen J, Brandt C, Nielsen AR, Hojman P, Whitham M, Febbraio MA, et al. Exercise induces a marked increase in plasma follistatin: evidence that follistatin is a contraction-induced hepatokine. Endocrinology. 2011;152(1):164-71.

66. Bostrom P, Wu J, Jedrychowski MP, Korde A, Ye L, Lo JC, et al. A PGC1-alphadependent myokine that drives brown-fat-like development of white fat and thermogenesis. Nature. 2012;481(7382):463-8.

67. Pedersen BK, Febbraio MA. Muscles, exercise and obesity: skeletal muscle as a secretory organ. Nat Rev Endocrinol. 2012.

68. Pedersen L, Pilegaard H, Hansen J, Brandt C, Adser H, Hidalgo J, et al. Exerciseinduced liver chemokine CXCL-1 expression is linked to muscle-derived interleukin-6 expression. J Physiol. 2011;589(Pt 6):1409-20.

69. Ellingsgaard H, Hauselmann I, Schuler B, Habib AM, Baggio LL, Meier DT, et al. Interleukin-6 enhances insulin secretion by increasing glucagon-like peptide-1 secretion from L cells and alpha cells. Nat Med. 2011;17(11):1481-9.

69. Ellingsgaard H, Hauselmann I, Schuler B, Habib AM, Baggio LL, Meier DT, et al. Interleukin-6 enhances insulin secretion by increasing glucagon-like peptide-1 secretion from L cells and alpha cells. Nat Med. 2011;17(11):1481-9.

70. Gleim GW, Stachenfeld NS, Coplan NL, Nicholas JA. Gender differences in the systolic blood pressure response to exercise. Am Heart J. 1991;121(2 Pt 1):524-30.

71. Tarnopolsky MA. Sex differences in exercise metabolism and the role of 17-beta estradiol. Med Sci Sports Exerc. 2008;40(4):648-54.

72. Lemieux S, Prud'homme D, Bouchard C, Tremblay A, Despres JP. Sex differences in the relation of visceral adipose tissue accumulation to total body fatness. Am J Clin Nutr. 1993;58(4):463-7.

73. Carter SL, Rennie C, Tarnopolsky MA. Substrate utilization during endurance exercise in men and women after endurance training. Am J Physiol Endocrinol Metab. 2001;280(6):E898-907.

74. Galassetti P, Tate D, Neill RA, Morrey S, Davis SN. Effect of gender on counterregulatory responses to euglycemic exercise in type 1 diabetes. J Clin Endocrinol Metab. 2002;87(11):5144-50.

75. Hellstrom L, Blaak E, Hagstrom-Toft E. Gender differences in adrenergic regulation of lipid mobilization during exercise. Int J Sports Med. 1996;17(6):439-47.

76. Arner P, Kriegholm E, Engfeldt P, Bolinder J. Adrenergic regulation of lipolysis in situ at rest and during exercise. J Clin Invest. 1990;85(3):893-8. PMCID: 296507.

77. Choy EH, Isenberg DA, Garrood T, Farrow S, Ioannou Y, Bird H, et al. Therapeutic benefit of blocking interleukin-6 activity with an anti-interleukin-6 receptor monoclonal antibody in rheumatoid arthritis: a randomized, double-blind, placebo-controlled, dose-escalation trial. Arthritis Rheum. 2002;46(12):3143-50.

78. Febbraio MA, Rose-John S, Pedersen BK. Is interleukin-6 receptor blockade the Holy Grail for inflammatory diseases? Clin Pharmacol Ther. 2010;87(4):396-8.

206